SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of Earliest Event Reported): March 15, 2005

FLOWSERVE CORPORATION

(Exact name of registrant as specified in its charter)

New York (State or other jurisdiction of incorporation) 1-13179 (Commission File Number) 31-0267900 (I.R.S. Employer Identification No.)

5215 N. O'Connor Blvd., Suite 2300, Irving, Texas (Address of principal executive offices)

75039 (Zip Code)

Registrant's telephone number, including area code: (972) 443 - 6500

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

provisions.		
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)	
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)	
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))	
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))	

TABLE OF CONTENTS

Item 1.01. Entry into a Material Definitive Agreement.

Item 9.01. Financial Statements and Exhibits.

Signature

Exhibit Index

Limited Waiver and Third Amendment to First Amended and Restated Credit Agreement

Amendment No. 1 to Receivables Purchase Agreement

Amendment No. 2 to Receivables Purchase Agreement

Limited Waiver and Amendment No. 3 to Receivables Purchase Agreement

Finance Contract

Letter Amendment to Finance Contract

Letter of Credit and Reimbursement Agreement

First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement

Table of Contents

Item 1.01. Entry into a Material Definitive Agreement.

On March 15, 2005, Flowserve Corporation, a New York corporation (the "Company"), and certain of its subsidiaries, entered into the following amendments to certain financing facilities:

Limited Waiver and Third Amendment to First Amended and Restated Credit Agreement

The Company and Flowserve France SAS, a subsidiary of the Company, entered into a Limited Waiver and Third Amendment (the "<u>Third Amendment</u>") to that certain First Amended and Restated Credit Agreement dated as of May 2, 2002 (as amended, the "<u>Credit Agreement</u>") with the lenders named in the Credit Agreement (collectively, the "Lenders"), and Bank of America, N.A., a national banking association, as administrative agent for the Lenders.

The Third Amendment, among other things:

- · provides for a waiver of the Lenders of any default or event of default under certain representations under the Credit Agreement;
- provides for a waiver of the Lenders regarding the Company's delivery of its audited financial statements for the fiscal year ended December 31, 2004 until September 30, 2005, as well as any cross-default under the Credit Agreement arising from any event of default under any agreement governing material indebtedness arising from the Company's failure to deliver similar financial information or annual or quarterly reports for the fiscal year ending December 31, 2004 or for the first three fiscal quarters of the fiscal year ending December 31, 2005; and
- permits the Company to issue up to \$325 million of New Unsecured Debt in order to redeem, purchase or otherwise acquire for consideration the Subordinated Notes (as these terms are defined in the Credit Agreement).

The above discussion of the Third Amendment is a summary description of certain amendments and waivers to the Credit Agreement and is qualified in its entirety by the terms and conditions of the Third Amendment. For complete descriptions of the terms and conditions summarized in this report, reference is made to the Third Amendment attached hereto as Exhibit 10.1 and incorporated herein by reference.

Limited Waiver and Amendment No. 3 to Receivables Purchase Agreement

The Company, Flowserve Receivables Corporation, a subsidiary of the Company, and Flowserve US Inc., a subsidiary of the Company, entered into a Limited Waiver and Amendment No. 3 (the "Amendment") relating to (1) that certain Receivables Purchase Agreement dated as of October 7, 2004 (as amended, the "Receivables Purchase Agreement") with the funding sources named in the Receivables Purchase Agreement (collectively, the "Financial Institutions") and Jupiter Securitization Corporation (together with the Financial Institutions, the "Purchasers"), and J.P. Morgan Chase Bank, N.A., as agent for the Purchasers, and (2) that certain Receivables Sale Agreement dated October 7, 2004 and that certain Performance Undertaking dated October 7, 2004 that were executed pursuant to the Receivables Purchase Agreement (collectively, the "Receivables Agreements").

Table of Contents

The Amendment, among other things:

- provides for a waiver of the Purchasers regarding the Company's delivery of its audited financial statements for the fiscal year ended December 31, 2004 until September 30, 2005, as well as any cross-default under the Receivables Agreements arising from any event of default under any agreement governing material indebtedness arising from the Company's failure to deliver similar financial information or annual or quarterly reports for the fiscal year ending December 31, 2004 or for the first three fiscal quarters of the fiscal year ending December 31, 2005; and
- extends the Reporting Completion Date with respect to the Epic system of the Originator from March 31, 2005 to May 31, 2005 (as such terms are defined in the Receivables Purchase Agreement).

The above discussion of the Amendment is a summary description of certain amendments and waivers to the Receivables Agreements and is qualified in its entirety by the terms and conditions of the Amendment. For complete descriptions of the terms and conditions summarized in this report, reference is made to the Amendment attached hereto as Exhibit 10.4 and incorporated herein by reference.

First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement

The Company and Flowserve B.V, a subsidiary of the Company, entered into a First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement (the "First Amendment") to that certain Letter of Credit and Reimbursement Agreement dated as of July 28, 2004 (the "Letter of Credit Agreement") with Calyon New York Branch, as administrative agent and issuing lender, and the other lenders named in the Letter of Credit Agreement (collectively, the "Lenders").

The First Amendment, among other things:

- provides for a waiver of the Lenders regarding the Company's delivery of its audited financial statements for the fiscal year ended December 31, 2004 until September 30, 2005, as well as any cross-default under the Letter of Credit Agreement arising from any event of default under any agreement governing material indebtedness arising from the Company's failure to deliver similar financial information or annual or quarterly reports for the fiscal year ending December 31, 2004 or for the first three fiscal quarters of the fiscal year ending December 31, 2005; and
- permits the Company to issue up to \$325 million of New Unsecured Debt (as defined in the Letter of Credit Agreement).

The above discussion of the First Amendment is a summary description of certain amendments and waivers to the Letter of Credit Agreement and is qualified in its entirety by the terms and conditions of the First Amendment. For complete descriptions of the terms and conditions summarized in this report, reference is made to the First Amendment attached hereto as Exhibit 10.8 and incorporated herein by reference.

Item 9.01. Financial Statements and Exhibits.

(c) Exhibits

Table of Contents

Exhibit Number	Description
10.1	Limited Waiver and Third Amendment to First Amended and Restated Credit Agreement, dated March 15, 2005, entered into by and among Flowserve Corporation, Flowserve France SAS, the lenders named therein and Bank of America, N.A., as administrative agent for the lenders.
10.2	Amendment No. 1 to Receivables Purchase Agreement, dated January 4, 2005, entered into by and among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.3	Amendment No. 2 to Receivables Purchase Agreement, dated January 14, 2005, entered into by and among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.4	Limited Waiver and Amendment No. 3 to Receivables Purchase Agreement, dated March 15, 2005, entered into by and among Flowserve Corporation, Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.5	Finance Contract, dated April 19, 2004, entered into by and among Flowserve Corporation, Flowserve B.V. and European Investment Bank.
10.6	Letter Amendment to Finance Contract, dated July 2, 2004, entered into by and among Flowserve Corporation, Flowserve B.V. and European Investment Bank.
10.7	Letter of Credit and Reimbursement Agreement, dated July 28, 2004, entered into by and among Flowserve B.V., Calyon New York Branch as administrative agent and issuing lender and the other lenders named therein.
10.8	First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement, dated March 15, 2005, entered into by and among Flowserve B.V., Calyon New York Branch as administrative agent and issuing lender and the other lenders named therein.

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

FLOWSERVE CORPORATION

By: /s/ Ronald F. Shuff
Ronald F. Shuff

Vice President, Secretary and General Counsel

Date: March 18, 2005

Exhibit Index

Exhibit Number	Description
10.1	Limited Waiver and Third Amendment to First Amended and Restated Credit Agreement, dated March 15, 2005, entered into by and among Flowserve Corporation, Flowserve France SAS, the lenders named therein and Bank of America, N.A., as administrative agent for the lenders.
10.2	Amendment No. 1 to Receivables Purchase Agreement, dated January 4, 2005, entered into by and among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.3	Amendment No. 2 to Receivables Purchase Agreement, dated January 14, 2005, entered into by and among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.4	Limited Waiver and Amendment No. 3 to Receivables Purchase Agreement, dated March 15, 2005, entered into by and among Flowserve Corporation, Flowserve Receivables Corporation, Flowserve US Inc., the funding sources named therein, Jupiter Securitization Corporation and J.P. Morgan Chase Bank, N.A. as agent for the purchasers.
10.5	Finance Contract, dated April 19, 2004, entered into by and among Flowserve Corporation, Flowserve B.V. and European Investment Bank.
10.6	Letter Amendment to Finance Contract, dated July 2, 2004, entered into by and among Flowserve Corporation, Flowserve B.V. and European Investment Bank.
10.7	Letter of Credit and Reimbursement Agreement, dated July 28, 2004, entered into by and among Flowserve B.V., Calyon New York Branch as administrative agent and issuing lender and the other lenders named therein.
10.8	First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement, dated March 15, 2005, entered into by and among Flowserve B.V., Calyon New York Branch as administrative agent and issuing lender and the other lenders named therein.

FLOWSERVE CORPORATION

LIMITED WAIVER AND THIRD AMENDMENT TO FIRST AMENDED AND RESTATED CREDIT AGREEMENT

This LIMITED WAIVER AND THIRD AMENDMENT TO FIRST AMENDED AND RESTATED CREDIT AGREEMENT (this "Third Amendment") is dated as of March 15, 2005 and entered into by and among Flowserve Corporation, a New York corporation (the "Company"), Flowserve France SAS (the "Subsidiary Borrower"), the Guarantors of the Company listed on the signature pages hereof (only for the purposes of Section 6), the financial institutions executing the Consent of Lender (the "Consent") in the form of Exhibit A annexed hereto (each individually a "Lender" and collectively the "Lenders"), and Bank of America, N.A., a national banking association, as administrative agent for the Lenders (in such capacity, the "Administrative Agent"), and is made with reference to that certain First Amended and Restated Credit Agreement dated as of May 2, 2002, as amended by that certain First Amendment to First Amended and Restated Credit Agreement dated as of June 30, 2003 and that certain Second Amendment to First Amended and Restated Credit Agreement dated as of June 24, 2004 (as so amended, the "Credit Agreement"), each by and among the Company, the Subsidiary Borrower, the Guarantors, the Lenders party thereto, the Administrative Agent and Credit Suisse First Boston, as syndication agent. Capitalized terms used herein without definition shall have the same meanings as set forth in the Credit Agreement.

RECITALS

WHEREAS, pursuant to Sections 5.04(a) and 5.04(c) of the Credit Agreement the Company is required to deliver the audited financial statements, opinion and the accountants' certificate referred to therein within 100 days after the end of each fiscal year;

WHEREAS, the Company has requested that the Required Lenders waive compliance with such requirements under Sections 5.04(a) and 5.04(c) of the Credit Agreement with respect to the fiscal year ended December 31, 2004 until September 30, 2005 and any consequences resulting from such noncompliance during such period;

WHEREAS, the Company and the Lenders desire to amend the Credit Agreement to (i) permit the Company to redeem, repurchase or otherwise acquire for consideration the Subordinated Notes, and (ii) amend the defined term New Subordinated Debt to include new senior unsecured debt;

NOW, **THEREFORE**, in consideration of the premises and the agreements, provisions and covenants herein contained, the parties hereto agree as follows:

Section 1. AMENDMENTS TO CREDIT AGREEMENT

A. Amendments to Definitions

1. The definition of "Consolidated EBITDA" in Section 1.01 of the Credit Agreement is hereby amended and restated in its entirety to read as follows:

"Consolidated EBITDA' shall mean, for any period, Consolidated Net Income for such period, plus (a) without duplication and to the extent deducted in determining such Consolidated Net Income, the sum of (i) Consolidated Interest Expense for such period, (ii) consolidated income tax expense for such period, (iii) all amounts attributable to depreciation and amortization for such period, (iv) any extraordinary losses or extraordinary non-cash charges for such period, (v) the amount of premium payments made by Company or its Subsidiaries associated with the repurchase or prepayment of the Subordinated Notes from the proceeds of the Fall 2001 Equity Issuance and the amount of such premium payments and unamortized fees associated with any further repurchase or prepayment of the Subordinated Notes to the extent such repurchase or prepayment is permitted hereunder, (vi) integration and restructuring charges in connection with the IDP Transactions and taken with respect to periods ended on or prior to December 31, 2001, (vii) integration and restructuring charges in connection with the Acquisition and taken with respect to periods ended on or prior to June 30, 2004, in an aggregate amount not to exceed \$40,000,000, and (viii) restructuring and integration charges taken with respect to periods beginning on July 1, 2003 and ended on or prior to December 31, 2004, in an aggregate amount not to exceed \$15,000,000, and minus (b) without duplication and to the extent included in determining such Consolidated Net Income, any extraordinary gains for such period, all determined on a consolidated basis in accordance with GAAP; provided that in the case of the Company, Consolidated EBITDA shall be determined with reference to Schedule 1.01(d)."

2. The definition of "Consolidated Interest Expense" in Section 1.01 of the Credit Agreement is hereby amended and restated in its entirety to read as follows:

""Consolidated Interest Expense' shall mean, for any period, for the Company and its Subsidiaries on a consolidated basis, the sum, without duplication, of (a) all interest, premium payments, fees, charges and related expenses payable by the Company and its Subsidiaries in connection with borrowed money (including capitalized interest) (other than premium payments associated with the repurchase or prepayment of the Subordinated Notes from proceeds of the Fall 2001 Equity Issuance and premium payments and unamortized fees associated with any further repurchase or prepayment of the Subordinated Notes to the extent such repurchase or prepayment is permitted hereunder) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP and payable in cash, (b) the portion of rent payable by the Company and its Subsidiaries with respect to such period under capital leases that is treated as interest in accordance with GAAP and payable in cash and (c) all fees, discounts, premiums, expenses or similar amounts incurred by the Company or any of its Subsidiaries in connection with the Receivables Program for such period, including purchase discounts

(net of any loss reserves), purchase premiums, operating expense fees, structuring fees, collection agent fees, unutilized purchase limit fees and other similar fees and expenses."

3. The definition of "New Subordinated Debt" in Section 1.01 of the Credit Agreement is hereby amended by deleting it in its entirety and substituting therefor the following defined term:

"New Unsecured Debt' means Indebtedness having the following characteristics: (i) the issuer shall be the Company and/or FFBV and such Indebtedness may be guaranteed by one or more of the Guarantors only, (ii) such Indebtedness shall be unsecured, (iii) if such Indebtedness is subordinated, such Indebtedness shall be subordinated in right of payment to the Obligations in a manner reasonably acceptable to the Administrative Agent, (iv) such Indebtedness shall not have any scheduled payment of principal, scheduled prepayment, scheduled mandatory redemption or sinking fund payment prior to December 31, 2009, (v) the Net Cash Proceeds of such Indebtedness shall be applied as required by Section 2.13(e), (vi) such Indebtedness shall not contain any provision prohibiting the creation or assumption of any Lien on any of the properties or assets of Company or its Subsidiaries, whether then owned or thereafter acquired, or prohibiting guaranties by Company or any of its Subsidiaries to secure payment of the Obligations or any agreement renewing, refinancing or extending the Obligations or this Agreement, (vii) the Company shall be in compliance with Sections 6.11, 6.12 and 6.13 on a *pro forma* basis after giving effect to the incurrence of such Indebtedness, (viii) other terms and conditions shall be no less favorable to the Company or its Subsidiaries or the Lenders in any material respect than the terms and conditions applicable to the Subordinated Notes, and (ix) such Indebtedness shall be issued pursuant to documentation reasonably satisfactory to the Administrative Agent."

4. The definition of "Total Debt" in Section 1.01 of the Credit Agreement is hereby amended deleting it in its entirety and substituting therefor the following defined term:

"'Total Debt' shall mean, at any time, the total consolidated Indebtedness of the Company and the Subsidiaries at such time (excluding (a) Indebtedness under Section 6.01(k), and (b) Indebtedness of the type described in clause (i) of the definition of such term and under Section 6.01(l), except in each case to the extent of any unreimbursed drawings or payments thereunder)."

B. Amendments to Article II - The Credits.

1. Section 2.13(e) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:

"(e) Subject to paragraph (k) below, in the event that any Loan Party or any subsidiary of a Loan Party shall receive Net Cash Proceeds from (i) the issuance or other disposition of Indebtedness for money borrowed of any Loan Party or any subsidiary of a Loan Party (other than Indebtedness for money borrowed permitted pursuant to Section 6.01; provided that Net Cash Proceeds from the issuance of any New Unsecured Debt shall be applied to redeem, repurchase, prepay or otherwise acquire for consideration the Subordinated Notes

(including any premium payments) and any remaining Net Cash Proceeds shall be applied to prepay Term Loans in accordance with Section 2.13(h)) or (ii) the establishment of the Receivables Program or any subsequent increase thereto, the Borrowers shall, substantially simultaneously with (and in any event not later than the fifth Business Day next following) the receipt of such Net Cash Proceeds by a Loan Party or a subsidiary, apply an amount equal to 100% of such Net Cash Proceeds to prepay outstanding Term Loans in accordance with Section 2.13(h)."

C. Amendments to Article VI -Negative Covenants

- 1. Section 6.01(o) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:
- "(o) Indebtedness constituting New Unsecured Debt in an aggregate principal amount not to exceed \$325,000,000 or its Dollar Equivalent; and"
- 2. Section 6.08 of the Credit Agreement is hereby amended and restated in its entirety to read as follows:

"Engage at any time in any business or business activity other than the business currently conducted by the Company and the Subsidiaries and business activities reasonably incidental thereto, including any activities permitted hereunder, which, in the case of Finsub, shall be limited solely to performing its obligations under the Receivables Program Documentation and, in the case of FFBV, shall be limited solely to performing its obligations under the Subordinated Note Documents and the Loan Documents and any document pursuant to which the New Unsecured Debt is issued; provided, however, Company shall be entitled to create a wholly-owned subsidiary engaged solely in the business of providing the insurance coverage required under Section 5.02 hereof solely to the Company and the Subsidiaries, so long as such subsidiary is adequately capitalized to satisfy the requirements of Section 5.02 and investments therein do not exceed \$1,000,000 in the aggregate (a "Captive Insurance Company")."

3. Section 6.09(b) of the Credit Agreement is hereby amended and restated in its entirety to read as follows:

"(b) (i) Make any distribution, whether in cash, property, securities or a combination thereof, other than regular scheduled payments of interest as and when due (to the extent not prohibited by applicable subordination provisions), in respect of, or pay, or offer or commit to pay, or directly or indirectly redeem, repurchase, retire or otherwise acquire for consideration, or set apart any sum for the aforesaid purposes, any subordinated Indebtedness (except the redemption, repurchase or prepayment of the Subordinated Notes with the proceeds of the issuance of New Unsecured Debt), or (ii) pay in cash any amount in respect of any Indebtedness or preferred equity interests that may at the obligor's option be paid in kind or in other securities; provided that the Borrower was permitted to repurchase and/or prepay the Subordinated Notes to the extent provided in the final proviso of Section 2.13(c) of the Original Credit Agreement."

Section 2. WAIVER OF CERTAIN FINANCIAL REPORTING COVENANTS

A. Subject to the terms and conditions set forth herein and in reliance on the representations and warranties of the Company herein contained, the Lenders hereby waive any Default or Event of Default arising from any incorrect representation under the last sentence of Section 3.16 of the Credit Agreement on the First Amended and Restated Credit Agreement Closing Date.

B. Subject to the terms and conditions set forth herein and in reliance on the representations and warranties of the Company herein contained, the Lenders hereby temporarily waive (i) compliance with the requirement under Sections 5.04(a) and (c) of the Credit Agreement that the financial statements for the fiscal year ending December 31, 2004 delivered pursuant to Section 5.04(a) of the Credit Agreement are audited and accompanied by an opinion of an independent public accountant and a certificate of such accountant certifying matters set forth in Section 5.04(c) of the Credit Agreement, (ii) any consequences or further affirmative obligations of the Company under the Credit Agreement resulting from such noncompliance, and (iii) any Default or Event of Default arising from an event of default under any agreement or instrument evidencing or governing Material Indebtedness of the Company or any Subsidiary arising from the failure to provide similar financial information (or opinions or certifications thereof) or annual or quarterly public filings during or for the fiscal year ending December 31, 2004 or for the first three fiscal quarters of the fiscal year ending December 31, 2005; in each case, during the period from the Third Amendment Effective Date (as defined in Section 3) through and including the earlier to occur of (a) the date of delivery of audited financial statements for the fiscal year ending December 31, 2004 and accompanying accountant's opinion and certificate in accordance with Sections 5.04(a) and (c) of the Credit Agreement to the Lenders, and (b) September 30, 2005; provided, however, all waivers and accommodations made to the Company in this Section 2B shall be rescinded and be null and void and of no force and effect upon the commencement by any holder of Material Indebtedness of the Company or any Subsidiary of any action exercising rights with respect to collateral or rights to accelerate arising from any default or event of default under any agreement or instrument evidencing or gover

C. The waivers set forth in Sections 2A and 2B (the "Waiver") shall be limited precisely as written and relate solely to the noncompliance or temporary noncompliance, as the case may be, by the Company with the provisions of the Credit Agreement in the manner and to the extent described above, and nothing in this Waiver shall be deemed to:

1. constitute a waiver of compliance by the Company with respect to (i) Section 5.04 of the Credit Agreement in any other instance or (ii) any other term, provision or condition of the Credit Agreement or any other instrument or agreement referred to therein; or

2. prejudice any right or remedy that Administrative Agent or any Lender may now have (except to the extent such right or remedy was based upon existing defaults that will not exist after giving effect to this Waiver) or may have in the future under or in connection with the Credit Agreement or any other instrument or agreement referred to therein.

D. Except as expressly set forth herein, the terms, provisions and conditions of the Credit Agreement shall remain in full force and effect and in all other respects are hereby ratified and confirmed.

Section 3. CONDITIONS TO EFFECTIVENESS FOR WAIVER AND AMENDMENT

Sections 1 and 2 of this Third Amendment shall become effective only upon satisfaction of all of the following conditions precedent (the date of such satisfaction being referred to herein as the "Third Amendment Effective Date"):

- A. On the Third Amendment Effective Date, (a) after giving effect to Sections 1 and 2 hereof, the representations and warranties contained in Section 4 hereof and in Article III of the Credit Agreement shall be true and correct as of such date, as though made on and as of such date; (b) after giving effect to Sections 1 and 2 hereof, no Default or Event of Default shall then exist; and (c) the Company shall deliver to the Administrative Agent a certificate signed by a Responsible Officer of Company confirming the foregoing;
- **B.** On or prior to the Third Amendment Effective Date, the Company shall deliver to the Lenders (or to the Administrative Agent for the Lenders) copies of this Third Amendment executed by each Loan Party;
 - C. On or prior to the Third Amendment Effective Date, the Required Lenders shall have executed the Consent; and
- **D.** The Administrative Agent shall have received payment, for the account of each Lender that executes this Third Amendment on or before 5:00 p.m. Eastern Standard Time on March 15, 2005, of an amendment fee equal to 0.02% of the sum of such Lender's Revolving Credit Commitment and the principal amount of Term Loans held by such Lender on the Third Amendment Effective Date.

Section 4. REPRESENTATIONS AND WARRANTIES

In order to induce the Lenders to enter into this Third Amendment, after giving effect to Sections 1 and 2 hereof, the Company represents and warrants to each Lender that the following statements are true, correct and complete on and as of the Third Amendment Effective Date:

- **A. Organization; Powers.** The Company and each of the Subsidiaries (i) (a) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (b) has all requisite power and authority to own its property and assets and to carry on its business as now conducted and as proposed to be conducted and (c) is qualified to do business in, and is in good standing in, every jurisdiction where such qualification is required, except where any such failure, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, and (ii) in the case of the Company and each Subsidiary that is a party hereto, has the power and authority to execute, deliver and perform its obligations under the Credit Agreement as amended by this Third Amendment (the "Amended Agreement").
- **B.** Authorization. The Third Amendment (i) has been duly authorized by all requisite corporate, and, if required, stockholder, action on the part of the Company and each

Subsidiary that is a party hereto and (ii) will not (a) violate (1) any provision of law, statute, rule or regulation, or of the certificate or articles of incorporation or other constitutive documents or by-laws of the Company or any Subsidiary, (2) any order of any Governmental Authority or (3) any provision of any indenture, agreement or other instrument to which the Company or any Subsidiary is a party or by which any of them or any of their property is or may be bound, (b) be in conflict with, result in a breach of or constitute (alone or with notice or lapse of time or both) a default under, or give rise to any right to accelerate or to require the prepayment, repurchase or redemption of any obligation under any such indenture, agreement or other instrument or (c) result in the creation or imposition of any Lien upon or with respect to any property or assets now owned or hereafter acquired by the Company or any Subsidiary (other than any Lien created under the First Amended and Restated Security Documents).

- **C. Enforceability.** This Third Amendment has been duly executed and delivered by each Loan Party which is a party hereto, and this Third Amendment and the Amended Agreement constitute a legal, valid and binding obligation of each such Loan Party enforceable against each such Loan Party in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.
- **D. Governmental Approvals.** No action, consent or approval of, registration or filing with or any other action by any Governmental Authority is or will be required in connection with the execution and delivery of this Third Amendment, and the performance by the Company of the Amended Agreement.
- **E. Incorporation of Representations and Warranties from Credit Agreement.** The representations and warranties contained in Article III of the Credit Agreement are incorporated herein by this reference and are and will be true, correct and complete in all material respects on and as of the Third Amendment Effective Date to the same extent as though made on and as of that date, except to the extent such representations and warranties specifically relate to an earlier date, in which case they were true, correct and complete in all material respects on and as of such earlier date.
- **F. Absence of Default.** No event has occurred and is continuing or will result from the consummation of the transactions contemplated by this Third Amendment that would constitute a Default or an Event of Default.

Section 5. MISCELLANEOUS

A. Reference to and Effect on the Credit Agreement and the Other Loan Documents.

1. On and after the Third Amendment Effective Date, each reference in the Credit Agreement to "this Agreement", "hereunder", "herein" or words of like import referring to the Credit Agreement, and each reference in the other Loan Documents to the "First Amended and Restated Credit Agreement", "Credit Agreement", "thereunder", "thereof" or words of like import referring to the Credit Agreement shall mean and be a reference to the Amended Agreement.

- 2. Except as specifically amended by this Third Amendment, the Credit Agreement and the other Loan Documents shall remain in full force and effect and are hereby ratified and confirmed.
- 3. The execution, delivery and performance of this Third Amendment shall not, except as expressly provided in Section 2, constitute a waiver of any provision of, or operate as a waiver of any right, power or remedy of the Administrative Agent or any Lender under, the Credit Agreement or any of the other Loan Documents.
- **B. Fees and Expenses.** The Company acknowledges that all costs, fees and expenses as described in Section 9.05 of the Credit Agreement incurred by the Administrative Agent and its counsel with respect to this Third Amendment and the documents and transactions contemplated hereby shall be for the account of the Company.
- C. Delivery of Financial Statements. The Company hereby agrees to provide within 100 days after the end of the fiscal year ending December 31, 2004, its unaudited consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Company and its consolidated Subsidiaries as of the close of the fiscal year ending 2004 and the results of its operations and the operations of such Subsidiaries during such year, and a certificate of a Financial Officer opining on or certifying such statements and certifying that no Event of Default or Default has occurred, specifying the nature and extent thereof and any corrective action taken or proposed to be taken with respect thereto, and, setting forth the Company's calculation of Excess Cash Flow.
- **D. Headings.** Section and subsection headings in this Third Amendment are included herein for convenience of reference only and shall not constitute a part of this Third Amendment for any other purpose or be given any substantive effect.
- **E. Applicable Law.** THIS THIRD AMENDMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES.
- **F. Counterparts.** This Third Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed an original, but all such counterparts together shall constitute but one and the same instrument; signature pages may be detached from multiple separate counterparts and attached to a single counterpart so that all signature pages are physically attached to the same document.

Section 6. ACKNOWLEDGEMENT AND CONSENT BY CREDIT SUPPORT PARTIES

A. Each of the Guarantors is a party to the First Amended and Restated Guarantee Agreement and certain of the First Amended and Restated Security Documents. The Guarantors are collectively referred to herein as the "Credit Support Parties," and the First Amended and Restated Guarantee Agreement and the First Amended and Restated Security Documents are collectively referred to herein as the "Credit Support Documents."

B. Each Credit Support Party hereby acknowledges that it has read this Third Amendment and consents to the terms thereof, and hereby confirms and agrees that, notwithstanding the effectiveness of this Third Amendment, the obligations of each Guarantor under each Credit Support Document to which it is a party or otherwise bound shall not be impaired or affected and each Credit Support Document is, and shall continue to be, in full force and effect and is hereby confirmed and ratified in all respects. Each Guarantor further agrees that nothing in the Credit Agreement, this Third Amendment or any other Loan Document shall be deemed to require the consent of such Guarantor to any future amendment to the Credit Agreement.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Third Amendment to be duly executed and delivered by their respective officers thereunto duly authorized as of the date first written above.

FLOWSERVE CORPORATION, as the Company

By /s/ John M. Nanos

Name: John M. Nanos Title: Assistant Secretary

FLOWSERVE FRANCE SAS, societe par actions simplifiee organized under the laws of France, as the Subsidiary Borrower

By /s/ John M. Nanos

Name: John M. Nanos Title: Director

FLOWSERVE US INC. FLOWSERVE INTERNATIONAL, INC. FLOWSERVE MANAGEMENT COMPANY **BW/IP-NEW MEXICO, INC.** FLOWSERVE INTERNATIONAL, LLC INGERSOLL-DRESSER PUMP COMPANY FLOWSERVE HOLDINGS, INC.

(for purposes of Section 6 only) as Credit Support Parties

By /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

S-1

FLOWSERVE FINANCE B.V., (for purposes of Section 6 only) as a Credit Support Party $\,$

By /s/ John M . Nanos

Name: John M. Nanos Title: Managing Director

By /s/ Adrianus H.G.M. Witters

Name: Adrianus H.G.M. Witters Title: Managing Director

FLOWSERVE INTERNATIONAL LIMITED, (for purposes of Section 6 only) as Credit Support Party

By /s/ John M. Nanos

Name: John M. Nanos Title: Director

S-2

BANK OF AMERICA, N.A., individually and as the Administrative Agent, the Collateral Agent and the Swingline Lender $\,$

By /s/ John Pocalyko

Name: John Pocalyko Title: Senior Vice President

S-3

EXHIBIT A

CONSENT OF LENDER

This Consent of Lender is delivered by the undersigned Lender with reference to that certain First Amended and Restated Credit Agreement dated as of May 2, 2002, as amended, among Flowserve Corporation, a New York corporation, Flowserve France SAS, societe par actions simplifiee organized under the laws of France, as the Subsidiary Borrower, the Lenders from time to time party thereto (the "Lenders"), Bank of America, N.A., as Administrative Agent, Swingline Lender and Collateral Agent, and Credit Suisse First Boston, as Syndication Agent. The undersigned Lender hereby consents to the Limited Waiver and Third Amendment to First Amended and Restated Credit Agreement dated as of _____, 2005.

	[Name of Lender]
Ву:	
Name:	
Title:	

AMENDMENT NO . 1

This AMENDMENT NO. 1 (this "Amendment"), is dated as of January 4, 2005, and is entered into by and among FLOWSERVE RECEIVABLES CORPORATION, a Delaware corporation (the "Seller"), FLOWSERVE US INC., a Delaware corporation, as the Servicer (the "Servicer"), the funding sources party hereto as the financial institutions (the "Financial Institutions"), JUPITER SECURITIZATION CORPORATION (together with the Financial Institutions, the "Purchasers"), J. P. MORGAN SECURITIES INC. and JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as agent (the "Agent") for the Purchasers.

WITNESSETH:

WHEREAS, the Seller, the Servicer, the Purchasers and the Agent are parties to that certain Receivables Purchase Agreement, dated as of October 7, 2004 (as amended, restated, supplemented or otherwise modified from time to time, the "RPA");

WHEREAS, the Seller, J. P. Morgan Securities Inc., Jupiter Securitization Corporation and the Agent are parties to that certain Fee Letter, dated as of October 7, 2004 (as amended, restated, supplemented or otherwise modified from time to time, the "Fee Letter" and together with the RPA, the "Agreements"); and

WHEREAS the parties hereto desire to amend the Agreements on the terms and conditions set forth below;

NOW THEREFORE, in consideration of the premises herein contained, and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>Defined Terms</u>. Capitalized terms used and not otherwise defined herein shall have the meanings assigned to such terms in the RPA.
- 2. <u>Amendments to the RPA</u>. Subject to the satisfaction of the conditions precedent set forth in Section 6 below, the RPA is hereby amended as follows:
- (a) All references in the RPA to "Bank One" or "Bank One, NA (Main Office Chicago)" shall mean and constitute references to "JPMorgan Chase Bank, N.A. (successor by merger to Bank One, NA (Main Office Chicago))."
- (b) The definition of "Eligible BAAN Receivable" set forth in Exhibit I to the RPA is hereby amended to delete the reference to the date "December 31, 2004" and to replace such date with the date "January 14, 2005."

- 3. <u>Amendments to the Fee Letter</u>. Subject to the satisfaction of the conditions precedent set forth in <u>Section 6</u> below, the Fee Letter is hereby amended as follows:
 - (a) All references in the Fee Letter to "Bank One" or "Bank One, NA (Main Office Chicago)" shall mean and constitute references to "JPMorgan Chase Bank, N.A. (successor by merger to Bank One, NA (Main Office Chicago))."
 - (b) The definition of "Facility Fee Rate" set forth in Exhibit I to the Fee Letter is hereby amended to delete each reference to the date "December 31, 2004" and to replace each reference to such date with the date "January 14, 2005."
- 4. <u>Representations and Warranties of the Seller</u>. In order to induce the parties hereto to enter into this Amendment, the Seller represents and warrants that:
 - (a) The representations and warranties of Seller set forth in Section 5.1 of the RPA are true, correct and complete on the date hereof as if made on and as of the date hereof and there exists no Amortization Event or Potential Amortization Event on the date hereof, provided that in the case of any representation or warranty in Section 5.1 of the RPA that expressly relates to facts in existence on an earlier date, the reaffirmation thereof under this Section 4(a) shall be made as of such earlier date.
 - (b) The execution and delivery by the Seller of this Amendment has been duly authorized by proper corporate proceedings of the Seller and this Amendment, and the Agreements, as amended by this Amendment, constitute the legal, valid and binding obligations of the Seller, enforceable against the Seller in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights generally.
- 5. <u>Representations and Warranties of the Servicer</u>. In order to induce the parties hereto to enter into this Amendment, the Servicer represents and warrants that:
 - (a) The representations and warranties of the Servicer set forth in Section 5.1 of the RPA are true, correct and complete on the date hereof as if made on and as of the date hereof and there exists no Amortization Event or Potential Amortization Event on the date hereof, provided that in the case of any representation or warranty in Section 5.1 of the RPA that expressly relates to facts in existence on an earlier date, the reaffirmation thereof under this Section 5(a) shall be made as of such earlier date.
 - (b) The execution and delivery by the Servicer of this Amendment has been duly authorized by proper corporate proceedings of the Servicer and this Amendment, and the Agreements, as amended by this Amendment, constitute the legal, valid and binding obligations of the Servicer, enforceable against the Servicer in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights generally.

- 6. <u>Conditions Precedent</u>. The amendments to the Agreements provided for hereunder shall become effective as of December 31, 2004 upon the Agent's receipt of each of the following:
 - (a) counterparts of this Amendment executed by the Seller, the Servicer, J. P. Morgan Securities Inc. and each Purchaser; and
 - (b) a Reaffirmation of Performance Undertaking in substantially the form attached hereto as <u>Exhibit A</u> hereto, executed by the Provider in respect of the Performance Undertaking.
 - 7. Ratification. The Agreements, as amended hereby, are hereby ratified, approved and confirmed in all respects.
- 8. <u>Reference to Agreements</u>. From and after the effective date hereof, each reference in any Agreement to "this Agreement", "hereof", or "hereunder" or words of like import, and all references to such Agreement in any and all agreements, instruments, documents, notes, certificates and other writings of every kind and nature shall be deemed to mean such Agreement as amended by this Amendment.
- 9. <u>Costs and Expenses</u>. The Seller agrees to pay all reasonable costs, fees and out-of-pocket expenses (including attorneys' fees and time charges of attorneys representing the Agent, which attorneys may be employees of the Agent) incurred by the Agent in connection with the preparation, execution and enforcement of this Amendment.

10. CHOICE OF LAW. THIS AMENDMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS.

11. <u>Execution of Counterparts</u>. This Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered by their duly authorized officers as of the date first written above:

FLOWSERVE RECEIVABLES CORPORATION, as Seller

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

FLOWSERVE US INC.,

as Servicer

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

Signature Page to Amendment No. 1

JUPITER SECURITIZATION CORPORATION

By: /s/ Leo Loughead

Name: Leo Loughead Title: Authorized Signer

JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as a Financial Institution and as Agent

By: <u>/s/ Leo Loughead</u> Name: Leo Loughead

Title: Managing Director, Capital Markets

J. P. MORGAN SECURITIES INC.

By: /s/ Leo Loughead

Name: Leo Loughead

Title: Managing Director, Capital Markets

Signature Page to Amendment No. 1

EXHIBIT A

FORM OF REAFFIRMATION OF PERFORMANCE UNDERTAKING

(attached)

REAFFIRMATION OF PERFORMANCE UNDERTAKING

January 4, 2005

JPMorgan Chase Bank, N.A. (successor by merger to Bank One, N.A. (Main Office Chicago)), as Agent c/o J.P. Morgan Securities Inc. 270 Park Avenue, 10th Floor New York, New York 10017

The undersigned, Flowserve Corporation, hereby:

- (i) acknowledges, and consents to, the execution of that certain Amendment No. 1 (the "Amendment"), of even date herewith, among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources party thereto as the Financial Institutions, Jupiter Securitization Corporation (together with the Financial Institutions, the "Purchasers"), J. P. Morgan Securities Inc. and JPMorgan Chase Bank, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as agent;
- (ii) reaffirms all of its obligations under that certain Performance Undertaking dated as of October 7, 2004 (the "Performance Undertaking") made by the undersigned in favor of the Agent and the Purchasers; and
- (iii) acknowledges and agrees that, after giving effect to the Amendment, such Performance Undertaking remains in full force and effect and is hereby ratified and confirmed.

FLO	WSERVE CORPORATION
By:	
	Name:
	Title:

AMENDMENT NO . 2

This AMENDMENT NO. 2 (this "Amendment"), is dated as of January 14, 2005, and is entered into by and among FLOWSERVE RECEIVABLES CORPORATION, a Delaware corporation (the "Seller"), FLOWSERVE US INC., a Delaware corporation, as the Servicer (the "Servicer"), the funding sources party hereto as the financial institutions (the "Financial Institutions"), JUPITER SECURITIZATION CORPORATION (together with the Financial Institutions, the "Purchasers"), and JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as agent (the "Agent") for the Purchasers.

WITNESSETH:

WHEREAS, the Seller, the Servicer, the Purchasers and the Agent are parties to that certain Receivables Purchase Agreement, dated as of October 7, 2004 (as amended, restated, supplemented or otherwise modified from time to time, the "Agreement"); and

WHEREAS the parties hereto desire to amend the Agreement on the terms and conditions set forth below;

NOW THEREFORE, in consideration of the premises herein contained, and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>Defined Terms</u>. Capitalized terms used and not otherwise defined herein shall have the meanings assigned to such terms in the Agreement.
- 2. <u>Amendment to the Agreement</u>. Subject to the satisfaction of the conditions precedent set forth in <u>Section 5</u> below, clause (i)(a) to the definition of the term "Eligible Receivable" set forth in Exhibit I to the Agreement is hereby amended and restated in its entirety to read as follows:

"(i) (a) if a natural person, is a resident of the United States or, if a corporation or other business organization, is organized under the laws of the United States or any political subdivision thereof and has its chief executive office in the United States;"

- 3. <u>Representations and Warranties of the Seller</u>. In order to induce the parties hereto to enter into this Amendment, the Seller represents and warrants that:
 - (a) The representations and warranties of Seller set forth in Section 5.1 of the Agreement are true, correct and complete on the date hereof as if made on and as of the date hereof and there exists no Amortization Event or Potential Amortization Event on the date hereof, provided that in the case of any representation or warranty in Section 5.1 of the Agreement that expressly relates to facts in existence on an earlier date, the reaffirmation thereof under this Section 3(a) shall be made as of such earlier date.

- (b) The execution and delivery by the Seller of this Amendment has been duly authorized by proper corporate proceedings of the Seller and this Amendment, and the Agreements, as amended by this Amendment, constitute the legal, valid and binding obligations of the Seller, enforceable against the Seller in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights generally.
- 4. <u>Representations and Warranties of the Servicer</u>. In order to induce the parties hereto to enter into this Amendment, the Servicer represents and warrants that:
 - (a) The representations and warranties of the Servicer set forth in Section 5.1 of the Agreement are true, correct and complete on the date hereof as if made on and as of the date hereof and there exists no Amortization Event or Potential Amortization Event on the date hereof, provided that in the case of any representation or warranty in Section 5.1 of the Agreement that expressly relates to facts in existence on an earlier date, the reaffirmation thereof under this Section 4(a) shall be made as of such earlier date.
 - (b) The execution and delivery by the Servicer of this Amendment has been duly authorized by proper corporate proceedings of the Servicer and this Amendment, and the Agreement, as amended by this Amendment, constitute the legal, valid and binding obligations of the Servicer, enforceable against the Servicer in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights generally.
- 5. <u>Conditions Precedent.</u> The amendment to the Agreement provided for hereunder shall become effective as of the date first written above upon the Agent's receipt of each of the following:
 - (a) counterparts of this Amendment executed by the Seller, the Servicer and each Purchaser; and
 - (b) a Reaffirmation of Performance Undertaking in substantially the form attached hereto as <u>Exhibit A</u> hereto, executed by the Provider in respect of the Performance Undertaking.
 - 6. Ratification. The Agreement, as amended hereby, is hereby ratified, approved and confirmed in all respects.
- 7. <u>Reference to Agreement</u>. From and after the effective date hereof, each reference in the Agreement to "this Agreement", "hereof", or "hereunder" or words of like import, and all references to the Agreement in any and all agreements, instruments, documents, notes, certificates and other writings of every kind and nature shall be deemed to mean the Agreement as amended by this Amendment.

8. <u>Costs and Expenses</u>. The Seller agrees to pay all reasonable costs, fees and out-of-pocket expenses (including attorneys' fees and time charges of attorneys representing the Agent, which attorneys may be employees of the Agent) incurred by the Agent in connection with the preparation, execution and enforcement of this Amendment.

9. CHOICE OF LAW, THIS AMENDMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS.

10. <u>Execution of Counterparts</u>. This Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered by their duly authorized officers as of the date first written above:

FLOWSERVE RECEIVABLES CORPORATION, as Seller

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

FLOWSERVE US INC.,

as Servicer

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

Signature Page to Amendment No. 2

JUPITER SECURITIZATION CORPORATION

By: /s/ Leo Loughead

Name: Leo Loughead Title: Authorized Signer

JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as a Financial Institution and as Agent

`

By: <u>/s/ Leo Loughead</u> Name: Leo Loughead

Title: Managing Director, Capital Markets

Signature Page to Amendment No. 2

EXHIBIT A

FORM OF REAFFIRMATION OF	PERFORMANCE UNDERTAKING

(attached)

REAFFIRMATION OF PERFORMANCE UNDERTAKING

As of January 14, 2005

JPMorgan Chase Bank, N.A. (successor by merger to Bank One, N.A. (Main Office Chicago)), as Agent c/o J.P. Morgan Securities Inc. 270 Park Avenue, 10th Floor New York, New York 10017

The undersigned, Flowserve Corporation, hereby:

- (i) acknowledges, and consents to, the execution of that certain Amendment No. 2 (the "Amendment"), of even date herewith, among Flowserve Receivables Corporation, Flowserve US Inc., the funding sources party thereto as the Financial Institutions, Jupiter Securitization Corporation (together with the Financial Institutions, the "Purchasers") and JPMorgan Chase Bank, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as agent;
- (ii) reaffirms all of its obligations under that certain Performance Undertaking dated as of October 7, 2004 (the "Performance Undertaking") made by the undersigned in favor of the Agent and the Purchasers; and
- (iii) acknowledges and agrees that, after giving effect to the Amendment, such Performance Undertaking remains in full force and effect and is hereby ratified and confirmed.

FLOV	VSERVE CORPORATION
By:	
	Name:
	Title:

LIMITED WAIVER AND AMENDMENT NO. 3

This LIMITED WAIVER AND AMENDMENT NO. 3 (this "Amendment"), is dated as of March 15, 2005, and is entered into by and among FLOWSERVE RECEIVABLES CORPORATION, a Delaware corporation ("SPV"), FLOWSERVE US INC., a Delaware corporation ("Flowserve"), FLOWSERVE CORPORATION, a New York corporation (the "Provider" and together with SPV and Flowserve, the "Flowserve Entities"), the funding sources party hereto as the financial institutions (the "Financial Institutions"), JUPITER SECURITIZATION CORPORATION (together with the Financial Institutions, the "Purchasers"), and JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as agent (the "Agent") for the Purchasers.

WITNESSETH:

WHEREAS, SPV, Flowserve, the Purchasers and the Agent are parties to that certain Receivables Purchase Agreement, dated as of October 7, 2004 (as amended, restated, supplemented or otherwise modified from time to time, the "RPA");

WHEREAS, SPV and Flowserve are parties to that certain Receivables Sale Agreement, dated as of October 7, 2004 (as amended, restated, supplemented or otherwise modified from time to time, the "RSA");

WHEREAS, the Provider executed a Performance Undertaking dated as of October 7, 2004 in favor of SPV (as amended, restated, supplemented or otherwise modified from time to time, the "Performance Undertaking" and together with the RSA and the RPA, the "Agreements");

WHEREAS, pursuant to (a) Section 7.1(a)(i)(A) of the RPA, SPV is required to deliver audited financial statements of the Provider and various other accompanying documents within 100 days after the end of each fiscal year of the Provider, (b) Section 4.1(a)(i)(A) of the RSA, Flowserve is required to deliver audited financial statements of the Provider and various other accompanying documents within 100 days after the end of each fiscal year of the Provider and (c) Section 7(a)(1) of the Performance Undertaking, the Provider is required to deliver audited financial statements of the Provider and various other accompanying documents within 100 days after the end of each fiscal year of the Provider;

WHEREAS, the Flowserve Entities have requested that the Agent and the Purchasers (as assignees of SPV's interests under the Performance Undertaking and the RSA) waive any noncompliance with such requirements under the Agreements with respect to the fiscal year ended December 31, 2004 until September 30, 2005 and any consequences resulting from such noncompliance during such period;

WHEREAS, the Agent and the Purchasers have agreed to grant such waivers on the terms and conditions set forth herein; and

WHEREAS, SPV, Flowserve, the Purchasers and the Agent desire to amend the RPA on the terms and conditions set forth below;

NOW THEREFORE, in consideration of the premises herein contained, and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto hereby agree as follows:

1. Defined Terms. Capitalized terms used and not otherwise defined herein shall have the meanings assigned to such terms in the RPA.

2. Limited Waiver.

(a) Subject to the terms and conditions set forth herein and in reliance on the representations and warranties of the Flowserve Entities herein contained, the Agent and the Purchasers hereby temporarily waive (i) compliance with the requirement (A) under Section 7.1(a)(i)(A) of the RPA that the financial statements of the Provider for the fiscal year ended December 31, 2004 delivered pursuant to Section 7.1(a)(i)(A) of the RPA are audited and accompanied by an opinion of Pricewaterhouse Coopers LLP, (B) under Section 4.1(a)(i)(A) of the RSA that the financial statements of the Provider for the fiscal year ended December 31, 2004 delivered pursuant to Section 4.1(a)(i)(A) of the RSA are audited and accompanied by an opinion of PricewaterhouseCoopers LLP and (C) under Section 7(a)(1) of the Performance Undertaking that the financial statements for the fiscal year ended December 31, 2004 delivered pursuant to Section 7(a)(1) of the Performance Undertaking are audited and accompanied by an opinion of PricewaterhouseCoopers LLP, (ii) any consequences or further affirmative obligations of any Flowserve Entity under any Agreement resulting from such noncompliance, and (iii) any Amortization Event, Potential Amortization Event, Termination Event or Potential Termination Event arising from an event of default under any agreement or instrument evidencing or governing Material Indebtedness of any Flowserve Entity or any of their respective Subsidiaries arising solely from the failure to provide similar financial information (or opinions or certifications thereof) or annual or quarterly public filings during or for the fiscal year ended December 31, 2004 or for the first three fiscal quarters of the fiscal year ending December 31, 2005; in each case, during the period from the date all the conditions precedent to this Amendment under Section 5 have been satisfied through and including the earlier to occur of (a) the date of delivery of audited financial statements for the fiscal year ended December 31, 2004 and accompanying independent public accountant's opinion in accordance with Section 7.1(a)(i)(A) of the RPA, Section 4.1(a)(i)(A) of the RSA and Section 7(a)(1) of the Performance Undertaking to the Agent, and (b) September 30, 2005; provided, however, all waivers and accommodations made to the Flowserve Entities in this Section 2(a) shall be rescinded and be null and void and of no force and effect upon the commencement by any holder of Material Indebtedness of any Flowserve Entity or any their respective Subsidiaries of any action exercising rights with respect to collateral or rights to accelerate arising from any default or event of default under any agreement or instrument evidencing or governing Material Indebtedness any Flowserve Entity or any their respective Subsidiaries.

- (b) The waivers set forth in <u>Section 2(a)</u> (the "<u>Waiver</u>") shall be limited precisely as written and relate solely to the noncompliance or temporary noncompliance, as the case may be, by the Flowserve Entities with the provisions of the Agreements in the manner and to the extent described above, and nothing in this Waiver shall be deemed to:
 - 1. constitute a waiver of compliance by SPV with respect to (i) Section 7.1(a)(i)(A) of the RPA in any other instance or (ii) any other term, provision or condition of the RPA or any other Transaction Document;
 - 2. constitute a waiver of compliance by Flowserve with respect to (i) Section 4.1(a)(i)(A) of the RSA in any other instance or (ii) any other term, provision or condition of the RSA or any other Transaction Document;
 - 3. constitute a waiver of compliance by the Provider with respect to (i) Section 7(a)(1) of the Performance Undertaking in any other instance or (ii) any other term, provision or condition of the Performance Undertaking or any other Transaction Document; or
 - 4. prejudice any right or remedy that SPV, the Agent or any Purchaser may now have (except to the extent such right or remedy was based upon existing defaults that will not exist after giving effect to this Waiver) or may have in the future under or in connection with any Agreement or any other Transaction Document.

Except as expressly set forth herein, the terms, provisions and conditions of each Agreement shall remain in full force and effect and in all other respects are hereby ratified and confirmed.

- 3. Amendment to the RPA. Subject to the satisfaction of the conditions precedent set forth in Section 5 below, Section 9.1(1) of the RPA is hereby amended and restated in its entirety to read as follows:
 - "(1) The Reporting Completion Date with respect to (1) the Oracle system of the Originator shall not have occurred by March 31, 2005 or (2) the Epic system of the Originator shall not have occurred by May 31, 2005."
 - 4. Representations and Warranties. Each of SPV, Flowserve and the Provider represents and warrants, as to itself only and not as to the others, that:
- (a) The representations and warranties of the Flowserve Entities set forth in Agreements (other than the representation and warranty of the Provider set forth in first sentence of Section 6(d) of the Performance Undertaking with respect to the June 30, 2004 financial statements of the Provider and its consolidated Subsidiaries) are true, correct and complete on the date hereof as if made on and as of the date hereof and there exists no Amortization Event, Potential Amortization Event, Termination Event or Potential Termination Event on the date hereof, provided, that, in the case of any representation or warranty in any Agreement that

expressly relates to facts in existence on an earlier date, the reaffirmation thereof under this Section 4(a) shall be made as of such earlier date.

- (b) The execution and delivery by SPV, Flowserve and the Provider of this Amendment has been duly authorized by proper corporate proceedings of SPV, Flowserve and the Provider and each of this Amendment and each Agreement, as amended by this Amendment, constitutes the legal, valid and binding obligation of SPV, Flowserve and the Provider, as applicable, enforceable against such Person in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws of general applicability affecting the enforcement of creditors' rights generally.
 - 5. Conditions Precedent. This Amendment shall become effective as of the date first written above upon the Agent's receipt of each of the following:
 - (a) counterparts of this Amendment executed by SPV, the Provider, Flowserve and each Purchaser; and
- (b) an amendment fee in an amount equal to \$15,000 in immediately available funds (which shall be fully earned and non-refundable as of the date paid).
 - 6. Ratification. Each Agreement (in the case of the RPA, as amended hereby) is hereby ratified, approved and confirmed in all respects.
- 7. <u>Reference to the Agreements</u>. From and after the effective date hereof, each reference any Agreement to "this Agreement", "hereof", or "hereunder" or words of like import, and all references to such Agreement in any and all agreements, instruments, documents, notes, certificates and other writings of every kind and nature shall be deemed to mean such Agreement as amended by this Amendment.
- 8. <u>Costs and Expenses</u>. Each Flowserve Entity agrees to pay all reasonable costs, fees and out-of-pocket expenses (including attorneys' fees and time charges of attorneys representing the Agent, which attorneys may be employees of the Agent) incurred by the Agent in connection with the preparation, execution and enforcement of this Amendment.

9. <u>CHOICE OF LAW.</u> THIS AMENDMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS.

- 10. Execution of Counterparts. This Amendment may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.
- 11. <u>Reaffirmation of Performance Undertaking</u>. The Provider consents to the execution by SPV and Flowserve of this Amendment, reaffirms all of its obligations under the

Performance Undertaking and acknowledges and agrees that, after giving effect to this Amendment, the Performance Undertaking remains in full force and effect and is hereby ratified and confirmed.

12. Delivery of Financial Statements. Each Flowserve Entity hereby agrees to provide to the Agent within 100 days after the end of the fiscal year of the Provider ended December 31, 2004, unaudited consolidated balance sheet and related statements of income, stockholders' equity and cash flows of the Provider showing the financial condition of the Provider and its consolidated Subsidiaries as of the close of the fiscal year ended December 31, 2004 and the results of its operations and the operations of such Subsidiaries during such year. SPV agrees that such financial statements shall be accompanied with a certificate of an Authorized Officer of SPV certifying such statements and certifying that no Amortization Event or Potential Amortization Event, has occurred or, if such an Amortization Event or Potential Amortization Event has occurred, specifying the nature and extent thereof and any corrective action taken or proposed to be taken with respect thereto. Flowserve agrees that such financial statements shall be accompanied with a certificate of an Authorized Officer of Flowserve certifying such statements and certifying that no Termination Event or Potential Termination Event, has occurred or, if such a Termination Event or Potential Termination Event, has occurred or, proposed to be taken with respect thereto.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed and delivered by their duly authorized officers as of the date first written above:

FLOWSERVE RECEIVABLES CORPORATION

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

FLOWSERVE US INC.

By: /s/ John M. Nanos

Name: John M. Nanos Title: Vice President

FLOWSERVE CORPORATION

By: /s/ John M. Nanos

Name: John M. Nanos Title: Assistant Secretary

JUPITER SECURITIZATION CORPORATION

By: /s/ Leo Loughead

Name: Leo Loughead Title: Authorized Signer

JPMORGAN CHASE BANK, N.A. (successor by merger to Bank One, NA (Main Office Chicago)), as a Financial Institution and as Agent

By: /s/ Leo Loughead

Name: Leo Loughead

Title: Managing Director, Capital Markets

Signature Page to Limited Waiver and Amendment No. 3

EUROPEAN INVESTMENT BANK

FI N° Agora N° 2003 0403

FLOWSERVE PRODUCT DEVELOPMENT PROJECT (GB)

FINANCE CONTRACT

between

EUROPEAN INVESTMENT BANK

and

FLOWSERVE CORPORATION

and

FLOWSERVE B.V.

2004

THIS CONTRACT IS MADE BETWEEN:

the European Investment Bank, having its seat in Luxembourg, represented by Thomas C. Barrett, Director and Alfonso Querejeta, Director

hereinafter called: "THE BANK",

of the first part,

and

Flowserve Corporation, a limited liability corporation incorporated under New York law, and having its registered office at CT Corporation System, 1633 Broadway, New York, NY 10019, represented by Renee J. Hornbaker, Vice President

(Authority of Signatory: Annex IA)

and

Flowserve B.V., a limited liability company incorporated under the laws of the Netherlands, and having its principal place of business at Parallelweg 6, 4878 AH Etten-Leur, The Netherlands, and registered under number 20052878, represented by Jan van Rens, Financial Director

(Authority of Signatory: Annex IB)

Etten Leur, The Netherlands

Irving, Texas USA

April 14, 2004

14 April 2004

hereinafter called respectively "Flowserve Corporation" and "Flowserve

of the second part.

B.V." and together "THE BORROWERS",

WHEREAS:

- Flowserve Corporation is presently carrying out a project consisting of research and development over the years 2003-2007 as more particularly described in the Technical Description contained in Schedule A and hereinafter called "the Project";
- the total cost of the Project is estimated by THE BANK at EUR 155.5 million, to be financed as follows:

	million EUK
— own resources and third party loans	85.5
— EIB loan	70.0
total:	155.5

...:111: ELID

- in order to enable it to implement this finance plan, THE BORROWERS have requested a loan from THE BANK in a total amount equivalent to EUR 70 million;
- the obligations of THE BANK under this Contract are conditional upon the prior execution, by a bank or syndicate of first-class banking institutions acceptable to THE BANK (such guarantor or guarantors are hereinafter collectively referred to as the "Guarantor"), of a guarantee agreement or agreements under which the Guarantor will guarantee the financial obligations of THE BORROWERS under each Tranche of this Contract (the "Guarantee");
- THE BANK, being satisfied that the financing of the Project falls within the scope of its functions and having regard to the matters herein recited, has decided to accede to the request of THE BORROWERS by opening in their favour under this Finance Contract (hereinafter called the "Contract") a credit in an amount of EUR 70 million (seventy million euro);
- the Board of Directors of Flowserve Corporation has authorised its entering into this Contract in the terms set out in Annex IIA and the Board of Directors of Flowserve B.V. has authorised its entering into this Contract in the terms set out in Annex IIB;
- THE BORROWERS have each delivered an opinion of their respective legal counsel in the terms set out in Annex III;
- references herein to Articles, Paragraphs, Articles, Recitals and Schedules are references respectively to articles, paragraphs and sub-paragraphs of, and recitals and schedules to, this Contract and the following terms have the respective meanings assigned to them in the following Recitals and Articles:

Term	Article
Affected Guarantor	7.02
Contract	fifth Recital
Credit	1.01

Term	Article
Disbursement Notice	1.02C
Disbursement Request	1.02A
EIB Redeployment Rate	4.02B
EUR	1.03
Expiry Date	4.03B
Fixed Rate Tranche	1.02B(b)
FSFR	3.01D
FSFR Conversion Date	3.01D(i)
FSFR Reference Period	3.01D
FSFR Tranche	1.02B(b)
GBP	1.03
Guarantee	fourth Recital
Guarantor	fourth Recital
Loan	2.01
Luxembourg Business Day	1.02C
Material Adverse Change	1.04(ii)
Maturity Date	3.01B(1)(ii)
Notice Day	3.01B(1)(11) 3.01B(2)(ii)(a)
Notice Day	* * * * * * * * * * * * * * * * * * * *
Notified	3.01B(2)(ii)(a) 1.06
Payment Date	5.03A
Prepayment Amount	4.02A
Prepayment Date	4.02A
Prepayment Notice	4.02A
Project	first Recital
Qualifying Guarantor	7.01
Quarter Day	3.01C
Reference Period	3.01B(1)
Relevant Business Day	5.03B
Relevant Interbank Rate	1.02B
Renewal Date	4.03B
Revisable Rate Tranche	1.02B(b)
Revision Date	3.01B(2)(ii)(a)
Revision Notice	3.01B(2)(ii)(a)
Scheduled Disbursement Date	1.02C(f)
Term Loan	4.03D
Test Date	3.01B(4)
Tranche	1.02A
VSFR	3.01C
VSFR Conversion	3.01C(i)
VSFR Conversion Date	3.01C(iii)
VSFR Conversion Option Date	3.01C(i)
VSFR Conversion Request	3.01C(i)
VSFR Converted Tranche	3.01C(iii)
VSFR Proposal	3.01C(ii)
VSFR Reference Period	3.01C
VSFR Tranche	1.02B(b)
USD	1.03

NOW THEREFORE it is hereby agreed as follows :

ARTICLE 1 Disbursement

1.01 Amount of Credit and Joint and Several Liability of THE BORROWERS

By this Contract THE BANK establishes in favour of THE BORROWERS, and THE BORROWERS accept, a credit (hereinafter the "Credit") in an amount equivalent to EUR 70 000 000 (seventy million euro), to be used for the sole purpose of financing the Project.

The liability of THE BORROWERS to make any and all payments under this Contract shall be joint and several.

1.02 Disbursement Procedures

- A. Until 18 months from the signature date THE BORROWERS or either of them may deliver to THE BANK up to five written requests (each hereinafter a "Disbursement Request") for disbursement of any of the Credit (hereinafter a "Tranche"). For the avoidance of doubt, THE BORROWERS shall not deliver more than five Disbursement Requests between them.
- B. Disbursement of each Tranche shall be subject to receipt by THE BANK of a Request, signed by a duly authorised officer of each of THE BORROWERS or of the relevant BORROWER (as the case may be), specifying:
 - (a) the amount of the Tranche to be disbursed, such amount shall be (save for the last Tranche to be disbursed under this Contract) at least EUR 10 000 000 of the Credit established hereunder;
 - (b) whether the Tranche is to bear a fixed rate of interest determined pursuant to Article 3.01A (such a Tranche being referred to as a "Fixed-Rate Tranche") or pursuant to Article 3.01B (such a Tranche being referred to as a "Revisable-Rate Tranche"), or a floating rate of interest determined pursuant to Article 3.01C (a "VSFR Tranche") or determined pursuant to Article 3.01D (a "FSFR Tranche");
 - (c) the currency in which THE BORROWERS prefer the Tranche to be disbursed, being a currency referred to in Paragraph 1.03 as being available for that type of Tranche;
 - (d) in case of a Fixed-Rate or a Revisable-Rate Tranche, the annual or semi-annual Payment Dates preferred by THE BORROWERS chosen in accordance with Article 5.03A;
 - (e) THE BORROWERS' preferred repayment terms and repayment schedule, chosen in accordance with Article 4.01; and
 - (f) the preferred date for disbursement, it being understood that THE BANK may disburse any Tranche up to four calendar months from the date of the relevant Disbursement Request.

THE BORROWERS or the relevant BORROWER (as the case may be) may also at their/its discretion specify in the Disbursement Request the following respective elements, if any, indicated by THE BANK without commitment to be applicable to each currency of the Tranche during its lifetime, that is to say:

- (i) in the case of a Fixed-Rate Tranche (and in case of a Revisable-Rate Tranche, until the Revision Date), the interest rate or rates; and
- (ii) in the case of a FSFR Tranche, the spread to the Relevant Interbank Rate.

For the purposes of this Contract generally, "Relevant Interbank Rate" means EURIBOR (as defined in Schedule B) or such other reference rate as appropriate for three months' deposits in the currency of the tranche and determined in the manner generally applied by THE BANK from time to time in its loan operations.

Each of THE BORROWERS or the relevant BORROWER (as the case may be) shall provide to THE BANK evidence that the person or persons making Disbursement Requests hereunder have been duly authorised to do so. Subject to the proviso to Article 1.02C, each Disbursement Request is irrevocable.

- C. Between 10 and 15 days before the date of disbursement THE BANK shall, if the Request conforms to Article 1.02B, and subject to Article 1.05, deliver to Flowserve Corporation a notice (hereinafter a "Disbursement Notice") which shall:
 - (a) confirm the amount of the Tranche to be disbursed as specified in the Request;
 - (b) specify the fixed or revisable interest rate or, as the case may be, specify whether the Tranche is a VSFR Tranche or a FSFR Tranche and, if the latter, the spread to the Relevant Interbank Rate applicable during the lifetime of the Tranche or until the FSFR Conversion Option Date (as defined in Article 3.01D), as the case may be;
 - (c) in relation of any FSFR Tranche, specify the FSFR Conversion Option Date, if any;
 - (d) specify the currency of the Tranche specified in the Disbursement Request;
 - (e) specify the applicable Payment Date(s);
 - (f) specify the repayment terms applicable thereto in accordance with Article 4.01; and
 - (f) specify the date on which the Tranche is scheduled to be disbursed (hereinafter the "Scheduled Disbursement Date"), disbursement being subject to the conditions of Article 1.03.

Provided that, if one or more of the elements specified in the Disbursement Notice does not conform to the corresponding element, if any, in the Disbursement Request, THE BORROWERS or the relevant BORROWER (as the case may be) may within three Luxembourg Business Days following receipt of the Disbursement Notice revoke the Disbursement Request by notice to THE BANK and thereupon the Disbursement Request and the Disbursement Notice shall be of no effect.

For the purposes of this Contract generally, "Luxembourg Business Day" means a day on which commercial banks are open for business in Luxembourg.

Disbursement shall be made to such bank account of either of THE BORROWERS as THE BORROWERS or the relevant BORROWER (as the case may be) shall specify in a notification to THE BANK, signed by a duly authorised officer of such, not less than fifteen days before the date of disbursement. Only one account may be specified for each Tranche.

1.03 Currency of Disbursement

Subject to availability, THE BANK shall disburse each Tranche in the currency for which THE BORROWERS or the relevant BORROWER (as the case may be) has expressed a preference. Each currency of disbursement shall be the euro ("EUR"), pounds Sterling ("GBP") United States dollars ("USD").

For the calculation of sums to be disbursed in currencies other than the euro, THE BANK shall apply the rate published by the European Central Bank in Frankfurt am Main, on such date within fifteen days preceding disbursement as THE BANK shall decide.

1.04 Conditions of Disbursement

Disbursement of each Tranche pursuant to Article 1.02 shall be subject to the fulfilment of the following conditions to the satisfaction of THE BANK:

- (i) the condition mentioned in Article 7.01 at least ten Luxembourg Business Days prior to the date fixed for disbursement;
- (ii) there shall not have occurred any Material Adverse Change in relation to either of THE BORROWERS since the date of this Contract and THE BORROWERS shall have provided confirmation in writing to this effect to THE BANK as of a date falling no earlier than fifteen days and not later than five days before the date of disbursement; provided that THE BANK shall act reasonably in determining the occurrence or non-occurrence of a Material Adverse Change; for the purpose of this subparagraph (ii), "Material Adverse Change" means any event or condition which materially impairs the relevant BORROWER's ability to perform its financial and other obligations under this Finance Contract; and
- (iii) no event described in Article 4.03C or 10.01 shall have occurred and be continuing in respect of either of THE BORROWERS.

1.05 Deferment of Disbursement

- A. THE BANK shall, at the request of THE BORROWERS or the relevant BORROWER (as the case may be), defer disbursement of any Tranche in whole or in part to a date specified by THE BORROWERS or the relevant BORROWER (as the case may be) being a date falling not more than six months from its Scheduled Disbursement Date. In such a case THE BORROWERS shall pay deferment indemnity as determined pursuant to Article 1.05C below. Any request for deferment shall have effect in respect of a Tranche only if it is made at least five Luxembourg Business Days before its Scheduled Disbursement Date.
- B. If any of the conditions referred to in Article 1.04 are not fulfilled as of the specified date, disbursement will be deferred to a date agreed between THE BANK and THE BORROWERS or the relevant BORROWER (as the case may be) falling not less than five Luxembourg Business Days following the fulfilment of all conditions of disbursement.
- C. If the disbursement of any Tranche is deferred whether on request of THE BORROWERS or the relevant BORROWER (as the case may be) or by reason of non-fulfilment of any of the conditions of disbursement, THE BORROWERS shall, upon demand by THE BANK, pay an indemnity on the amount of which disbursement is deferred. Such indemnity shall accrue from the Scheduled Disbursement Date to the

actual disbursement date or, as the case may be, until the date of cancellation of the Tranche at a rate equal to R₁ minus R₂, where:

R₁ means the rate of interest that would have applied from time to time pursuant to Article 3.01 and the relevant Disbursement Notice, if the Tranche had been disbursed on the Scheduled Disbursement Date:

and

R₂ means the Relevant Interbank Rate less 0.125% (12.5 basis points); provided that for the purpose of determining the Relevant Interbank Rate in relation to this Article 1.05, the relevant periods provided for in Schedule B shall be successive periods of one month commencing on the Scheduled Disbursement Date.

Furthermore, the indemnity:

- (i) if the deferment exceeds one month in duration, shall accrue at the end of every month;
- (ii) shall be calculated using the day count convention applicable to \mathbf{R}_1 ; and
- (iii) where R_2 exceeds R_1 , shall be set at zero.

1.06 Cancellation and Suspension of Credit

A. THE BORROWERS or either of them may at any time, by notice given to THE BANK cancel, in whole or in part, and with immediate effect, the undisbursed portion of the Credit. However, the notice shall have no effect on a Tranche whose current Scheduled Disbursement Date falls within five Luxembourg Business Days following the date of the notice. THE BORROWERS shall, save as stated in the next paragraph, incur no liability by way of indemnity or otherwise to THE BANK by reason of such cancellation.

If THE BORROWERS or either of them cancel a Tranche that has been the subject of a Disbursement Notice (such a Tranche, whatever its interest rate basis, being referred to as "Notified"), THE BORROWERS shall, if the Tranche is a Fixed-Rate Tranche, a Revisable-Rate Tranche or a FSFR Tranche, pay an indemnity on the cancelled amount. Such indemnity shall be calculated, in accordance with Article 4.02B and C, in respect of the period running from the date (the "Starting Date"), being the later of (a) the date of the cancellation notice and (b) the Scheduled Disbursement Date of the Tranche, until its scheduled final repayment date. For this purpose (i) references in Article 4.02B and C to the Prepayment Date shall refer to Starting Date and (ii) the cancelled amount shall be deemed to be disbursed and repaid upon the Starting Date. The cancellation indemnity shall be payable within seven days of demand.

- B. THE BANK may, by notice to THE BORROWERS or either of them, in whole or in part suspend the undisbursed portion of the Credit at any time, and with immediate effect:
 - (a) under the conditions mentioned in Article 10.01 or Article 4.03;

or

(b) if exceptional circumstances shall arise which adversely affect THE BANK's access to relevant international capital markets.

Such suspension shall continue until THE BANK notifies THE BORROWERS or either of them that it may again draw on the Credit or, as the case may be, until THE BANK cancels the suspended amount as provided below.

Furthermore, THE BANK may, by notice to THE BORROWERS or either of them, in whole or in part cancel the undisbursed portion of the Credit under the conditions of indents (a) and (b) of Article 1.06B above, subject always (in the case of Article 4.03) to fulfillment of the respective procedures specified in each paragraph of Article 4.03. THE BANK may also, by notice to THE BORROWERS or either of them, cancel a Tranche the disbursement of which has been deferred under Article 1.05A or B by more than six months in aggregate.

However, THE BANK shall not be entitled to suspend or cancel, on grounds of paragraph (b) of Article 1.06B above, any Notified Tranche.

C. If THE BANK suspends a Notified Tranche on the sole ground of Article 1.06B indent (a), THE BORROWERS shall pay an indemnity on the suspended amount in the manner provided for in Article 1.05C.

If THE BANK cancels a Notified Fixed-Rate Tranche or a Revisable-Rate Tranche on the sole ground of the said indent (a) of Article 1.06B, THE BORROWERS shall pay an indemnity on the cancelled amount, calculated in accordance with Article 1.06A, provided that no indemnity is payable upon cancellation upon the sole ground of an event mentioned in Article 4.03(A).

If THE BANK cancels a Notified VSFR Tranche or a FSVR Tranche under the said indent (a) of Article 1.05B on the ground of an event mentioned in Article 10.01, THE BORROWER shall pay an indemnity on the cancelled amount calculated on the basis of Article 10.03(2) and accruing over the period running from the later of (a) the date of the cancellation notice and (b) the Scheduled Disbursement Date of the Tranche, until its final scheduled repayment date.

1.07 Sums Due under Article 1

Sums due under Article 1 shall be payable in euro or, if different, the currency of the Tranche concerned.

ARTICLE 2
The Loan

2.01 Amount of the Loan

The loan (hereinafter called "the Loan") shall comprise the aggregate of the amounts in the currencies disbursed by THE BANK hereunder, each such amount as notified by THE BANK upon the occasion of the disbursement of a Tranche.

2.02 Currency of Repayments

Each repayment of a Tranche under Article 4 or, as the case may be, Article 10 shall be effected in its currency of disbursement.

2.03 Currency of Interest and Other Charges

Interest and other charges payable on, or in respect of, a Tranche under Article 3.01, Article 4 or, where applicable, Article 10 shall be calculated and be payable in the currency in which such Trance is repayable.

Any other sum due hereunder shall be paid in the currency specified by THE BANK having regard to the currency of the expenditure to be reimbursed thereby.

2.04 Notification by THE BANK

After each disbursement of a Tranche, THE BANK shall deliver to THE BORROWERS or the relevant BORROWER (as the case may be) a summary statement showing the disbursement date, currency and amount, repayment terms and the interest rate of and for that Tranche. Such statement shall include the relevant amortisation table referred to in Article 4.01.

ARTICLE 3 Interest

3.01 Interest Rates

A. Fixed-Rate Tranches

THE BORROWERS shall pay interest on the outstanding balance of each Fixed-Rate Tranche semi-annually or annually in arrears on the relevant Payment Date specified in Article 5.03A or, if different, as specified in the relevant Disbursement Notice commencing on the first such Payment Date following the date of disbursement of the Tranche. Interest shall be calculated at the rate specified in the Disbursement Notice, which rate shall be the rate applicable at the date of the Disbursement Notice, in accordance with the applicable principles from time to time laid down by the governing bodies of THE BANK for loans made at a fixed rate of interest, denominated in the currency of the Tranche and bearing equivalent terms for the repayment of capital and the payment of interest.

B. Revisable-Rate Tranches

- (1) In respect of any Revisable-Rate Tranche, THE BORROWERS shall pay interest on the daily outstanding balance at a nominal fixed interest applicable to successive Reference Periods as determined pursuant to Article 3.01B(2) semi-annually or annually in arrears on the relevant Payment Date.
 - "Reference Period" means a period commencing on the date of disbursement of the relevant Tranche or the date of expiry of the immediately preceding relevant Reference Period and determined as follows:
 - (i) the first Reference Period shall commence on the date of disbursement of the relevant Tranche and shall expire on such date as is specified in the relevant Disbursement Notice being no earlier than a payment date (determined pursuant to Article 5.03A) specified in the relevant Disbursement Notice for the payment of interest on that Tranche, most

closely preceding the fourth anniversary of the date of drawdown of such Tranche and no later than four years prior to the final maturity of such Tranche; and

- (ii) each subsequent Reference Period shall begin immediately upon expiry of the immediately preceding Reference Period and shall expire on such date as is specified in the relevant notice of acceptance given by THE BORROWERS or the relevant BORROWER (as the case may be) pursuant to Article 3.01B(2)(ii)(b) below being a date falling at least four years after the date of expiry of the first Reference Period and being a date falling no later than a date (determined pursuant to Article 5.03A) specified in the relevant Disbursement Notice for the payment of interest on that Tranche, most closely preceding the tenth anniversary of the date of drawdown of such Tranche (in case of a bullet prepayment, the seventh anniversary of the date of drawdown of such Tranche) (such date being hereinafter referred as the "Maturity Date").
- (2) (i) interest on the relevant Tranche shall be payable at a fixed rate determined by THE BANK for the first Reference Period to be the rate applicable at the date of issue of the relevant Disbursement Notice to loans to be made by THE BANK denominated in the currency of the Tranche for the financing of projects located in the European territories of member countries of the European Community on the date of disbursement of the Tranche in question and having (i) a life equal to that of the Reference Period and (ii) during that life the same terms for repayment and the payment of interest as the Tranche in question. The rate of interest for the first Reference Period shall be specified in the relevant Disbursement Notice.
 - (ii) For each subsequent Reference Period the interest rate applicable to the relevant Tranche shall be determined as follows:
 - (a) before 12 noon (Luxembourg time) on any day on which THE BANK is open for business in Luxembourg (the "Notice Day") falling in the interval between 60 and 30 days (the "Notice Period") prior to the commencement of each subsequent Reference Period (the "Revision Date"), upon request of THE BORROWERS or the relevant BORROWER (as the case may be) within the Notice Period, and where reasonably practicable, THE BANK shall, subject to Article 3.01B(4), indicate by irrevocable notice (the "Revision Notice") the fixed interest rate applicable to the Tranche for each choice of duration of that Reference Period, as permitted by Article 3.01B(1)(ii), to the extent that funding is available to THE BANK for each duration;
 - (b) by not later than 17:00 Luxembourg time on the Notice Day, THE BORROWERS or the relevant BORROWER (as the case may be) shall indicate to THE BANK by irrevocable notice its acceptance or rejection of a rate indicated by THE BANK for one of the permitted durations; upon delivery of such notice of acceptance THE BORROWERS shall be obliged to pay interest at the specified rate for a Reference Period of the specified duration; failing timely delivery of an acceptance by THE BORROWERS or the relevant BORROWER (as the case may be) in due time of an interest rate indicated by THE BANK, the relevant offer shall lapse.

If on a date falling one month before the Revision Date in question, by 17:00 hours Luxembourg time, THE BANK has not received any notice from THE BORROWERS or the relevant BORROWER (as the case may be) indicating its acceptance of an interest rate applicable to the relevant Tranche from that Revision Date, THE BORROWERS shall be obliged to prepay the Tranche on the Revision Date in question.

(3) THE BORROWERS shall pay interest on Revisable-Rate Tranches annually or semi-annually in arrears on the date(s) specified in the Disbursement Notice and determined in accordance with Article 5.03A commencing on the first such date following the date of disbursement of the Tranche.

C. VSVR Tranches

THE BORROWERS shall pay interest on the outstanding balance of each VSFR Tranche at VSFR (as defined below) quarterly in arrears on the relevant Payment Date, subject to a ceiling of the Relevant Interbank Rate plus 13 basis points (0.13% p.a.), as determined for successive VSFR Reference Periods commencing on or after the first relevant Quarter Day; provided that, if such Tranche is not disbursed on a Quarter Day or, where a Quarter Day is not a Relevant Business Day, on the nearest succeeding day that is a Relevant Business Day, interest for the period from the date of disbursement to the first Quarter Day shall be payable at the Relevant Interbank Rate.

THE BANK shall notify VSFR or other applicable rate to THE BORROWERS or to the relevant BORROWER (as the case may be) within ten days following the commencement of the VSFR Reference Period to which it applies.

For the purpose of this Article 3.01C:

"VSFR" means a variable-spread floating interest rate, that is to say a general interest rate determined by THE BANK for each VSFR Reference Period in accordance with the applicable principles from time to time laid down by the governing bodies of THE BANK for loans made at a floating rate of interest, not defined by reference to the Relevant Interbank Rate.

"Quarter Day" means each 15th March, 15th June, 15th September and 15th December.

"VSFR Reference Period" means each period from, and including, one Quarter Day to, but excluding, the next following Quarter Day.

VSFR Conversion

Any Variable-Spread Floating-Rate Tranche, or a part thereof which is not less than an amount of EUR 10 million, may be converted at the request of THE BORROWERS or the relevant BORROWER (as the case may be) on the following conditions:

(i) Upon receipt by THE BANK of a Conversion Request from THE BORROWERS or the relevant BORROWER (as the case may be) on a Luxembourg Business Day falling not later than 75 days before a VSFR Conversion Option Date, (as defined below) after prior consultation with THE BORROWERS or the relevant BORROWER (as the case may be), and on condition that the necessary funds are available to THE BANK for this purpose, THE BANK shall, in the period between 60 and 30 days before the VSFR Conversion Option Date, propose to THE BORROWERS or the relevant BORROWER (as the case may be) the fixed rate of interest to be applicable to the Variable-Spread Floating-Rate Tranche (or relevant part thereof) for its remaining term and shall propose that

this rate shall apply as from the VSFR Conversion Option Date, with interest to be payable annually or semi-annually in arrear on the date or dates to be specified (such change of interest rate basis being referred to as "VSFR Conversion"); and such rate shall be the rate of interest generally applicable to loans denominated in the currency of the Variable-Spread Floating-Rate Tranche, as the case may be, and granted by THE BANK to borrowers in the EU on the same repayment terms and remaining life as the Tranche in question and on the same periodicity for the payment of interest as is to be contained in the VSFR Conversion Request.

- (ii) For the purposes of this paragraph, "VSFR Conversion Option Date" means any one of 15th March, 15th June, 15th September and 15th December falling not later than four years before the final repayment date of the relevant Variable-Spread Floating-Rate Tranche; and "VSFR Conversion Request" means a written notice from THE BORROWER requesting THE BANK to submit to it a proposal for converting to a fixed-rate basis the whole of the Variable-Spread Floating-Rate Tranche, or a part outstanding, which in either case is in an amount of not less than EUR 10 million.
 - Any such proposal by THE BANK (the "VSFR Proposal") shall be made subject to any amendments to this Contract which the BANK in its discretion considers reasonably necessary for the purpose of amending the applicable interest rate basis or causing the financial terms of the Contract to conform to THE BANK's standard terms of lending for fixed-rate loans at the date of the VSFR Proposal.
- (iii) Any notice by THE BORROWERS or the relevant BORROWER (as the case may be) accepting the VSFR Proposal in respect of any Variable-Spread Floating-Rate Tranche shall be served in writing on THE BANK by 5 p.m. (Luxembourg time) on the same day as its receipt of the VSFR Proposal; provided that THE BANK shall give notice to THE BORROWERS or the relevant BORROWER (as the case may be) as early as possible on the date of issue of the VSFR Proposal; the VSFR Conversion Option Date in respect of which a VSFR Proposal is accepted is hereinafter referred to as the "VSFR Conversion Date" and a Variable-Spread Floating-Rate Tranche, or the part thereof, subject to VSFR Conversion hereunder is hereinafter referred to as a "VSFR Converted Tranche" and in relation to Articles 4 and 5 and Paragraph 10.03, a Fixed-Rate Tranche.
- (iv) In the event of THE BORROWERS' or the relevant BORROWER's (as the case may be) acceptance of the VSFR Proposal, (a) any accrued interest on the Tranche, or part thereof, to be converted shall be payable on the VSFR Conversion Date; (b) interest on the VSFR Converted Tranche shall thereafter be payable semi-annually or annually on a date or dates to be agreed between THE BORROWERS or the relevant BORROWER (as the case may be) and THE BANK; and (c) any unconverted part of the Variable-Spread Floating-Rate Tranche shall continue to accrue interest and remain repayable on the same terms as those on which it was disbursed.
- (v) In the event of THE BORROWERS' or the relevant BORROWER's (as the case may be) refusal or non-acceptance of a VSFR Proposal or of such failing to make a written request under paragraph (i) above or of a failure by the relevant parties to reach the agreement mentioned in item (a) of sub-paragraph (iv) above, the Variable-Spread Floating-Rate interest basis as specified in this paragraph shall remain in full force and effect in respect of the balance of the Tranche outstanding.

D. FSFR Tanches

THE BORROWERS shall pay interest on the outstanding balance of each FSFR Tranche at FSFR (as defined below) quarterly in arrears on the relevant Payment Dates. For the purpose of this Article 3.01D:

"FSFR" means a fixed-spread floating interest rate, that is to say an annual percentage rate equal to the Relevant Interbank Rate plus or minus the Spread, calculated by THE BANK for each successive FSFR Reference Period. The rate of interest for each FSFR Reference Period shall be the rate so notified by THE BANK to THE BORROWERS or the relevant BORROWER (as the case may be) within 10 days following commencement of the FSFR Reference Period to which such rate applies.

"Spread" means such fixed spread to the Relevant Interbank Rate (being either plus or minus) determined by THE BANK and notified to THE BORROWERS or the relevant BORROWER (as the case may be) in the relevant Disbursement Notice.

"FSFR Reference Period" means each period of three months from, and including, one Payment Date to, but excluding, the next following Payment Date, provided that the first FSFR Reference Period shall commence on the date of disbursement of the Tranche and end on, but exclude, the first Payment Date applicable to that Tranche.

FSFR Conversion

The FSFR basis as specified in Sub-paragraph 3.01D will be modified on the following conditions:

- (i) Upon receipt by THE BANK of a notice from THE BORROWERS or the relevant BORROWER (as the case may be) on a Luxembourg Business Day falling not later than 75 days prior to the conversion date as specified in the Disbursement Notice (hereinafter called the "FSFR Conversion Date") and subject to availability of the necessary funds for such purpose (as determined by THE BANK at its discretion), THE BANK shall, subject to a request by THE BORROWERS or the relevant BORROWER (as the case may be), before 11.00 a.m. Luxembourg time on any Luxembourg Business Day falling in the period between 60 and 30 days prior to the FSFR Conversion Date, indicate by irrevocable notice before 12.00 noon Luxembourg time on the date of the request of THE BORROWERS or the relevant BORROWER (as the case may be), the fixed or VSFR interest rate then applicable to loans from THE BANK's own resources to be made on the FSFR Conversion Date in the currency of the Fixed-Spread Floating Rate Tranche and having (a) annual or semi-annual payment of interest, in case of a fixed rate of interest, and annual, semi-annual or quarterly payment of interest, in case of floating rate of interest, on dates specified in THE BORROWERS' or the relevant BORROWER's (as the case may be) request and (b) the repayment terms and remaining life of the Tranche as specified in such request.
- (ii) Any delivery to THE BANK of THE BORROWERS' or the relevant BORROWER's (as the case may be) irrevocable written acceptance shall be made not later than by 17:00 hours Luxembourg time on the day on which THE BANK made the relevant offer. Such acceptance shall reflect THE BANK's offer in respect of the currency, the annual date, semi-annual or quarterly dates, as the case may be, for payment of interest, the provision of a repayment schedule and the fixed interest rate or floating interest rate basis for the Tranche applicable to its final maturity date.

- (iii) In the event of THE BORROWERS' or the relevant BORROWER's (as the case may be) acceptance of THE BANK's offer, (a) any amendments to this Contract shall be effected by an agreement between THE BANK and THE BORROWERS, (b) any accrued interest shall be payable on the FSFR Conversion Date and (c) interest on the converted Tranche shall from and including the FSFR Conversion Date be payable annually, semi-annually or quarterly, as the case may be, on any date(s) mentioned for that purpose in the aforementioned offer, on the daily outstanding balance of the Tranche in the currency of the Tranche and at the rate specified in this offer.
 - Failing THE BORROWERS' or the relevant BORROWER's (as the case may be) timely acceptance of any offer made by THE BANK, it shall immediately lapse.
- (iv) If on a date falling one month before the FSFR Conversion Date by 17:00 hours Luxembourg time, THE BANK has not received any notice from THE BORROWERS or the relevant BORROWER (as the case may be) indicating its acceptance of an offer made by THE BANK pursuant to indent (ii) of this Sub-paragraph 3.01D, THE BORROWERS shall be obliged to prepay the Tranche on the FSFR Conversion Date, provided that if such a date is not a Relevant Business Day, the Tranche shall be paid on the nearest succeeding Relevant Business Day, unless such day falls in the following month, in which case it shall be payable on the nearest preceding Relevant Business Day.

For the purposes of this Contract, the Tranche shall be referred to as:

"Fixed-Rate Tranche" if its interest basis has been converted into fixed rate, pursuant to Sub-paragraph C or D of this Paragraph 3.01; and

"VSFR Tranche" if its interest basis has been converted into VSFR rate, pursuant to Sub-paragraph D of this Paragraph 3.01.

3.02 Interest on Overdue Sums

Without prejudice to Article 10 and by way of exception to Article 3.01, interest shall accrue on any overdue sum payable under the terms of this Contract from the due date to the date of payment at a rate equal to the Relevant Interbank Rate plus 2% (200 basis points). For the purpose of determining the Relevant Interbank Rate in relation to this Article 3.02, the relevant periods for the purposes of Schedule B shall be successive periods of one month commencing on the due date.

However, interest shall be charged at the rate defined in Article 3.01 plus 0.25% (25 basis points), if the latter rate exceeds, for any given relevant period, the rate specified in the preceding paragraph.

If the overdue sum is in a currency other than the currency of the Loan, the following rate shall apply namely the Relevant Interbank Rate (offer side) which is generally retained by THE BANK for transactions in that currency plus 2% (200 basis points) calculated in accordance with the market practice for such rate.

ARTICLE 4 Repayment

4.01 Normal Repayment

THE BORROWERS shall repay each Tranche by semi-annual or annual installments on the dates specified in the relevant Disbursement Notice in accordance with the terms of the amortisation table specified in the relevant statement mentioned in Article 2.04 and providing (subject to the proviso in the following paragraph): (1) in the case of Fixed-Rate Tranches and Revisable-Rate Tranches for repayment either (a) on a constant annuity basis so that the aggregate of principal and interest payable in respect of the Tranche shall be nearly as possible the same on each repayment date or (b) by equal installments of principal; or (2) in the case of FSFR Tranches and VSFR Tranches, for repayment by equal installments of principal.

Each amortisation table shall be drawn up on the basis that repayment of each Tranche shall begin not later than the first date in the year mentioned in the relevant Disbursement Notice falling after the third anniversary of disbursement of the Tranche and shall end on a date mentioned in the relevant Disbursement Notice falling after the elapse of not less than four years and not more than ten years from the date of disbursement; provided that, if so requested by THE BORROWERS or the relevant BORROWER (as the case may be), THE BANK may agree that a Tranche shall be repayable in a single installment on a date mentioned in the relevant Disbursement Notice falling not before three and not after seven years from the date of disbursement.

If necessary upon the determination (pursuant to Article 3.01B) of the interest rate applicable to a Revisable-Rate Tranche after its Revision Date, THE BANK shall deliver a modified repayment schedule to THE BORROWERS or the relevant BORROWER (as the case may be) which shall replace the schedule notified pursuant to Article 2.04.

4.02 Voluntary Prepayment

- A. Subject to paragraphs B and C below, THE BORROWERS may prepay all or part of any Tranche, together with accrued interest thereon, upon giving one month's prior written notice (hereafter a "Prepayment Notice") specifying the amount thereof to be prepaid (the "Prepayment Amount") and the date on which THE BORROWERS propose to effect prepayment (the "Prepayment Date"), which date shall be a Payment Date for that Tranche.
- B. In respect of each Prepayment Amount of a Fixed-Rate Tranche or Revisable-Rate Tranche, THE BORROWERS shall pay to the Bank on the Prepayment Date an indemnity equal to the present value (as of the Prepayment Date) of the excess, if any, of:
 - (X) the interest that would accrue thereafter on the Prepayment Amount from the Prepayment Date to its scheduled final repayment date or, in the case of a Revisable-Rate Tranche, that would accrue from the Prepayment Date to the next following Revision Date, if it were not prepaid, over
 - (Y) the interest that so would accrue if it were calculated at the EIB Redeployment Rate in effect one month prior to the Prepayment Date, less 15 Basis Points.

In this Contract, "EIB Redeployment Rate" means THE BANK's standard rate of interest for fixed-rate loans denominated in the currency of, and having the same debt service profile to the final maturity or, in the case of a Revisable-Rate Tranche, to the next Revision Date, as the Prepayment Amount.

The said present value shall be calculated at a discount rate equal to the EIB Redeployment Rate, applied as of each relevant Payment Date.

THE BORROWERS may prepay a Revisable-Rate Tranche without penalty on any Revision Date, save that an indemnity calculated in the manner provided above shall be due to THE BANK if the prepayment takes place after THE BORROWERS or the relevant BORROWER (as the case may be) have/has already notified the Bank that it accepts one of the interest rate options specified in a Revision Notice issued by THE BANK for the relevant Revision Date.

- C. In respect of each Prepayment Amount of a FSFR Tranche, THE BORROWERS shall pay to the Bank on the Prepayment Date an indemnity equal to the present value (as of the Prepayment Date) of the excess, if any, of:
 - (X) the spread to the Relevant Interbank Rate applicable to the FSFR Tranche under Article 3.01D from the Prepayment Date to its scheduled final repayment date, over
 - (Y) the spread to the Relevant Interbank Rate applicable to a loan that would be made by THE BANK on FSFR terms and having the same debt service profile to the final maturity as the unexpired term of the FSFR Tranche in question, on conditions prevailing one month prior to the Prepayment Date.

The said present value shall be calculated at a discount rate, applied as of each relevant Payment Date. The discount rate shall be THE BANK's standard rate of interest for fixed-rate loans denominated in the currency of, and having the same debt service profile to the final maturity as, the Prepayment Amount.

- D. THE BORROWERS may prepay a VSFR Tranche without penalty on any relevant Payment Date.
- E. THE BANK shall notify THE BORROWERS or the relevant BORROWER (as the case may be), not later than 15 days prior to the Prepayment Date, of the Prepayment Amount, of the interest due thereon, and, in case of prepayment of a Fixed-Rate Tranche or Revisable-Rate Tranche or FSFR-Rate Tranche, of the indemnity payable under paragraph B or C above or, as the case may be, that no indemnity is due.

Not later than 17:00 hours Luxembourg time on the day of receipt of such notice (or 17:00 hours on the following day if the notice was delivered later than 12:00 hours Luxembourg time), THE BORROWERS or the relevant BORROWER (as the case may be) shall notify the Bank either:

- (i) that it confirms the Prepayment Notice on the terms specified by THE BANK; or
- (ii) that it withdraws the Prepayment Notice.

If THE BORROWERS or the relevant BORROWER (as the case may be) give the confirmation under (i), THE BORROWERS shall effect the prepayment. If THE BORROWERS or the relevant BORROWER (as the case may be) withdraw the Prepayment Notice or fails to confirm in due time, it may not effect the prepayment. Save as aforesaid, the Prepayment Notice shall be binding and irrevocable.

4.03 Compulsory Prepayment

- A. If the total cost of the Project should fall significantly short of the figure stated in the Recitals, THE BANK may in proportion to the shortfall demand prepayment of one or more Tranches of the Loan together with interest accrued thereon by notice to THE BORROWERS and THE BORROWERS shall comply with such notice. For the purposes of this Article 4.03A a fall in the total cost of the Project shall be significant where it results in THE BANK providing 50% or more of the financing for the Project.
- B. If on or before the date (the "Renewal Date") falling two months prior to the expiry date (the "Expiry Date") of the obligations of any Guarantor under any Guarantee, the obligations of each such Guarantor under the Guarantee have not been renewed in full in favour of THE BANK with effect from the Expiry Date by notice to THE BANK given according to the provisions of such Guarantee, and it has not been possible on or before the Renewal Date to replace with effect from the Expiry Date each Guarantor not renewing such guarantee obligations by another Guarantor or Guarantors each of which is a Qualifying Guarantor (as defined in Article 7.01) on terms and conditions satisfactory to THE BANK, THE BORROWERS shall prepay such part of the outstanding amount of each Tranche as would otherwise cease to be guaranteed under the relevant Guarantee as of the Expiry Date, together with accrued interest thereon and an amount calculated in accordance with the terms of Article 4.02B and C on the first Business Day to occur on or after the date falling one month after the Renewal Date.
- C. If either of THE BORROWERS inform THE BANK that an event of the nature described in Article 8.02(b)(iii) has or is likely to occur or if THE BANK has reasonable cause to believe any such event has occurred or is about to occur, THE BANK may demand that THE BORROWERS consult with it.

Such consultation shall take place within thirty days from the date of such demand. If, after the elapse of thirty days from the date of such demand, THE BANK acting reasonably is of the opinion that such event has, or is likely to have, a material adverse effect on the future servicing of the Loan or on the financial stability of either of THE BORROWERS it may request THE BORROWERS to prepay the Loan, together with accrued interest thereon and an amount calculated in accordance with the terms of Article 4.02B and C. THE BORROWERS shall effect payment of the amount demanded on the date specified by THE BANK, such date being a date falling not less than thirty days from the date of the demand.

D. If either of THE BORROWERS voluntarily prepay more than euro 10 million or the equivalent thereof of any other loan (herein a "Term Loan") with an outstanding term of more than five years, THE BANK may demand prepayment of such proportion of the Loan as the amount repaid of the Term Loan bears to the aggregate outstanding amount of all Term Loans.

THE BANK shall address its demand, if any, to THE BORROWERS within four weeks of receipt of notice under Article 8.02(b)(iv). Any sum demanded by THE BANK shall be paid, together with accrued interest, on the date indicated by THE BANK, which date shall not precede the date of prepayment of the Term Loan in question.

Prepayment of a Term Loan by means of a new loan having a term at least equal to the unexpired term of the loan prepaid shall not be considered to be a prepayment.

For the avoidance of this Article 4.03D shall not apply to any early repayment of Flowserve Corporation's Senior Subordinated Notes due 15 August 2010 or its Revolving Credit Agreement which expires in June 2006.

If THE BANK demands prepayment under this Article 4.03, or upon a demand made pursuant to Article 7.02, it shall be made on the date specified by THE BANK. In case of prepayment, the sums to be prepaid shall be paid together with accrued interest and a compensation, if any, calculated on the amount of the Loan to be prepaid in accordance with Article 4.02B and C. If such payment date specified by THE BANK does not coincide with an interest payment date for the Tranche concerned, THE BORROWERS shall pay to THE BANK such amount as THE BANK shall certify in respect of broken funding arrangements and its administrative costs relating to the prepaid Tranches of the Loan together with interest accrued thereon by notice to THE BORROWERS and THE BORROWERS shall comply with such notice.

4.04 General Provisions regarding Prepayment

In case of any partial prepayment of the Loan due under this Article 4 THE BORROWERS may select which Tranche or Tranches are to be prepaid.

In case of partial prepayment of a Tranche, each amount prepaid shall be applied pro rata in reduction of each outstanding instalment.

Sums prepaid under this Article 4 may not be redrawn.

This Article 4 shall not prejudice Article 10.

ARTICLE 5 Payments

5.01 Place of Payment

Sums payable by THE BORROWERS under this Contract shall be paid to the account notified by THE BANK to THE BORROWERS. THE BANK shall give such notice not less than fifteen days before the due date for the first payment by THE BORROWERS and shall notify any change of account not less than fifteen days before the date of the first payment to which the change applies.

This period of notice does not apply in the case of payment under Article 10.

5.02 Calculation of Payments relating to a Fraction of a Year

Any amount due by THE BORROWERS by way of interest, indemnity or otherwise hereunder, and calculated in respect of any fraction of a year, shall be calculated:

- (a) in respect of any Fixed-Rate Tranche or Revisable-Rate Tranche, on the basis of a year of three hundred and sixty days and a month of thirty days;
- (b) in respect of any VSFR Tranche or any FSFR Tranche, on the basis of a year of three hundred and sixty days and the number of days elapsed;

(c) in respect of any FSFR Tranche expressed in GBP, on the basis of a year of three hundred and sixty five days and the number of days elapsed.

5.03 Dates of Payment

- A. For the purposes of this Contract, a "Payment Date" means a date on which interest is payable pursuant to the following provisions:
 - (1) interest payments due in respect of Fixed-Rate Tranches and Revisable-Rate Tranches are payable on the date(s) specified in the relevant Disbursement Notice in each year;
 - (2) interest payments due in respect of VSFR Tranches are payable on Quarter Days;

and

- (3) interest payments due in respect of FSFR Tranches are payable on the dates specified in the relevant Disbursement Notice.
- B. For the purposes of this Contract generally, payments due in any currency on a date which is not a Relevant Business Day (as defined below) are payable as follows:
 - (a) in respect of any Fixed-Rate Tranche, Revisable-Rate Tranche or VSFR Tranche, on the nearest succeeding Relevant Business Day;
 - (b) in respect of any FSFR Tranche, on the nearest succeeding Relevant Business Day, unless such day falls in the following month, in which case it shall be payable on the nearest preceding Relevant Business Day.

For the purposes of this Article 5.03B, "Relevant Business Day" means in respect of the euro a day on which the Trans-European Automated Real-time Gross Settlement Express Transfer payment system operates and in respect of currencies other than the euro, a day on which banks are open for normal business in the principal domestic financial centre of the currency concerned.

C. Sums due hereunder other than of principal or interest are payable within seven days of receipt by THE BORROWER of the demand by THE BANK.

A sum due from THE BORROWERS shall be deemed paid when it is received by THE BANK.

ARTICLE 6 Particular Undertakings

6.01 Use of Loan

THE BORROWERS shall use the proceeds of the Loan and the other funds mentioned in the third Recital for the sole purpose of financing the Project.

6.02 Completion of the Project

THE BORROWERS shall carry out and complete the Project in accordance with, and by the date specified in, the Technical Description contained in Schedule A as may be amended from time to time.

6.03 Increased Cost of the Project

If the total cost of the Project should exceed the figure stated in the Recitals, THE BORROWERS shall obtain the additional finance required without recourse to THE BANK, so as to enable completion of the Project in accordance with the Technical Description. THE BORROWERS shall inform THE BANK without delay of its plans for funding the excess cost.

6.04 Tendering Procedures

THE BORROWERS shall, so far as appropriate and possible and in a manner satisfactory to THE BANK, purchase goods, secure services and order works for the Project by international tender, open at least to nationals of all countries which are signatories of the Agreement on the European Economic Area, or respect, insofar as the Project enters into its sphere of application, any directive of the Council of the European Union.

6.05 Insurance

So long as the Loan is outstanding, THE BORROWERS shall ensure that all works and property forming part of the Project are insured with first-class insurance companies.

6.06 Maintenance

So long as the Loan is outstanding, THE BORROWERS shall maintain, repair, overhaul and renew all property forming part of the Project as required to maintain it in repair and in good working order.

6.07 Operation of the Project

So long as the Loan is outstanding, THE BORROWERS shall, unless THE BANK shall have consented otherwise in writing, retain title to and possession of the assets which comprise the Project and maintain the Project in substantially continuous operation in accordance with its original purpose and in accordance with applicable environmental and other laws.

THE BANK may withhold its consent only where the proposed action would prejudice THE BANK's interests as lender to THE BORROWERS or would render the Project ineligible for financing by THE BANK, applying the criteria of Article 267 of the Treaty establishing the European Community.

6.08 Non-disposal of Assets

THE BORROWERS shall not either in a single transaction or in a series of transactions whether related or not and whether voluntarily or involuntarily, sell, transfer, lease, or otherwise dispose of all or a substantial part of its assets if such disposal of assets would, in the reasonable opinion of THE BANK, have a material adverse effect on the ability of THE BORROWERS to perform its obligations hereunder, save in case such disposals are made with the prior written consent of THE BANK.

6.09 Pari Passu Undertaking

Each of THE BORROWERS undertake that its obligations under this Contract are direct and unconditional obligations of such BORROWER and rank and will rank at least *pari passu* with all other unsecured and unsubordinated obligations (including contingent obligations) of such BORROWER other than those obligations to its unsecured creditors which would, on a winding-up of such BORROWER, be preferred by operation of law.

ARTICLE 7 Security

7.01 Guarantee

The obligations of THE BANK hereunder in respect of each Tranche are conditional upon the prior execution and delivery to THE BANK of a duly executed Guarantee whereby the Guaranter unconditionally guarantees, as primary obligor and not merely as surety, the due performance of the financial obligations of THE BORROWERS under each Tranche disbursed under this Contract.

The Guarantor shall at all times be a Qualifying Guarantor (as defined below).

For the purpose of this Contract, "Qualifying Guarantor" means a bank or other financial institution, which satisfies one of the following conditions:

- (a) at the time of issue of the Guarantee or, as the case may be, at the time it accedes to the Guarantee, each credit rating that it holds, in respect of its most recent unsecured and unsubordinated long-term issue on any capital market, is not lower than: "A" if the rating is assigned by Standard and Poor's; "A2" if the rating is assigned by Moody's; and "A" if the rating is assigned by Fitch Ratings; and that such bank or other financial institution is otherwise acceptable to THE BANK; or
- (b) is accepted by THE BANK in writing, with a copy to THE BORROWERS, subject to the conditions THE BANK may, in its discretion, deem appropriate, and to the acceptance thereof by the relevant Guarantor.

7.02 Cash Collateral and Substitution of Guarantor

- A. If, at any time while any Tranche is outstanding, in respect of any Qualifying Guarantor including the Guarantor:
 - (a) any one rating of its most recent unsecured and unsubordinated long term debt is equal to or lower than:
 - a. A-, assigned by Standard and Poor's;
 - b. A3, assigned by Moody's; or
 - c. A-, assigned by Fitch Ratings; or

- (b) all of Standard and Poor's, Moody's and Fitch Ratings cease to rate its unsecured and unsubordinated long term debt; or
- (c) in the reasonable opinion of THE BANK such Qualifying Guarantor has failed to comply with any condition specified in THE BANK's notice of acceptance delivered under Article 7.01(b),

THE BANK may request THE BORROWERS to consult with it within 15 days from the date of THE BANK's request at the venue specified in the request.

By a notice delivered to THE BORROWERS not earlier than 30 days from the date of such request, THE BANK may demand that THE BORROWERS shall procure that, within a reasonable period specified in such notice, the relevant Guarantor which is the object of such notice (hereinafter the "Affected Guarantor"):

- (i) shall be substituted as Guarantor under the Guarantee by a Qualifying Guarantor; or
- (ii) shall make, with a bank approved by THE BANK, a cash collateral deposit in the currency specified by THE BANK in its notice, charged in favour of THE BANK on such terms as THE BANK may reasonably require, as security for the Affected Guarantor's obligations under the Guarantee; or
- (iii) shall execute other security offering protection in manner, form and substance acceptable to THE BANK.

If none of the foregoing action is taken within the period specified by THE BANK in its relevant notice and to its satisfaction, THE BORROWERS shall, upon demand by THE BANK, immediately prepay to THE BANK an amount equal to that Affected Guarantor's percentage participation of the Tranche as at that date, together with all unpaid interest accrued to the date of prepayment on the amount prepaid and any other sums then payable under this Contract on the amount prepaid.

For the avoidance of doubt, the provisions of this Article 7.02A shall not in any way limit or restrict the right of the THE BANK to demand prepayment of the Loan pursuant to Article 10.

B. If an event of the nature described in Article 10.01A(c), (d) or (e) occurs in relation to any Guarantor, THE BORROWERS shall replace such Guarantor with a Qualifying Guarantor. If THE BORROWERS do not replace the Guarantor within 30 days of the date when the said event occurred, THE BANK may require THE BORROWERS to prepay immediately all or part of the relevant Tranche outstanding, together with all unpaid interest accrued to the date of prepayment on the amount prepaid and all other sums accrued payable under this Contract on the amount prepaid.

7.03 Fees in respect of Replacement of Guarantor

THE BORROWERS shall be permitted to renew each Guarantee (pursuant to Article 3.02 of the relevant Guarantee) or replace the Guarantor thereunder without payment of

any fee or penalty one time during the period that the Loan is outstanding. Thereafter each renewal or replacement shall be subject to a fee of EUR 2 500.

Any replacement of Guarantor under Article 7.02 shall also incur a fee of EUR 2 500.

ARTICLE 8 Information and Visits

8.01 Information concerning the Project

THE BORROWERS shall:

- (a) deliver to THE BANK:
 - (i) on the date falling six months after completion of the Project a report detailing how completion was effected and including a summary of works carried out and their respective costs.
 - (ii) from time to time, any such further documents or information concerning the progress, financing and operation of the Project as THE BANK may reasonably require;
- (b) ensure that its accounting records clearly show its operations relating to the financing and execution of the Project;
- (c) submit for the approval of THE BANK without delay any material change to the plans, timetable or financing requirements for the Project as previously disclosed to THE BANK; and
- (d) generally, inform THE BANK of any fact or event which might substantially prejudice or affect the conditions of execution or operation of the Project.

8.02 Information concerning THE BORROWERS

THE BORROWERS shall each:

- (a) deliver to THE BANK:
 - each year, within thirty days after publication a copy of its annual report including its audited balance sheet and profit and loss
 account, as well as of the statement of auditors relating thereto; and
 - (ii) from time to time, such further information on its financial position in general as THE BANK may reasonably request;
- (b) inform THE BANK immediately of:
 - (i) any alteration to its constitutional documents or modification of a legal requirement directly relating to its business activities and of any proposal or decision known to it which envisages the introduction of such alteration or modification and which is expected to have a material adverse effect on the ability of such BORROWER to comply with its obligations hereunder;

- (ii) any decision made to sell, transfer, lease or otherwise dispose of all or a substantial part of its assets, as provided for in Article 6.08;
- (iii) its belief or, as the case may be, reasonable grounds for belief that an acquisition of control of such BORROWER by a third party will occur or has occurred, such information to be communicated as soon as such BORROWER has reasonable cause to believe that the majority of such BORROWER's outstanding voting shares are or will shortly be held or controlled by a single natural or legal person, or by a group of such persons acting in concert;
- (iv) as soon as reasonably practicable of any decision made by it (in respect of Article 4.03D) or any fact which obliges it or any demand made to it to prepay in advance of maturity more than euro 10 million of any loan with an outstanding term exceeding five years which would have the effect of triggering the provisions of Article 4.03D;
- (v) any circumstance which obliges such BORROWER, and any demand made to it, to make, by reason of default, any prepayment in excess of EUR 10 million in respect of any loan or other financial indebtedness; and
- (c) generally, inform THE BANK forthwith of any fact or event which might materially jeopardise the performance of any material obligation of such BORROWER under this Contract.

8.03 Visits

THE BORROWERS shall permit persons designated by THE BANK to inspect the sites, installations and works comprising the Project and to conduct such checks as they may wish and provide them, or ensure that they are provided, with all necessary assistance for the purpose.

ARTICLE 9 Charges and Expenses

9.01 Taxes, Duties and Fees

THE BORROWERS shall pay all taxes, duties or fees and other impositions of whatsoever nature, including stamp duty and registration fees, arising out of the execution or implementation of this Contract, including the creation of any security for the Loan.

THE BORROWERS shall pay all principal, interest, commission and other amounts due under this Contract gross without deduction of any national or local impositions whatsoever; provided that, if either BORROWER is obliged to make any such deduction, it will gross up the payment to THE BANK so that after deduction, the net amount received by THE BANK is equivalent to the sum due.

9.02 Other Charges

THE BORROWERS shall bear all charges and expenses, including out-of-pocket professional, travel, banking, transfer or exchange charges incurred in connection with:

(i) the preparation, execution, implementation, administration and termination of this Contract or any related document;

- (ii) any amendment, supplement or waiver under this Contract or any related document; and
- (iii) the creation and realisation of any security for the Loan.

THE BANK shall provide documentary support for any such charges or expenses fourteen days before making its demand for payment.

ARTICLE 10 Prepayment upon an Event of Default

10.01 Right to demand repayment

THE BORROWERS shall prepay the Loan or any part thereof together with accrued interest forthwith upon demand being made therefore by THE BANK:

A. immediately:

- (a) if any information or document given to THE BANK by either of THE BORROWERS in connection with the negotiation of this Contract or during its lifetime proves to have been incorrect in any material particular;
- (b) if THE BORROWERS fail on the due date to repay any part of the Loan, to pay interest thereon or to make any other payment to THE BANK as herein provided;
- (c) if, following default, either of THE BORROWERS is required to make any prepayment in excess of EUR 10 million in respect of any loan or other financial indebtedness;
- (d) if either of THE BORROWERS petitions for its bankruptcy, declares to stop paying, seeks a moratorium on payment, is the subject of a petition for bankruptcy by a third party, is declared bankrupt or makes or seeks to make a composition with its creditors without being declared bankrupt;
- (e) if an order is made or an effective resolution is passed for the winding up of either of THE BORROWERS or if either ceases or resolves to cease to carry on the whole or any substantial part of its business or activities, save in the course of a merger or reconstruction previously consented to by THE BANK;
- (f) if an encumbrancer takes possession of, or a receiver, liquidator or administrator is appointed over, a substantial part of the assets of either of THE BORROWERS or if any distress, execution, sequestration or other process is levied or enforced upon or sued out against any of their property and is not discharged or stayed within fourteen days;
- (g) if any event occurs or step is taken which, in the opinion of THE BANK is likely to jeopardise the servicing of the Loan or to adversely affect any security therefore; or
- B. upon expiry of a reasonable period of time specified in a notice served by THE BANK on THE BORROWERS, without the matter being remedied to the satisfaction of THE BANK:

- (a) if either of THE BORROWERS fail to comply with any obligation under this Contract other than one mentioned in Article 10.01 indents (A)(b); or
- (b) if any fact or circumstance stated in the Recitals, alters or ceases to exist and if this either prejudices the interests of THE BANK as lender to THE BORROWERS or adversely affects the operation of the Project.

10.02 Other Rights at Law

Article 10.01 shall not restrict any other right of THE BANK at law to require prepayment of the Loan.

10.03 Indemnities

(1) Fixed-Rate Tranche and Revisable-Rate Tranches

In case of demand under Article 10.01 in respect of any Fixed-Rate Tranche or Revisable-Rate Tranche, THE BORROWERS shall pay to THE BANK a sum calculated in accordance with the procedures laid down in Article 4.02B on the sum which has become due and payable. Such sum shall accrue from the due date for payment specified in THE BANK's notice of demand and be calculated on the basis that prepayment is effected on that date.

(2) VSFR Tranche and FSFR Tranches

In case of a demand under Article 10.01 in respect of a FSFR Tranche or a VSFR Tranche, THE BORROWERS shall pay to the Bank a sum equal to the present value of 0.15 % (15 Basis Points) per annum calculated and accruing on the amount due to be prepaid in the same manner as interest would have been calculated and have accrued, if that amount would have remained outstanding according to the original amortisation schedule of the Tranche.

Such present value shall be determined using a discount rate, applied as of each relevant Payment Date. The discount rate shall be the Bank's standard rate of interest for fixed-rate loans denominated in the currency of, and having the same debt service profile to the final maturity as, the amount in respect of which the demand is made.

Amounts due by THE BORROWERS pursuant to this Article shall be payable on the date of prepayment specified in THE BANK's demand.

10.04 Non-Waiver

No failure or delay by THE BANK in exercising any of its rights under this Article 10 shall be construed as a waiver of such right.

10.05 Application of Sums received

Sums received following a demand under Article 10 shall be applied first in payment of costs, damages, commissions and interest in that order and secondly in reduction of outstanding instalments of principal of such Tranche in inverse order of maturity as THE BANK may select.

ARTICLE 11 Law and Jurisdiction

11.01 Law

The formation and validity of this Contract and the relations between the parties thereto shall be governed by English law.

11.02 Jurisdiction

All disputes concerning this Contract shall be submitted to the jurisdiction of the Courts of England. the Bank appoints The Securities Management Trust Limited whose present address is 19 Old Jewry, London EC2 to be its agent for the purpose of accepting service of legal process.

11.03 Evidence of Sums due

The certificate of THE BANK as to any amount due to THE BANK under this Contract and any record of THE BANK of any such amount shall, absent manifest error, each be prima facie evidence of that amount.

ARTICLE 12 Final Clauses

12.01 Addresses

Notices and other communications given hereunder by one party to this Contract to the other shall be sent to the latter's address set out below or to such other address as the addressee shall have previously notified in writing as its new address for such purpose:

- for THE BANK: 100, boulevard Konrad Adenauer

L-2950 Luxembourg

(attn. Credit Risk Department)

- for Flowserve Corporation: Flowserve World Headquarters

5215 North O'Connor Blvd., Suite 2300

Irving, TX 75039

USA

– for Flowserve B.V.: Parallelweg 6

4878 AH Etten-Leur The Netherlands

12.02 Form of Notice

Notices and other communications, for which fixed periods are laid down in this Contract or which themselves fix periods binding on the addressee, shall be served by hand delivery, registered letter or telegram, with recorded delivery, or by any other means of transmission

which affords evidence of receipt by the addressee, notably telex. The date of registration or, as the case may be, the stated date of receipt of transmission shall be conclusive for the determination of a period.

12.03 Recitals, Schedules and Annexes

The Recitals and the following Schedules form part of this Contract:

- Schedule A Technical Description

- Schedule B Definition of EURIBOR

The following Annexes are attached hereto:

- Annex IA Authority of Signatory of Flowserve Corporation

- Annex IB Authority of Signatory of Flowserve B.V.

- Annex IIA Copy of Resolution of the Board of Directors of

Flowserve Corporation

- Annex IIB Copy of Resolution of the Board of Directors of

Flowserve B.V.

- Annex III Form of Legal Opinion

IN WITNESS WHEREOF the parties hereto have caused this Contract to be executed in three originals. Each page hereof has been initialled on behalf of THE BANK and on behalf of THE BORROWER by the undersigned or their representatives.

Signed for and on behalf of EUROPEAN INVESTMENT BANK

/s/ Thomas C. Barrett Director

April 19, 2004

/s/ Alphonso Querejeta Director

April 19, 2004

Signed for and on behalf of FLOWSERVE CORPORATION

/s/ Renee J. Hornbaker Vice President

April 14, 2004

Signed for and on behalf of FLOWSERVE B.V.

/s/ Jan van Rens Financial Director

April 14, 2004

DEFINITION OF EURIBOR

"EURIBOR" means:

- (i) in respect of any Reference Period of one month or more, the rate of interest for deposits in euro for a term being the number of whole months corresponding to the duration of the period; and
- (ii) in respect of a Reference Period of less than a month, the rate of interest for deposits in euro for a term of one month,

as published at 11.00 a.m. Brussels time or at a later time acceptable to the Bank on the day (the "Reset Date") which falls two TARGET Business Days prior to the first day of the Reference Period (the period for which the rate is taken being hereinafter called the "Representative Period"), on Reuters page EURIBOR 01 or its successor page or, failing which, by any other means of publication chosen for this purpose by the Bank.

If such rate is not so published, the Bank shall request the principal euro-zone office of four major banks in the euro-zone to quote the rate at which euro deposits in the amount of the relevant Tranche are offered by each of them as at approximately 11:00 a.m., Brussels time, on the Reset Date to prime banks in the euro-zone interbank market for a period equal to the relevant Reference Period. If at least two quotations are provided, the rate for that Reset Date will be the arithmetic mean of the quotations.

If fewer than two quotations are provided as requested, the rate for that Reset Date will be the arithmetic mean of the rates quoted by major banks in the euro-zone, selected by the Bank, at approximately 11:00 a.m., Brussels time, on the Reset Date to leading European banks for loans in euros for the relevant Reference Period in the amount of the relevant Tranche.

The Bank may include in this Schedule any additional provisions regarding EURIBOR adopted by the EURIBOR Steering Committee created by and acting under the supervision of EURIBOR FBE and EURIBOR ACI, to be effective on the first day of a Reference Period upon the Bank giving five TARGET Business Days' written notice to the Borrower.

Flowserve Corporation Flowserve World Headquarters 5215 N. O'Connor Blvd. Suite 2300 Irving, TX 75039 USA

TNT

For the attention of Dave Chavenson, Vice President and Treasurer

and

Flowserve B.V. Parallelweg 6 4878 AH Etten-Leur The Netherlands

Attention Company Secretary

Luxembourg, July 2, 2004 JU/SES/sm/2004

Subject: Flowserve Product Development (GB)

Finance Contract between European Investment Bank, Flowserve Corporation and Flowserve B.V. dated Irving 14th April 2004, Etten Leur 14tth April 2004 and Luxembourg 19th April 2004 (the "Finance Contract")

Dear Sirs,

We refer to our discussions concerning the provision of a letter or letters of credit as security under the Finance Contract in the same manner as guarantees. We are agreeable to such security provided that any such letter of credit is in a form acceptable to us and that the following amendments are made to the Finance Contract:

- 1. Article 7.01 shall become Article 7.01A and a new Article 7.01B shall be inserted in the following form:
- "7.01B THE BORROWER may provide a letter of credit ("Letter of Credit") rather than a guarantee provided such Letter of Credit is issued by a Qualifying Guarantor. Where a Letter of Credit is issued rather than a Guarantee;
 - (a) such Letter of Credit may be for the period until the Tranche and all other amounts outstanding in respect of such Tranche under the Finance Contract are repaid in full or for such shorter period or periods as THE BANK may agree provided that the initial shorter period (the "Initial Period") is for a minimum of twenty four months and shall be renewable on the basis set out in Article 7.01B(b) below:
 - (b) if on or before a date falling three months prior to the date of expiry of the Initial Period or, as the case may be, the expiry of any extension period agreed under this Article 7.01B, THE BORROWER shall have by notice in writing (with prior written confirmation acceptable to THE BANK from the Qualifying Issuer) to THE BANK (the "Extension Notice") offered to extend the duration of the obligations of Qualifying Guarantor under the Letter of

Credit for a minimum period of 360 days, THE BANK shall notify in writing its acceptance (the "Extension Acceptance") of such extension to THE BORROWER and Qualifying Issuer on or before a date falling two months prior to the date of expiry of the Initial Period or, as the case may be, the expiry of any extension period agreed under this Article 7.01B. If THE BORROWER fails to provide THE BANK with an Extension Notice within the time limits set out above THE BORROWER shall, upon demand by THE BANK, immediately prepay to THE BANK the relevant Tranche as at that date, together with all unpaid interest accrued in respect of that Tranche to the date of prepayment on the amount prepaid and compensation calculated in accordance with Article 4.02."

2. References in the Finance Contract to the Guarantee shall be deemed to include references to any Letter of Credit and references to the Guarantor shall include an issuer of a Letter of Credit. For the avoidance of doubt Article 7.03 shall apply to each renewal of a Letter of Credit in the same manner as to renewals of any Guarantee.

Would you kindly signify your acceptance of the foregoing by countersigning the enclosed copy of this letter at the point indicated below, and by returning the letter to the Bank by fax to H. Chamberlain/ S. Sturmer (Fax N° +352 4379 3697) and courier by 7th July 2004 at the latest.

Yours faithfully, EUROPEAN INVESTMENT BANK

> /s/ A. Querejeta Director Legal Department—Operations

/s/ P. Jedefors Director—Credit Risk Department

Agreed and Accepted

/s/ David F. Chavenson

for and on behalf of **FLOWSERVE CORPORATION** by: David F. Chavenson, Vice President date:

/s/ John M. Nanos

for and on behalf of **FLOWSERVE B.V.** by: John M. Nanos, Director date:

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

BY AND AMONG

FLOWSERVE B.V.,

AS APPLICANT

FLOWSERVE CORPORATION

AND

THE MATERIAL SUBSIDIARIES LISTED ON SCHEDULE 1 HERETO,

AS GUARANTORS

CALYON NEW YORK BRANCH,

AS ADMINISTRATIVE AGENT AND ISSUING LENDER

AND

THE LENDERS NAMED HEREIN,

LENDERS

* * *

\$89,250,000

* * *

JULY 28, 2004

TABLE OF CONTENTS

		PAGE
SECTION 1	DEFINITIONS.	1
1.1	Defined Terms.	
1.2	Number and Gender of Words; Other References	
1.3	Accounting Terms.	
1.5	ACCOUNTERING TERMS	
SECTION 2	LETTER OF CREDIT FACILITY	18
2.1	Letter of Credit Commitment	18
2.2	Request for Issuance of Letter of Credit	20
2.3	Amendment Procedures.	20
2.4	Extension of Credit Expiration Date	20
2.5	Mandatory Letter of Credit Commitment Reduction	21
SECTION 3	REIMBURSEMENT AND PAYMENT OBLIGATIONS	21
3.1	Reimbursement of Drawings Under a Letter of Credit	
3.2	Fees and Commissions	
3.3	Computation of Rates and Fees	
3.4	Additional Costs.	
3.5	Expenses; Indemnification.	23
3.6	Taxes	24
3.7	Form and Currency of Payments.	26
3.8	Acceleration of Undrawn Amounts.	27
3.9	Maximum Rate	28

4.1	Guaranty of Payment	28
4.2	Obligations Unconditional	2
4.3	Modifications	2
4.4	Waiver of Rights.	30
4.5	Reinstatement	30
4.6		
	es30	_
4.7	Subrogation	3(
SECTION 5	CONDITIONS PRECEDENT TO ISSUANCE OF THE LETTER OF CREDIT	31
5.1		
5.2		
5.2		
5.4		
5.5	·p	
5.6		
5.7		
5.8		
5.9		
5.1	0 EIB Loan	32
5.1	1 Parent Credit Agreement	32
5.1	2 Additional Information	32
anamrov. 6	REPRESENTATIONS AND WARRANTIES.	2
SECTION 6		
6.1	Organization, Powers	32

SECTION 4

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

i

6.2	Authorization	. 3
6.3	Enforceability	. 3
6.4	Governmental Approvals	
6.5	Financial Statements	. 3
6.6	No Material Adverse Change	. 3
6.7	Title to Properties; Possession Under Leases	. 3
6.8	Subsidiaries	. 3
6.9	Litigation; Compliance with Laws	. 3
6.10	Agreements	. 3
6.11	Federal Reserve Regulations	. 3
6.12	Investment Company Act; Public Utility Holding Company Act	. 3
6.13	Tax Returns	. 3
6.14	No Material Misstatements	. 3
6.15	Employee Benefit Plans	. 3
6.16	Environmental Matters	. 3
6.17	Labor Matters.	. 3
6.18	Solvency	. 3
6.19	Projections.	. 3
6.20	Senior Indebtedness	
6.21	Consolidated Coverage Ratio.	. 3
6.22	EIB Loan	
SECTION 7	AFFIRMATIVE COVENANTS OF THE CREDIT PARTIES.	. 3
7.1	Existence; Businesses and Properties.	. 3
7.2	Insurance	
7.3	Obligations and Taxes	
7.4	Financial Statements, Reports, etc.	
7.5	Litigation and Other Notices.	
7.6	Maintaining Records; Access to Properties and Inspections	
7.7	Use of Proceeds.	
7.8	Further Assurances	
7.9	Interest Rate Protection.	
7.10	EIB Loan	
7.10	EID BOWN	2
SECTION 8	NEGATIVE COVENANTS OF THE CREDIT PARTIES.	Δ
8.1	Indebtedness.	
8.2	Liens	
8.3	Sale and Lease-Back Transactions.	
8.4	Investments, Loans and Advances.	
8.5	Mergers, Consolidations, Sales of Assets and Acquisitions	
8.6	Dividends and Distributions, Restrictive Agreements.	
8.7	Transactions with Affiliates	
8.8	Business of the Credit Parties	
8.9	Other Indebtedness and Agreements.	
8.10	Capital Expenditures.	
8.11	Interest Coverage Ratio	
8.12	Fixed Charge Coverage Ratio.	
8.12	Maximum Leverage Ratio.	
8.14	Designated Senior Indebtedness.	
8.14	Fiscal Year	
8.15	fiscal leaf	. 4

SECTION 9	EVENTS OF DEFAULT.	49
9.1	Events of Default	50
9.2	Remedies Upon Event of Default	51
SECTION 10	AGREEMENT AMONG LENDERS.	E 0
10.1	Administrative Agent.	
10.1	Expenses.	
10.2	Proportionate Absorption of Losses	
10.3	Delegation of Duties; Reliance.	
10.4	Limitation of Liability.	
10.5	Default; Collateral	
10.6	Limitation of Liability	
10.7	Relationship of Lenders.	
10.9	Benefits of Agreement	
10.10	Obligation Several	55
SECTION 11	MI SCELLANEOUS.	
SECTION II	Headings.	
11.1	Nonbusiness Days.	
11.2	Nondusiness Days	
11.3	Form and Number of Documents.	
11.4		
	Survival	
11.6	Governing Law	
11.7	Invalid Provisions	
11.8	Entirety	
11.9	Jurisdiction; Venue; Service of Process; Jury Trial	
11.10	Amendments, Consents, Conflicts, and Waivers	
11.11	Multiple Counterparts	
11.12	Successors and Assigns; Assignments and Participations	
11.13	Discharge Only Upon Payment in Full; Reinstatement in Certain Circumstances	
11.14	Judgment Currency	
11.15	Setoff	
11.16	Confidentiality	
11.17	USA PATRIOT Act Notice	62

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

iii

SCHEDULES AND EXHIBITS

Schedule 1	Material Subsidiaries
Schedule 1.1(a)	Fifth Third Letters of Credit
Schedule 2.1	Lenders and Commitments
Schedule 6.8	Subsidiaries
Schedule 6.9	Litigation
Schedule 6.16	Environmental Matters
Schedule 8.1(a)	Existing Indebtedness
Schedule 8.1(d)	Existing Purchase Money Indebtedness
Schedule 8.1(e)	Existing Capital Lease Obligations
Schedule 8.1(f)	Outstanding Industrial Revenue Bonds
Schedule 8.1(g)	Existing Indebtedness of Foreign Subsidiaries
Schedule 8.2	Existing Liens
Schedule 8.4	Existing Intercompany Loans
Exhibit A	Assignment of Cash Collateral Account
Exhibit B	Form of Assignment and Acceptance
Exhibit C	Form of Letter of Credit

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

iv

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

THIS LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT is made and entered into as of this 28th day of July, 2004, by and between FLOWSERVE B.V., a company organized under the laws of the Netherlands (the "COMPANY" or "FBV"), FLOWSERVE

CORPORATION, a New York corporation, and the Material Subsidiaries (hereinafter defined) listed on SCHEDULE 1 hereto (each a "GUARANTOR" and collectively, the "GUARANTORS"), the LENDERS (hereinafter defined), and CALYON NEW YORK BRANCH, a duly licensed branch under the New York Banking Law of a foreign banking corporation organized under the laws of the Republic of France, as Administrative Agent (hereinafter defined) for itself and the other Lenders.

WITNESSETH:

WHEREAS, the Company has requested that (a) the Issuing Lender issue for the Company's account, a letter of credit in the amount of \$89,250,000 in order to support that certain credit arrangement between the Company and the EIB, and (ii) the Lenders purchase participations in such letter of credit from Issuing Lender; and

WHEREAS, the Issuing Lender is willing to issue the requested letter of credit for the account of the Company and the Lenders are willing to purchase participations therein under the terms and conditions hereinafter set forth, including, without limitation, the guarantee of the reimbursement obligations of the Company by the Guarantors;

NOW, THEREFORE, for and in consideration of the terms, agreements and covenants set forth herein, the parties hereto do hereby agree as follows:

SECTION 1 DEFINITIONS

1.1 DEFINED TERMS. For the purposes of this Agreement, unless the context otherwise requires, the following terms shall have the respective meanings assigned to them in this SECTION 1 or in the Section or recital referred to:

"ABN" means ABN Amro Bank N.V. and any successor thereto.

"ABN STANDBY CREDIT" means bank guarantees, surety and performance bonds, letters of credit, and similar financial accommodations issued by ABN or any Affiliate thereof for the account of the Parent or any of its Subsidiaries solely to support contractual obligations of the Parent and its Subsidiaries incurred in the ordinary course of business of the Parent and its Subsidiaries.

"ACQUISITION" means the Parent's acquisition of (i) 100% of the share capital of Invensys plc, and (ii) certain assets of certain subsidiaries of Invensys plc.

"ADMINISTRATIVE AGENT" means Calyon New York Branch, and its permitted successor or successors as administrative agent for the Lenders under this Agreement.

"AFFILIATE" of any Person means any other individual or entity who directly or indirectly controls, or is controlled by, or is under common control with, such Person, and, for purposes of this definition only, "control," "controlled by," and "under common control with" mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of voting securities, by contract, or otherwise).

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

1

"AGREEMENT" means this Letter of Credit and Reimbursement Agreement, of which this SECTION 1 forms a part, together with all amendments and modifications thereof and supplements and attachments thereto.

"APPLICABLE LETTER OF CREDIT FEE" means, for any day, with respect to the Letter of Credit, the applicable fee set forth below based upon the Leverage Ratio as of the relevant date of determination:

1	> or equal to 4.00 to 1.0	2.50%
2	> or equal to 3.50 to 1.0 < 4.00 to 1.0	2.25%
3	> or equal to 3.00 to 1.0 $<$ 3.50 to 1.0	2.00%
4	> or equal to 2.50 to 1.0 < 3.00 to 1.0	1.75%
5	< 2.50 to 1.0	1.50%

Each change in the Applicable Letter of Credit Fee resulting from a change in the Leverage Ratio shall be effective on and after the date of delivery to the Administrative Agent of the financial statements and certificates required by SECTION 7.4(a) (i) or 7.4(a) (ii) and SECTION 7.4(a) (iii), respectively, indicating such change until the date immediately preceding the next date of delivery of such financial statements and certificates indicating another such change. Notwithstanding the foregoing, (a) at any time during which the Parent has failed to deliver the financial statements and certificates required by SECTION 7.4(a) (i) or 7.4(A) (ii) and SECTION 7.4(a) (iii), respectively, or (b) at any time after the occurrence and during the continuance of an Event of Default, the Leverage Ratio shall be deemed to be in Level 1 for purposes of determining the Applicable Letter of Credit Fee.

"ASSET SALE" means the sale, transfer, or other disposition (by way of merger, casualty, condemnation, or otherwise), other than an Asset Swap, by the Parent or any of its Subsidiaries (other than the Company or the Material Subsidiaries) to any Person other than the Parent or any other Credit Party of: (a) any capital stock or other equity interests of any of the Parent's Subsidiaries (other than directors' qualifying shares) or (b) any other assets of the Parent or any of its Subsidiaries (other than (i) inventory, damaged, obsolete or worn out assets, scrap and Permitted Investments, in each case disposed of in the ordinary course of business, (ii) the sale of Program Receivables pursuant to the Receivables Program, (iii) dispositions between or among Subsidiaries of the Parent that are not Credit Parties, (iv) dispositions from Credit Parties to Subsidiaries of the Parent that are not Credit Parties of assets having an aggregate value not in excess of the Dollar Equivalent amount of \$25,000,000 (from and after August 8, 2000) and (v) sales, transfers, or other dispositions (in addition to those described in CLAUSES (i) through (IV) above) in any fiscal year of the Parent of assets having an aggregate value not in excess of \$5,000,000).

"ASSET SWAP" means any transfer of assets of the Parent or any of its Subsidiaries (other than the Company or the Material Subsidiaries) to any Person other than the Parent or any Affiliate of the Parent in exchange for assets of such Person if such exchange would qualify, whether in part or in full, as a like-kind exchange pursuant to Section 1031 of the Code. Nothing in this definition shall require the Parent or any of its Subsidiaries to elect that Section 1031 of the Code be applicable to any Asset Swap.

"ASSIGNMENT AND ACCEPTANCE AGREEMENT" means an Assignment and Acceptance Agreement substantially in the form of EXHIBIT B attached hereto.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

2

"ASSIGNMENT OF CASH COLLATERAL ACCOUNT" means that certain Assignment of Cash Collateral Account substantially in the form of EXHIBIT A attached hereto and incorporated herein by reference, executed by the Company for the benefit of the Administrative Agent (for the benefit of the Lenders) with respect to the Cash Collateral Account.

"ATTRIBUTABLE OBLIGATION" of any Person means, with respect to any Sale and Lease-Back Transaction of such Person as of any particular time, the present value at such time discounted at the rate of interest implicit in the terms of the lease of the obligations of the lessee under such lease for net rental payments during the remaining term of the lease (including any period for which such lease has been extended or may, at the option of such Person, be extended).

"AVAILABLE AMOUNT" means, for each day on which the Letter of Credit shall remain in effect, the aggregate maximum amount which is available to be drawn under the Letter of Credit.

"BENEFICIARY" means the EIB.

"BOARD" shall mean the Board of Governors of the Federal Reserve System

of the United States of America.

"BUSINESS DAY" means any day other than a Saturday, Sunday, or day on which banks in New York City are authorized or required by Law to close; provided, however, that when used in connection with the LIBOR Rate, the term "Business Day" shall also exclude any day on which banks are not open for dealings in deposits in Dollars in the applicable interbank market.

"CAPITAL EXPENDITURES" means, for any period, (a) the additions to property, plant, and equipment and other capital expenditures of the Parent and its consolidated Subsidiaries that are (or should be) set forth in a consolidated statement of cash flows of the Parent for such period prepared in accordance with GAAP, and (b) Capital Lease Obligations incurred by the Parent and its consolidated Subsidiaries during such period, but excluding in each case any such expenditure made to restore, replace, or rebuild property to the condition of such property immediately prior to any damage, loss, destruction, or condemnation of such property, to the extent such expenditure is made with insurance proceeds, condemnation awards, or damage recovery proceeds relating to any such damage, loss, destruction, or condemnation; provided, however, that (i) amounts reinvested as contemplated in the proviso to CLAUSE (a) in the definition of Net Cash Proceeds and (ii) the acquisition of assets pursuant to any Permitted Acquisition shall not be deemed Capital Expenditures.

"CAPITAL LEASE OBLIGATIONS" of any Person shall mean the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP, and the amount of such obligations shall be the capitalized amount thereof determined in accordance with GAAP.

"CAPTIVE INSURANCE COMPANY" shall have the meaning assigned to it in SECTION 8.8 hereof.

"CASH COLLATERAL" shall have the meaning assigned to it in SECTION 3.8 hereof.

"CASH COLLATERAL ACCOUNT" means the segregated cash collateral account of the Company at the Administrative Agent, over which the Administrative Agent shall have sole dominion and control, established pursuant to SECTION 3.8 hereof, styled Flowserve B.V. Cash Collateral Account, which account is pledged to the Administrative Agent (for the benefit of the Lenders) pursuant to the Assignment of Cash Collateral Account.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

3

"CASH COLLATERAL CALL EVENT" means the occurrence and continuance of an Event of Default.

"CASUALTY PROCEEDS" shall have the meaning assigned to such term in the "Mortgages" defined in and delivered pursuant to the Parent Credit Agreement.

"CLOSING DATE" means the later of (a) the date of the Agreement, or (b) the date the Company has satisfied all of the conditions precedent set forth in SECTION 5 hereof.

"CODE" means the Internal Revenue Code of 1986, as amended, or any successor federal tax code, and any reference to any statutory provision shall be deemed to be a reference to any successor provision or provisions.

"COMMITTED SUM" means, as the case may be, the amount stated beside each Lender's name on the most-recently amended SCHEDULE 2.1 to this Agreement (which amount is subject to increase, reduction, or cancellation in accordance with this Agreement).

"COMPANY" shall have the meaning assigned to it in the preamble hereof.

"CONDEMNATION PROCEEDS" shall have the meaning assigned to such term in the "Mortgages" defined in and delivered pursuant to the Parent Credit Agreement.

"CONSOLIDATED EBITDA" means, for any period, Consolidated Net Income for such period, plus (a) without duplication and to the extent deducted in determining such Consolidated Net Income, the sum of (i) Consolidated Interest Expense for such period, (ii) consolidated income tax expense for such period, (iii) all amounts attributable to depreciation and amortization for such period, (iv) any extraordinary losses or extraordinary non-cash charges for such period, (v) the amount of premium payments made by Parent or its Subsidiaries associated with the repurchase or prepayment of the Subordinated Notes from the proceeds of the Fall 2001 Equity Issuance and the amount of such premium payments associated with any further repurchase or repayment of the Subordinated Notes to the extent such repurchase or prepayment is permitted hereunder, (vi) integration and restructuring charges in connection with the IDP Transactions and taken with respect to periods ended on or prior to December 31, 2001, (vii) integration and restructuring charges in connection with the Acquisition and taken with respect to periods ended on or prior to June 30, 2004, in an aggregate amount not to exceed \$40,000,000, and (viii) restructuring and integration charges taken with respect to periods beginning on July 1, 2003 and ended on or prior to December 31, 2004, in an aggregate amount not to exceed \$15,000,000, and minus (b) without duplication and to the extent included in determining such Consolidated Net Income, any extraordinary gains for such period, all determined on a consolidated basis in accordance with GAAP.

"CONSOLIDATED FIXED CHARGES" means, for any period, the sum of (a) Consolidated Interest Expense for such period, (b) the aggregate amount of scheduled principal payments (whether or not made) during such period in respect of long term Indebtedness of the Parent and its Subsidiaries (other than payments made by the Parent or any of its Subsidiaries to the Parent or any other Subsidiary of the Parent), (c) Capital Expenditures for such period, (d) the aggregate amount of Taxes paid in cash by the Parent and its Subsidiaries during such period, and (e) the aggregate amount of Restricted Payments made in cash by the Parent and its Subsidiaries during such period in accordance with SECTION 8.6 (other than SECTION 8.6(a)(II)).

"CONSOLIDATED INTEREST EXPENSE" means, for any period, for the Parent and its Subsidiaries on a consolidated basis, the sum, without duplication, of: (a) all interest, premium payments (other than premium payments associated with the repurchase or prepayment of the Subordinated Notes from

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

4

proceeds of the Fall 2001 Equity Issuance), fees, charges, and related expenses payable by the Parent and its Subsidiaries in connection with borrowed money (including capitalized interest and interest paid on intercompany loans) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP and payable in cash, (b) the portion of rent payable by the Parent and its Subsidiaries with respect to such period under capital leases that is treated as interest in accordance with GAAP and payable in cash, and (c) all fees, discounts, premiums, expenses, or similar amounts incurred by the Parent or any of its Subsidiaries in connection with the Receivables Program for such period, including purchase discounts (net of any loss reserves), purchase premiums, operating expense fees, structuring fees, collection agent fees, unutilized purchase limit fees, and other similar fees and expenses.

"CONSOLIDATED NET INCOME" means, for any period, the net income or loss of the Parent and its Subsidiaries for such period determined on a consolidated basis in accordance with GAAP; provided that there shall be excluded: (a) the income of any Subsidiary of the Parent to the extent that the declaration or payment of dividends or similar distributions by such Subsidiary of that income is not at the time permitted by operation of the terms of its charter or any agreement, instrument, judgment, decree, statute, rule, or regulation applicable to such Subsidiary, and (b) the income or loss of any person accrued prior to the date it becomes a Subsidiary of the Parent or is merged into or consolidated with the Parent or any Subsidiary of the Parent or the date that such Person's assets are acquired by the Parent or any Subsidiary of the Parent.

"CREDIT EXPIRATION DATE" means May 31, 2006, the expiry date of the Letter of Credit, or any subsequent expiry date of the Letter of Credit if extended pursuant to SECTION 2.4.

"CREDIT PARTIES" means the Company, the Parent and each Material

Subsidiary; and "CREDIT PARTY" means any one of them.

"CURRENT ASSETS" means, at any time, the consolidated current assets (other than cash and Permitted Investments) of the Parent and its Subsidiaries.

"CURRENT LIABILITIES" means, at any time, the consolidated current liabilities of the Parent and its Subsidiaries at such time, but excluding, without duplication, (a) the current portion of any long-term Indebtedness and (b) outstanding "Revolving Loans" and "Swingline Loans" under and as defined in the Parent Credit Agreement.

"DEBTOR RELIEF LAWS" means any applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, insolvency, fraudulent conveyance, reorganization or similar laws affecting the rights, remedies or recourse of creditors generally, including without limitation the United States Bankruptcy Code and all amendments thereto, as are in effect from time to time.

"DEFAULT" means any event or condition which upon notice, lapse of time or both would constitute an Event of Default.

"DEFAULT RATE" means a per annum rate of interest equal from day to day to the lesser of (a)(i) the LIBOR Rate, plus (ii) the Applicable Letter of Credit Fee, plus (iii) 2% or (b) the Maximum Rate.

"DEFEASED IRBS" means the industrial revenue bonds due January 15, 2015 and issued by the City of Albuquerque, New Mexico in the aggregate principal amount of \$15,000,000.

"DETERMINING LENDERS" means for all purposes under the Loan Papers, (i) on any date of determination occurring prior to the date upon which the Letter of Credit Commitment has been

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

5

terminated, those Lenders who collectively hold at least 51% of the Letter of Credit Commitment; and (ii) on any date of determination occurring on or after the date upon which the Letter of Credit Commitment has been terminated, those Lenders who collectively hold at least 51% of the LC Exposure.

"DOLLAR EQUIVALENT AMOUNT" means: (a) with respect to any amount of any currency on any day, the equivalent amount in Dollars of such amount of such currency as determined by Administrative Agent using the applicable Exchange Rate on such day, and (b) with respect to any amount of Dollars, such amount.

"DOLLAR SUBORDINATED NOTE INDENTURE" means the indenture dated as of August 8, 2000, between the Parent, the guarantors identified therein, and The Bank of New York, as trustee, as in effect on the date hereof and as hereafter amended from time to time in accordance with the requirements thereof and of this Agreement, pursuant to which the Dollar Subordinated Notes are issued.

"DOLLAR SUBORDINATED NOTES" means the Parent's 12-1/4% Senior Subordinated Notes due 2010, in an initial aggregate principal amount of \$290,000,000 and an aggregate outstanding principal amount as of the date hereof of \$188,500,000, issued pursuant to the Dollar Subordinated Note Indenture and any notes issued by the Parent in exchange for the Dollar Subordinated Notes, as contemplated by the Dollar Subordinated Note Indenture, with substantially identical terms as the Dollar Subordinated Notes.

"DOLLARS" and the sign "\$" mean the lawful currency of the United States of America.

"DOMESTIC SUBSIDIARIES" shall mean all Subsidiaries of the Parent incorporated or organized under the laws of the United States of America, any state thereof or the District of Columbia.

"DRAWING" means a demand for payment under the Letter of Credit in accordance with its terms.

"EIB" means the European Investment Bank, having its seat in Luxembourg.

"EIB LOAN" means the loan from the EIB to the Company and the Parent, as evidenced by that certain Finance Contract dated as of April 14, 2004, between the Company, the Parent, and the EIB, as amended by letter agreement dated June 30, 2004, and as it may otherwise be amended or supplemented from time to time.

"ELIGIBLE ASSIGNEE" means: (a) a Lender; (b) an Affiliate of a Lender with a senior issuer rating from Moody's and S&P equal to or greater than the rating of the assigning Lender; and (c) any other Person approved by the Administrative Agent and Issuing Lender and, unless an Event of Default has occurred and is continuing at the time any assignment is effected in accordance with SECTION 11.12(b), the Company (such approval not to be unreasonably withheld by the Administrative Agent, Issuing Lender and the Company); provided, however, that neither the Company nor any Affiliate of the Company shall qualify as an Eligible Assignee.

"EMU LEGISLATION" shall mean the legislative measures of the European Union for the introduction of, changeover to or operation of the Euro in one or more member states.

"ENVIRONMENTAL LAWS" means all applicable federal, state, local, and foreign laws (including common law), treaties, regulations, rules, ordinances, codes, decrees, judgments and orders (including consent orders), in each case, relating to protection of the environment, natural resources, human health and safety as related to Hazardous Materials or the presence, Release of, or exposure to, Hazardous Materials, or the generation, manufacture, processing, distribution, use, treatment, storage, transport, recycling, or handling of, or the arrangement for such activities with respect to, Hazardous Materials.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

6

"ENVIRONMENTAL LIABILITY" means liabilities, obligations, claims, actions, suits, judgments, or orders under or relating to any Environmental Law for any damages, injunctive relief, losses, fines, penalties, fees, expenses (including fees and expenses of attorneys and consultants) or costs, including those arising from or relating to: (a) any action to address the on- or off-site presence, Release of, or exposure to, Hazardous Materials; (b) permitting and licensing, administrative oversight, insurance premiums, and financial assurance requirements; (c) any personal injury (including death), property damage (real or personal), or natural resource damage; and (d) the compliance or non-compliance with any Environmental Law.

"EQUITY ISSUANCE" means any issuance or sale by the Parent or any of its Subsidiaries of any capital stock or other equity interests of the Parent or any Subsidiary of the Parent, as applicable, or any obligations convertible into or exchangeable for, or giving any Person a right, option, or warrant to acquire such capital stock or equity interests or such convertible or exchangeable obligations, except in each case for (a) any issuance or sale to the Parent or any of its Subsidiaries, (b) any issuance of directors' qualifying shares, and (c) sales or issuances of common stock of the Parent to management or employees of the Parent or any of its Subsidiaries under any employee stock option or stock purchase plan or employee benefit plan in existence from time to time.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations promulgated and rulings issued thereunder from time to time.

"ERISA AFFILIATE" means any trade or business (whether or not incorporated) that, together with any Credit Party, is treated as a single employer under Section 414(b) or (c) of the Code, or solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.

"ERISA EVENT" means: (a) any "reportable event," as defined in Section 4043 of ERISA or the regulations issued thereunder, with respect to a Plan (other than an event for which the 30-day notice period is waived); (b) the existence with respect to any Plan of an "accumulated funding deficiency" (as defined in Section 412 of the Code or Section 302 of ERISA), whether or not waived; (c) the filing pursuant to Section 412(d) of the Code or Section 303(d) of ERISA of an application for a waiver of the minimum funding standard with respect to any Plan; (d) the incurrence by any Credit Party or any of its ERISA

Affiliates of any liability under Title IV of ERISA with respect to the termination of, or withdrawal from, any Plan or the withdrawal or partial withdrawal of any Credit Party or any of its ERISA Affiliates from any Multiemployer Plan; (e) the receipt by any Credit Party or any of its ERISA Affiliates from the PBGC or a plan administrator of any notice relating to the intention to terminate any Plan or Plans or to appoint a trustee to administer any Plan; (f) the adoption of any amendment to a Plan that would require the provision of security pursuant to Section 401(a)(29) of the Code or Section 307 of ERISA; (g) the receipt by any Credit Party or any of its ERISA Affiliates of any notice, or the receipt by any Multiemployer Plan from any Credit Party or any of its ERISA Affiliates of any notice, concerning the imposition of Withdrawal Liability or a determination that a Multiemployer Plan is, or is expected to be, insolvent or in reorganization, within the meaning of Title IV of ERISA; (h) the occurrence of a "prohibited transaction" (within the meaning of Section 4975 of the Code) with respect to which any Credit Party incurs liability; (i) any other event or condition with respect to a Plan or Multiemployer Plan that could reasonably be expected to result in liability of any Credit Party; or (j) any Foreign Benefit Event.

"EURO" and the sign "E" mean the single currency of the European Union as constituted by the Treaty on European Union and as referred to in the ${\tt EMU}$ Legislation.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

7

"EURO SUBORDINATED NOTE INDENTURE" means the indenture dated as of August 8, 2000, between FFBV, the Parent, the other guarantors identified therein, and The Bank of New York, as trustee, as in effect on the date hereof and as hereafter amended from time to time in accordance with the requirements thereof and of this Agreement, pursuant to which the Euro Subordinated Notes are issued.

"EURO SUBORDINATED NOTES" means FFBV's 12-1/4% Senior Subordinated Notes due 2010, in an initial aggregate principal amount of E100,000,000 and an aggregate outstanding principal amount as of the date hereof of E65,000,000, issued pursuant to the Euro Subordinated Note Indenture and any notes issued by FFBV in exchange for the Euro Subordinated Notes, as contemplated by the Euro Subordinated Note Indenture, with substantially identical terms as the Euro Subordinated Notes.

"EVENT OF DEFAULT" shall have the meaning assigned to it in SECTION 9.1 hereof.

"EXCESS CASH FLOW" means, for any fiscal year of the Parent, (a) the sum, without duplication, of (i) Consolidated EBITDA for such fiscal year and (ii) reductions to noncash working capital of the Parent and it Subsidiaries for such fiscal year (i.e., the decrease, if any, in Current Assets minus Current Liabilities from the beginning to the end of such fiscal year), other than any such reduction attributable solely to the establishment during such fiscal year of the Receivables Program, less (b) the sum, without duplication, of (i) the amount of any Taxes payable in cash by the Parent and its Subsidiaries with respect to such fiscal year, (ii) Consolidated Interest Expense for such fiscal year, (iii) Capital Expenditures made in cash in accordance with SECTION 8.10 during such fiscal year, except to the extent financed with the proceeds of Indebtedness, Equity Issuances, Casualty Proceeds, Condemnation Proceeds, or other proceeds that would not be included in Consolidated EBITDA, (iv) permanent repayments of Indebtedness (other than repurchases or prepayments of the Subordinated Notes out of proceeds of the Fall 2001 Equity Issuance) made by the Parent and its Subsidiaries during such fiscal year, but only to the extent that such prepayments by their terms cannot be reborrowed or redrawn and do not occur in connection with a refinancing of all or any portion of such Indebtedness, (v) for the fiscal years ending December 31, 2002, December 31, 2003, and December 31, 2004, cash restructuring expenses and cash integration expenses, to the extent added to Consolidated Net Income in determining Consolidated EBITDA for such year, (vi) for the fiscal year ending December 31, 2002, capitalized cash restructuring expenses and cash integration expenses in an amount not to exceed \$10,000,000, (vii) the amount of extraordinary cash losses for such period to the extent added to Consolidated Net Income in determining Consolidated EBITDA for such year, and (viii) additions to noncash working capital for such fiscal year (i.e., the increase, if any, in Current Assets minus Current Liabilities from the beginning to the end of such fiscal year); provided that to the extent

otherwise included therein, the gains and losses attributable to Asset Sales shall be excluded from the calculation of Excess Cash Flow.

"EXCHANGE RATE" means with respect to any currency on any date of determination, the spot rate at which such currency may be exchanged into Dollars on such date determined by reference to such publicly available service for displaying exchange rates for such date as may be agreed upon from time to time by the Administrative Agent and the Company or, in the absence of such agreement, the "EXCHANGE RATE" shall instead be the Administrative Agent's spot rate of exchange in the interbank market where its foreign currency exchange operations in respect of such currency are then being conducted, at or about 10:00 a.m., local time, on such date for the purchase of Dollars with such currency, for delivery two (2) Business Days later; provided, however, that if, at the time of any such determination, no such spot rate can reasonably be quoted, the Administrative Agent may use any method (including obtaining quotes from two (2) or more market makers for such currency) as it reasonably deems applicable to determine such rate, and such determination shall be conclusive absent manifest error.

"FALL 2001 EQUITY ISSUANCE" means the public offering of the Parent's common stock made by the Parent on November 21, 2001 resulting in gross cash proceeds of \$154,890,850.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

8

"FBV" shall have the meaning assigned to it in the preamble hereof.

"FBV CAPITAL EXPENDITURES" means, for any period, (a) the additions to property, plant, and equipment and other capital expenditures of the Company and its consolidated Subsidiaries that are (or should be) set forth in a consolidated statement of cash flows of the Company for such period prepared in accordance with GAAP, and (b) Capital Lease Obligations incurred by the Company and its consolidated Subsidiaries during such period, but excluding in each case any such expenditure made to restore, replace, or rebuild property to the condition of such property immediately prior to any damage, loss, destruction, or condemnation of such property, to the extent such expenditure is made with insurance proceeds, condemnation awards, or damage recovery proceeds relating to any such damage, loss, destruction, or condemnation.

"FBV EBITDA" means, for any period, FBV Net Income for such period, plus (a) without duplication and to the extent deducted in determining such FBV Net Income, the sum of (i) FBV Interest Expense for such period; (ii) consolidated income tax expense for such period; (iii) all amounts attributable to depreciation and amortization for such period; (iv) any extraordinary losses for such period; (v) any expenses resulting from earn-out payments, royalties, management fees, redistributions paid to any of the Parent's Subsidiaries that are not consolidated with the Company's consolidated financial statements, and minus (b) without duplication and to the extent included in determining such FBV Net Income, any extraordinary gains for such period, all determined on a consolidated basis in accordance with GAAP.

"FBV FIXED CHARGE COVERAGE RATIO" means, for any period, the ratio of (a) FBV EBITDA minus FBV Capital Expenditures for such period to (b) FBV Fixed Charges for such period.

"FBV FIXED CHARGES" means, for any period, the sum of (a) FBV Interest Expense for such period, (b) the aggregate amount of scheduled principal payments (whether or not made) during such period in respect of long term Indebtedness of the Company and its Subsidiaries (other than payments made by the Company or any of its Subsidiaries to the Company or any other Subsidiary of the Company), (c) the aggregate amount of Taxes paid in cash by the Company and its Subsidiaries during such period, and (d) the aggregate amount of distributions paid in cash by the Company and its Subsidiaries for the purposes of dividends, earn-out payments, loans, royalties, management fees, redistributions, principal payments of intercompany loan payables, and other similar items (reasonably determined by the Administrative Agent) paid by Company to any Subsidiary of the Parent that is not consolidated with the Company.

"FBV INTEREST EXPENSE" means for any period, for the Company and its Subsidiaries on a consolidated basis, the sum, without duplication, of: (a) all

interest, premium payments, fees, charges, and related expenses payable by the Company and its Subsidiaries in connection with borrowed money (including capitalized interest and interest paid on intercompany loans) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP and payable in cash, and (b) the portion of rent payable by the Company and its Subsidiaries with respect to such period under capital leases that is treated as interest in accordance with GAAP and payable in cash.

"FBV LEVERAGE RATIO" means, on any date, the ratio of the total consolidated Indebtedness of the Company and its Subsidiaries on such date (excluding Indebtedness under SECTION 8.1(c)) to FBV EBITDA for the period of four consecutive fiscal quarters most recently ended on or prior to such date.

"FBV NET INCOME" means, for any period, the net income or loss of the Company and its Subsidiaries for such period determined on a consolidated basis in accordance with GAAP.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

9

"FFBV" means Flowserve Finance B.V., a Netherlands corporation (besloten vennootschap) and a wholly owned Subsidiary of the Parent.

"FEDERAL FUNDS RATE" means, for any day, the rate per annum (rounded upwards, if necessary, to the nearest 1/100 of 1%) determined (which determination shall be conclusive and binding, absent manifest error) by the Administrative Agent to be equal to the weighted average of the rates on overnight Federal funds transactions with member banks of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate charged to the Administrative Agent (in its individual capacity) on such day on such transactions as determined by the Administrative Agent (which determination shall be conclusive and binding, absent manifest error).

"FIFTH THIRD LETTER OF CREDIT" means each letter of credit previously issued for the account of the Parent or any of its Subsidiaries that: (a) is outstanding on the date hereof, and (b) is listed on SCHEDULE 1.1(a).

"FINANCIAL OFFICER" of any Person means the chief financial officer, principal accounting officer, treasurer, or controller of such Person.

"FINSUB" means a bankruptcy-remote, wholly owned Subsidiary of the Parent, organized solely for the purpose of engaging in the Receivables Program.

"FOREIGN BENEFIT EVENT" means, with respect to any Foreign Pension Plan, (a) the existence of unfunded liabilities in excess of the amount permitted under any applicable Law, or in excess of the amount that would be permitted absent a waiver from a Governmental Authority, (b) the failure to make the required contributions or payments, under any applicable Law, on or before the due date for such contributions or payments, (c) the receipt of a notice by a Governmental Authority relating to the intention to terminate any such Foreign Pension Plan or to appoint a trustee or similar official to administer any such Foreign Pension Plan, or alleging the insolvency of any such Foreign Pension Plan and (d) the incurrence of any liability in excess of the Dollar Equivalent Amount of \$5,000,000 by any Credit Party under applicable Law on account of the complete or partial termination of such Foreign Pension Plan or the complete or partial withdrawal of any participating employer therein, or (e) the occurrence of any transaction that is prohibited under any applicable Law that results in the incurrence of any liability by any Credit Party, or the imposition on any Credit Party of any fine, excise tax, or penalty resulting from any noncompliance with any applicable Law, in each case in excess of the Dollar Equivalent Amount of \$5,000,000.

"FOREIGN PENSION PLAN" shall mean any benefit plan which under applicable Law is required to be funded through a trust or other funding vehicle other than a trust or funding vehicle maintained exclusively by a Governmental

Authority.

"FOREIGN SUBSIDIARY" means any Subsidiary of the Parent that is not a Domestic Subsidiary.

"GAAP" means generally accepted accounting principles of the Accounting Principles Board of the American Institute of Certified Public Accountants and the Financial Accounting Standards Board that are applicable from time to time.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

10

"GOVERNMENTAL AUTHORITY" means any: (a) local, state, municipal, or federal judicial, executive, or legislative instrumentality; (b) private arbitration board or panel; or (c) central bank.

"GOVERNMENTAL REQUIREMENT" means all judgments, orders, writs, injunctions, decrees, awards, laws, ordinances, statutes, regulations, rules, franchises, permits, certificates, licenses, authorization and the like and any other requirements of any government or any commission, board, court, agency, instrumentally or political subdivision thereof.

"GUARANTORS" means the Parent and each Material Subsidiary, and "GUARANTOR" means any one of them.

"GUARANTY" means the Guaranty of each Guarantor made under SECTION 4 hereof.

"HAZARDOUS MATERIALS" means: (a) any petroleum products or byproducts and all other hydrocarbons, coal ash, radon gas, asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, chlorofluorocarbons, and all other ozone-depleting substances, in each case regulated by any Environmental Law, and (b) any chemical, material, substance, or waste that is prohibited, limited, or regulated by or pursuant to any Environmental Law.

"HEDGING AGREEMENT" means any interest rate protection agreement, foreign currency exchange agreement, commodity price protection agreement, or other interest or currency exchange rate or commodity price hedging arrangement.

"IDP" means Ingersoll-Dresser Pump Company, a Delaware general partnership.

"IDP PURCHASE AGREEMENT" means the Purchase Agreement dated as of February 9, 2000, as amended by Amendment No. 1 dated as of July 14, 2000 among the Parent, Flowserve RED Corporation, IDP Acquisition, LLC, and Ingersoll Rand Company.

"IDP TRANSACTIONS" means the acquisition of the partnership interests in IDP pursuant to the IDP Purchase Agreement on or about August 8, 2000.

"INDEBTEDNESS" of any Person means, without duplication: (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, or similar instruments, (c) all obligations of such Person upon which interest charges are customarily paid, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such person, (e) all obligations of such Person issued or assumed as the deferred purchase price of property or services (excluding trade accounts payable and accrued obligations incurred in the ordinary course of business), (f) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (q) all quarantees by such person of Indebtedness of third parties, (h) all Capital Lease Obligations of such Person, (i) all obligations of such person as an account party in respect of letters of credit, and (j) all obligations of such Person in respect of bankers' acceptances. The Indebtedness of any Person shall include the Indebtedness of any partnership in which such Person is a general partner. In addition, for purposes of this Agreement, the Indebtedness of Finsub shall also include all consideration provided to Finsub by the purchaser of Program Receivables less any amounts collected with respect to such Program Receivables. Notwithstanding the foregoing, so long as the Defeased IRBs are owned by the Parent or a wholly owned subsidiary, neither the

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

11

"INVESTMENT GRADE RATINGS" means that the credit rating of the Parent's senior unsecured, non-credit-enhanced long-term debt is (a) BBB- or higher, as determined by S&P, and (b) Baa3 or higher, as determined by Moody's.

"ISSUANCE FEE" shall have the meaning assigned to it in SECTION 3.2(b) hereof.

"ISSUING LENDER" means Calyon New York Branch or such other Lender, if any, that becomes the issuer of the Letter of Credit under this Agreement.

"LAWS" means all applicable statutes, laws, treaties, ordinances, tariff requirements, rules, regulations, orders, writs, injunctions, decrees, judgments, opinions, or interpretations of any Governmental Authority.

"LC EXPOSURE" means, at any time and without duplication, under this Agreement the sum of (a) the aggregate undrawn portion of the Available Amount plus (b) the aggregate unpaid reimbursement obligations of the Company in respect of Drawings.

"LENDERS" means, on any date of determination, the financial institutions named on SCHEDULE 2.1 (as the same may be amended from time to time by the Administrative Agent to reflect the assignments made in accordance with SECTION 11.12(c) of this Agreement), and subject to the terms and conditions of this Agreement, their respective successors and assigns, but not any Participant who is not otherwise a party to this Agreement.

"LETTER OF CREDIT" means the irrevocable standby letter of credit issued by the Issuing Lender (a) in an amount of up to \$89,250,000, (b) supporting the EIB Loan, and (c) in the form of EXHIBIT C attached hereto.

"LETTER OF CREDIT COMMITMENT" shall have the meaning assigned to it in SECTION 2.1 hereof.

"LETTER OF CREDIT FEE" shall have the meaning assigned to it in SECTION 3.2(c) hereof.

"LEVERAGE RATIO" means, on any date for the Parent, the ratio of Total Debt on such date to Consolidated EBITDA for the period of four consecutive fiscal quarters most recently ended on or prior to such date; provided, however, if at any time the Leverage Ratio is being determined, the Parent or any of its Subsidiaries shall have completed a Permitted Acquisition or an Asset Sale since the beginning of the relevant four fiscal quarter period, the Leverage Ratio shall be computed on a pro forma basis (in accordance with the last sentence of SECTION 8.4(g)) as if such Permitted Acquisition or Asset Sale and any related incurrence of Indebtedness, had occurred at the beginning of such period.

"LIBOR RATE" means, for any day, a fluctuating rate of interest equal to (a) the offered quotation which appears on the page of the Telerate Screen which displays an average rate of the British Bankers' Association for the Dollar for a three-month period at or about 10:00 A.M. (London time) on such date, or (b) if such page or such service shall cease to be available, such other page or such other service for the purpose of displaying an average rate of the British Bankers' Association as the Administrative Agent, after consultation with the Company, shall reasonably select; provided that, to the extent that an interest rate is not ascertainable pursuant to the foregoing provisions of this definition, the "LIBOR Rate," for any day, shall be a fluctuating rate of interest determined by the Administrative Agent to be the rate at which deposits in Dollars are offered for a three-month period to major banks in the European interbank market by the Administrative Agent at approximately 11:00 a.m. (London time) on such date.

"LIEN" means a mortgage, pledge, lien, security interest or other charge or encumbrance, or any other type of preferential arrangement to secure or provide for the payment of any obligation of any Person, whether arising by contract, operation of law or otherwise (including, without limitation, the interest of a vendor or lessor under any conditional sale agreement, capital lease or other title retention agreement).

"LOAN PAPERS" means: (a) this Agreement, certificates delivered pursuant to this Agreement, and Exhibits and Schedules hereto; (b) all agreements, documents, or instruments in favor of the Administrative Agent, Issuing Lender, or Lenders (or the Administrative Agent on behalf of Lenders) delivered pursuant to this Agreement or otherwise delivered in connection with all or any part of the Obligation; (c) the Letter of Credit; and (d) all renewals, extensions, or restatements of, or amendments or supplements to, any of the foregoing.

"MARGIN STOCK" shall have the meaning assigned to such term in Regulation U.

"MATERIAL ADVERSE CHANGE" means a material adverse change in, or a material adverse effect upon, the financial condition of the Parent and its Subsidiaries, taken as a whole.

"MATERIAL INDEBTEDNESS" means Indebtedness (other than Indebtedness under this Agreement), or obligations in respect of one or more Hedging Agreements, of any Credit Party in an aggregate principal amount exceeding the Dollar Equivalent Amount of \$10,000,000. For purposes of determining Material Indebtedness, the "principal amount" of the obligations of any Credit Party in respect of any Hedging Agreement at any time shall be the maximum aggregate amount (giving effect to any netting agreements) that such Credit Party would be required to pay if such Hedging Agreement were terminated at such time.

"MATERIAL SUBSIDIARY" means each of the Persons listed on SCHEDULE 1 hereto.

"MAXIMUM AMOUNT and MAXIMUM RATE" respectively mean, for each Lender, the maximum non-usurious amount and the maximum non-usurious rate of interest which, under applicable Law, such Lender is permitted to contract for, charge, take, reserve, or receive on the Obligation.

"MOODY'S" means Moody's Investors Service, Inc., or any successor thereto.

"MULTIEMPLOYER PLAN" shall mean a multiemployer plan as defined in Section 4001(a)(3) of ERISA.

"NET CASH PROCEEDS" means: (a) with respect to any Asset Sale, the cash proceeds (including cash proceeds subsequently received (as and when received) in respect of noncash consideration initially received), net of (i) selling expenses (including broker's fees or commissions, legal fees, transfer and similar taxes, and the Parent's good faith estimate of income taxes paid or payable in connection with such sale), (ii) amounts provided as a reserve, in accordance with GAAP, against any liabilities under any indemnification obligations or purchase price adjustment associated with such Asset Sale (provided that, to the extent and at the time any such amounts are released from such reserve, such amounts shall constitute Net Cash Proceeds), and (iii) the principal amount, premium or penalty, if any, interest and other amounts on any Indebtedness for borrowed money which is secured by the asset sold in such Asset Sale and which is required to be repaid with such proceeds (other than any such Indebtedness assumed by the purchaser of such asset); provided, however, that if (x) the Parent shall deliver a certificate of a Financial Officer to the Administrative Agent at or prior to the time of receipt thereof setting forth the Parent's intent to reinvest such proceeds in productive assets of a kind then used or usable in the business of the Parent and its Subsidiaries within 270 days of receipt of such proceeds and (y) no Default or Event of Default shall have occurred and shall be continuing at the time of such certificate or at the proposed

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

Cash Proceeds except to the extent not so used or contractually committed to be used at the end of such 270-day period, at which time such proceeds shall be deemed to be Net Cash Proceeds; and (b) with respect to any issuance or disposition of Indebtedness or any Equity Issuance or the initial sale of Program Receivables pursuant to the Receivables Program (or any subsequent increase thereto permitted hereunder), the cash proceeds thereof, net of all taxes and fees, commissions, costs and other expenses incurred in connection therewith. Any "boot" or other non-like-kind assets received in connection with an Asset Swap shall, to the extent received in cash or at the time converted into cash, be considered cash proceeds from the sale of an asset.

"NEW SUBORDINATED DEBT" means Indebtedness having the following characteristics: (i) the obligor shall be the Parent, (ii) such Indebtedness shall be unsecured, (iii) such Indebtedness shall be subordinated in right of payment to the Obligations in a manner reasonably acceptable to the Administrative Agent, (iv) such Indebtedness shall not have any scheduled payment of principal, scheduled prepayment, scheduled mandatory redemption, or sinking fund payment prior to December 31, 2009, (v) such Indebtedness shall not contain any provision prohibiting the creation or assumption of any Lien on any of the properties or assets of the Parent or its Subsidiaries, whether then owned or thereafter acquired, to secure payment of the Obligations or any agreement renewing, refinancing, or extending the Obligations or this Agreement, (vi) the Parent shall be in compliance with SECTIONS 8.11, 8.12, and 8.13(a) on a pro forma basis after giving effect to the incurrence of such Indebtedness, and (vii) other terms and conditions shall be no less favorable to the Parent or its Subsidiaries or the Lenders in any material respect than the terms and conditions applicable to the Subordinated Notes.

"OBLIGATION" means all present and future indebtedness, fees, commissions, obligations, and liabilities of any Credit Party, and all renewals and extensions thereof, or any part thereof, now or hereafter owed to the Administrative Agent, Issuing Lender, or any Lender, arising pursuant to this Agreement, the Letter of Credit or the Assignment of Cash Collateral Account, and including, without limitation, any and all interest accruing thereon, and attorneys' fees incurred in the enforcement or collection thereof.

"ORIGINAL CREDIT AGREEMENT" means that certain Credit Agreement dated as of August 8, 2000, among the Parent, Bank of America, N.A., as administrative agent and collateral agent, and certain other agents and lenders party thereto, as amended, restated, modified, or supplemented from time to time.

"ORIGINAL CURRENCY" shall have the meaning assigned to it in SECTION 11.14(a) hereof.

"OTHER CURRENCY" shall have the meaning assigned to it in SECTION 11.14(a) hereof.

"OTHER TAXES" shall have the meaning assigned to it in SECTION 3.6(c) hereof.

"PARENT" means Flowserve Corporation, a New York corporation.

"PARENT CREDIT AGREEMENT" means that certain First Amended and Restated Credit Agreement dated as of May 2, 2002, by and among the Parent, Bank of America, N.A., as administrative agent, and the other lenders and agents party thereto, as amended, restated, modified, or supplemented from time to time.

"PARTICIPANT" shall have the meaning assigned to it in SECTION 11.12(e) hereof.

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA.

"PERMITTED ACQUISITION" shall have the meaning assigned to it in SECTION $8.4\,\mathrm{(g)}$ hereof.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

14

"PERMITTED INVESTMENTs" means:

(a) direct obligations of, or obligations the principal of and interest

on which are unconditionally guaranteed by, the United States of America (or by any agency thereof to the extent such obligations are backed by the full faith and credit of the United States of America), in each case maturing within one year from the date of acquisition thereof;

- (b) investments in commercial paper maturing within 270 days from the date of acquisition thereof and having, at such date of acquisition, one of the three highest credit ratings obtainable from S&P or from Moody's;
- (c) investments in certificates of deposit, banker's acceptances, and time deposits maturing within one year from the date of acquisition thereof issued or guaranteed by or placed with, and money market deposit accounts issued or offered by, the Administrative Agent or any domestic office of any commercial bank organized under the laws of the United States of America or any state thereof that has a combined capital and surplus and undivided profits of not less than \$500,000,000;
- (d) fully collateralized repurchase agreements with a term of not more than 30 days for securities described in CLAUSE (a) above and entered into with a financial institution satisfying the criteria of CLAUSE (c) above;
- (e) investments in "money market funds" within the meaning of Rule 2a-7 of the Investment Company Act of 1940, as amended, substantially all of whose assets are invested in investments of the type described in CLAUSES (a) through (d) above; and
- (f) other short-term investments utilized by Foreign Subsidiaries in accordance with normal investment practices for cash management in investments of a type analogous to the foregoing.

"PERSON" means an individual, sole proprietorship, partnership, joint venture, association, trust, estate, business trust, corporation, not-for-profit corporation, sovereign government or agency, instrumentality, or political subdivision thereof, or any similar entity or organization.

"PLAN" means any employee pension benefit plan (other than a Multiemployer Plan) subject to the provisions of Title IV of ERISA or Section 412 of the Code or Section 307 of ERISA, and in respect of which any Credit Party or any ERISA Affiliate is (or, if such plan were terminated, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"PRESCRIBED FORMS" means such duly executed form(s) or statement(s), and in such number of copies, which may, from time to time, be prescribed by Law and which, pursuant to applicable provisions of: (a) an income tax treaty between the United States and the country of residence of the Administrative Agent or a Lender providing the form(s) or statement(s); (b) the Code; or (c) any applicable rule or regulation under the Code, permit the Company to make payments hereunder for the account of the Administrative Agent or such Lender free of deduction or withholding of income or similar taxes.

"PROGRAM RECEIVABLES" means all trade receivables and related contract rights originated and owned by the Parent or any of its Subsidiaries and sold pursuant to the Receivables Program.

"PROJECTIONS" shall have the meaning assigned thereto in SECTION 5.6(a)(i) hereof.

"PRO RATA or PRO RATA PART" means on any date of determination for any Lender: (a) at any time prior to the termination of the Letter of Credit Commitment, the proportion that such Lender's Committed

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

15

Sum bears to the Letter of Credit Commitment, or (b) at any time on or after the termination of the Letter of Credit Commitment, such Lender's proportionate part (whether held directly or through a participation therein and determined after giving effect to any participations) of the LC Exposure.

"RECEIVABLES PROGRAM" means, collectively, (a) the sale of, or transfer of interests in, Program Receivables to Finsub in a "true sale" transaction, and

(b) the sale of, or transfer of interests in, such Program Receivables by Finsub to persons that are not Affiliates of the Parent; provided that all governing terms and conditions (including any terms or conditions providing for recourse to the Parent or any of its Subsidiaries (other than Finsub)) of the Receivables Program shall be subject to the prior written approval of the Administrative Agent, which approval shall not be unreasonably withheld.

"RECEIVABLES PROGRAM DOCUMENTATION" means all written agreements that may from time to time be entered into by the Parent and/or any of its Subsidiaries in connection with the Receivables Program, as such agreements may be amended, supplemented, or otherwise modified from time to time in accordance with the provisions thereof and hereof.

"REGULATION T" means Regulation T of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.

"REGULATION U" means Regulation U of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.

"REGULATION X" means Regulation X of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.

"RELEASE" means any release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching, or migration into or through the environment or within or upon any building, structure, facility, or fixture.

"REPRESENTATIVES" means representatives, officers, directors, employees, attorneys, and agents.

"RESPONSIBLE OFFICER" means: (a) (i) in the case of a corporation or trust where such Person has appointed officers, its president or any vice president, and, in any case where two Responsible Officers are acting on behalf of such Person, the second such Responsible Officer may be a secretary or assistant secretary, or (ii) in the case of a corporation or trust where such Person has not appointed officers, a director or managing director; (b) in the case of a limited partnership, the Responsible Officer of the general partner, acting on behalf of such general partner in its capacity as general partner; and (c) in the case of a limited liability company, the Responsible Officer of the managing member, acting on behalf of such managing member in its capacity as managing member.

"RESTRICTED PAYMENT" shall have the meaning assigned to such term in Section 8.6(a).

"RIGHTS" means rights, remedies, powers, privileges, and benefits.

"S&P" means Standard & Poor's Ratings Group, a division of McGraw Hill, Inc., a New York corporation, or any successor thereto.

"SALE AND LEASE-BACK TRANSACTION" of any Person means any arrangement entered into by such Person or any Subsidiary of such Person, directly or indirectly, whereby such Person or any Subsidiary of such Person shall sell or transfer any property, whether now owned or hereafter acquired, and whereby such Person or any Subsidiary of such Person shall then or thereafter rent or lease as lessee such property

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

16

or any part thereof or other property which such Person or any Subsidiary of such Person intends to use for substantially the same purpose or purposes as the property sold or transferred.

"SEC" means the Securities and Exchange Commission or any successor thereto.

"SUBSIDIARY" of any Person means any corporation, partnership, joint venture, or other entity of which more than 50% of the outstanding capital stock or other equity interests having ordinary voting power (irrespective of whether or not at the time capital stock or other equity interest of any other class or classes of such corporation, partnership, joint venture or other entity shall or

might have voting power upon the occurrence of any contingency) is at the time directly or indirectly owned by such Person.

"SUBORDINATED NOTES" means the Dollar Subordinated Notes and the Euro Subordinated Notes.

"TAXES" shall have the meaning assigned thereto in SECTION 3.6(a) hereof.

"TOTAL DEBT" means, at any time, the total consolidated Indebtedness of the Parent and its Subsidiaries at such time (excluding (a) Indebtedness under SECTION 8.1(k); (b) Indebtedness under SECTION 8.1(o); and (c) Indebtedness of the type described in CLAUSE (i) of the definition of such term and under SECTION 8.1(l), except in each case to the extent of any unreimbursed drawings or payments thereunder.

"U.C.P." means the Uniform Customs and Practice for Documentary Credits (1993 Revision), International Chamber of Commerce Publication No. 500.

"WITHDRAWAL LIABILITY" means liability to a Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan, as such terms are defined in Part I of Subtitle E of Title IV of ERISA.

- 1.2 NUMBER AND GENDER OF WORDS; OTHER REFERENCES. Unless otherwise specified, in the Loan Papers: (a) where appropriate, the singular includes the plural and vice versa, and words of any gender include each other gender, (b) heading and caption references may not be construed in interpreting provisions, (c) monetary references are to currency of the United States of America, (d) section, paragraph, annex, schedule, exhibit, and similar references are to the particular Loan Paper in which they are used, (e) references to "telecopy," "facsimile," "fax," or similar terms are to facsimile or telecopy transmissions, (f) references to "including" mean including without limiting the generality of any description preceding that word, (g) the rule of construction that references to general items that follow references to specific items are limited to the same type or character of those specific items is not applicable in the Loan Papers, (h) references to any Person include that Person's heirs, personal representatives, successors, trustees, receivers, and permitted assigns, (i) references to any Law include every amendment or supplement to it, rule and regulation adopted under it, and successor or replacement for it, and (j) references to any Loan Paper or other document include every renewal and extension of it, amendment and supplement to it, and replacement or substitution for it.
- 1.3 ACCOUNTING TERMS. All accounting terms not specifically defined herein shall be construed in accordance with GAAP and policies consistent with those applied in the preparation of the consolidated financial statements referred to herein.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

17

SECTION 2 LETTER OF CREDIT FACILITY

2.1 LETTER OF CREDIT COMMITMENT.

- (a) The Issuing Lender irrevocably agrees, subject to and in accordance with the terms and conditions contained in this Agreement, to issue the Letter of Credit in an aggregate amount not to exceed at any time outstanding \$89,250,000 (the "LETTER OF CREDIT COMMITMENT"). The rules of the U.C.P. shall apply to the Letter of Credit.
- (b) Immediately upon the issuance by Issuing Lender of the Letter of Credit, Issuing Lender shall be deemed to have sold and transferred to each other Lender, and each other such Lender shall be deemed irrevocably and unconditionally to have purchased and received from Issuing Lender, without recourse or warranty, an undivided interest and participation, to the extent of such Lender's Pro Rata Part in the Letter of Credit and all Rights of Issuing Lender in respect thereof (other than Rights to receive certain fees provided for in SECTION 3.2(c)). Upon issuance, renewal, amendments or extension of the Letter of Credit, Issuing Lender shall provide copies thereof to

the Administrative Agent and each other Lender.

(c) In order to induce Issuing Lender to issue and maintain the Letter of Credit and Lenders to participate therein, the Company agrees to pay or reimburse the Administrative Agent for the benefit of the Issuing Lender and the other Lenders (i) on the date on which any draft is paid, the amount of any draft paid by Issuing Lender and (ii) no later than three (3) days after demand therefor, the amount of any fees (in addition to the fees described in SECTION 3) Issuing Lender customarily charges to a Person similarly situated in the ordinary course of its business for amending letters of credit, for honoring drafts, and taking similar action in connection with letters of credit. The Company's obligations under this SECTION 2.1(c) shall be absolute and unconditional under any and all circumstances and shall be made in accordance with the terms and conditions of this Agreement, including, without limitation, any of the following circumstances: (A) the existence of any claim, setoff, defense, or other Right which the Company may have at any time against the beneficiary named in the Letter of Credit, any transferee of the Letter of Credit (or any Person for whom any such transferee may be acting), the Administrative Agent, Issuing Lender, any Lender, or any other Person, whether in connection with this Agreement, the Letter of Credit, the transactions contemplated herein, or any unrelated transactions (including any underlying transaction between the Company and the beneficiary named in the Letter of Credit); or (B) any draft, certificate, or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid, or any statement therein being untrue or inaccurate in any respect. To the extent any funding of a draft has been made by Lenders pursuant to SECTION 2.1(e), the Administrative Agent shall promptly distribute any such payments received from the Company with respect to such draft to the Issuing Lender and all other Lenders funding such draft according to their Pro Rata Part. Interest on any amounts remaining unpaid by the Company under this clause at any time from and after the date such amounts become payable until paid in full shall be payable by the Company to the Administrative Agent at the Default Rate. In the event any payment by the Company received by the Administrative Agent with respect to the Letter of Credit and distributed to the Issuing Lender and the other Lenders on account of their participations therein is thereafter set aside, avoided, or recovered from the Administrative Agent in connection with any receivership, liquidation, or bankruptcy proceeding, each Lender which received such distribution shall, upon demand by the Administrative Agent, contribute such Lender's ratable portion of the amount set aside, avoided, or recovered, together with interest at the rate required to be paid by the Administrative Agent upon the amount required to be repaid by it.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

18

(d) If any draft shall be presented for payment under the Letter of Credit, the Issuing Lender shall promptly notify the Administrative Agent thereof and Administrative Agent shall promptly notify the Company of the date and amount of such draft; provided that failure to give any such notice shall not affect the obligations of the Company hereunder. Issuing Lender shall make payment upon presentment of a draft for honor unless it appears that presentment on its face does not comply with the terms of the Letter of Credit, regardless of whether (i) any default or potential default under any other agreement has occurred and (ii) the obligations under any other agreement have been performed by the beneficiary or any other Person (and Issuing Lender shall not be liable for any obligation of any Person thereunder). The Administrative Agent, Issuing Lender and Lenders shall not be responsible for, and the Company's reimbursement obligations for honored drafts shall not be affected by, any matter or event whatsoever (including, without limitation, the validity or genuineness of documents or of any endorsements thereof, even if such documents should in fact prove to be in any respect invalid, fraudulent, or forged), or any dispute among the Company, the beneficiary of the Letter of Credit, or any other Person to whom the Letter of Credit may be transferred, or any claims whatsoever of the Company against any beneficiary of the Letter of Credit or any such transferee; provided that, nothing in this

Agreement shall constitute a waiver of the Company's Rights to assert any claim based upon the gross negligence or willful misconduct of the Administrative Agent, Issuing Lender or any Lender.

- (e) If the Company fails to reimburse the Administrative Agent as provided in SECTION 2.1(c) within 24 hours of the demand therefor by the Administrative Agent, the Administrative Agent shall promptly notify each Lender of such failure, of the date and amount of the draft paid, and of such Lender's Pro Rata Part thereof. Each Lender shall promptly and unconditionally make available to the Administrative Agent in immediately available funds such Lender's Pro Rata Part of such unpaid reimbursement obligation, which funds shall be paid to Administrative Agent on or before the close of business on the Business Day on which such notice was given by the Administrative Agent (if given prior to 1:00 p.m., New York time) or on the next succeeding Business Day (if notice was given after 1:00 p.m., New York time). All such amounts payable by any such Lender shall include interest thereon accruing at the Federal Funds Rate from the day the applicable draft is paid by Issuing Lender to (but not including) the date such amount is paid by such Lender to the Administrative Agent. The obligations of Lenders to make payments to the Administrative Agent with respect to the Letter of Credit shall be irrevocable and not subject to any qualification or exception whatsoever (other than the gross negligence or willful misconduct of the Administrative Agent or Issuing Lender) and shall be made in accordance with the terms and conditions of this Agreement under all circumstances, including, without limitation, any of the following circumstances: (i) any lack of validity or enforceability of this Agreement or any of the Loan Papers; (ii) the existence of any claim, setoff, defense, or other Right which the Company may have at any time against the beneficiary named in the Letter of Credit, any transferee of the Letter of Credit (or any Person for whom any such transferee may be acting), the Administrative Agent, Issuing Lender, any Lender, or any other Person, whether in connection with this Agreement, the Letter of Credit, the transactions contemplated herein, or any unrelated transactions (including any underlying transaction between the Company and the beneficiary named in the Letter of Credit); (iii) any draft, certificate, or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid, or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; and (iv) the occurrence of any Event of Default.
- (f) The Company acknowledges that the Letter of Credit will be deemed issued upon delivery to its beneficiary or the Company. If the Company requests the Letter of Credit be delivered to the Company rather than the beneficiary, and the Company subsequently cancels the

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

19

Letter of Credit, the Company agrees to return it to Issuing Lender together with the Company's written certification that it has never been delivered to such beneficiary. If the Letter of Credit is delivered to its beneficiary pursuant to the Company's instructions, no cancellation thereof by the Company shall be effective without written consent of such beneficiary to the Administrative Agent and Issuing Lender and return of the Letter of Credit to Issuing Lender.

(g) Issuing Lender agrees with each Lender that it will exercise and give the same care and attention to the Letter of Credit as it gives to its other letters of credit, and that it shall have no liability to any other Lender with respect to the Letter of Credit (other than liability arising from the gross negligence or willful misconduct of Issuing Lender). Each Lender, Administrative Agent and the Company agree that, in paying any draw under the Letter of Credit, Issuing Lender shall not have any responsibility to obtain any document (other than any documents required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person delivering any such document. The Administrative Agent, Issuing Lender, Lenders, and their respective Representatives shall not be liable to any other Lender or Company for the use which may be made of the Letter of Credit or for any acts or

omissions of any beneficiary thereof in connection therewith. Any action, inaction, error, delay, or omission taken or suffered by the Administrative Agent, Issuing Lender or any of their respective Representatives under or in connection with the Letter of Credit, the draws, drafts, or documents relating thereto, or the transmission, dispatch, or delivery of any message or advice related thereto, if in good faith and in conformity with applicable Laws and in accordance with the standards of care specified in the U.C.P., shall be binding upon the Company and Lenders and shall not place the Administrative Agent, Issuing Lender or any of their respective Representatives under any resulting liability to the Company or any Lender. Any action taken or omitted to be taken by the Administrative Agent or Issuing Lender under or in connection with the Letter of Credit if taken or omitted in the absence of gross negligence or willful misconduct shall not create for the Administrative Agent or Issuing Lender any resulting liability to any Lender or the Company. Notwithstanding any other limitations of liability provided in this SECTION 2.1, neither the Administrative Agent nor Issuing Lender shall be absolved from liability arising from its gross negligence or willful misconduct in connection with the Letter of Credit.

- (h) Although referenced in the Letter of Credit, terms of any particular agreement or other obligation to the beneficiary are not in any manner incorporated herein. The fees and other amounts payable with respect to the Letter of Credit shall be as provided in this Agreement, and drafts under the Letter of Credit shall be deemed part of the Obligation.
- 2.2 REQUEST FOR ISSUANCE OF LETTER OF CREDIT. The Company hereby applies to the Issuing Lender and requests the Issuing Lender to issue, execute and deliver for its account on the Closing Date, the Letter of Credit in an amount of up to the Letter of Credit Commitment. Subject to the terms and conditions of this Agreement, the Issuing Lender hereby agrees to issue the Letter of Credit.
- 2.3 AMENDMENT PROCEDURES. The Company may, from time to time, on or after the Closing Date and the initial issuance of the Letter of Credit, upon not less than ten (10) Business Days prior notice to the Administrative Agent and Issuing Lender, request that amendments be issued to the Letter of Credit which modify the terms of the Letter of Credit, as may be necessary. Any such amendment shall not be effective until it has been approved by the Administrative Agent, Issuing Lender and Determining Lenders, and acknowledged and accepted by the Beneficiary.
- 2.4 EXTENSION OF CREDIT EXPIRATION DATE. The Company may request that the Credit Expiration Date be extended to a date which is no later than the one year after the then-current Credit

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

20

Expiration Date; provided that, (a) any such extension request shall be made in writing (a "FACILITY EXTENSION REQUEST") by the Company and delivered to the Administrative Agent no more than one-hundred fifty (150) days prior to (but no later than one-hundred twenty (120) days prior to) the then-current Credit Expiration Date; (b) no Default or Event of Default shall exist and be continuing; on the date of such Facility Extension Request and on the date of such extension, if applicable; (c) any such Facility Extension Request shall be accompanied by a detailed consolidated budget for the Company and its Subsidiaries for the next succeeding fiscal year (including a projected consolidated balance sheet and related statements of projected operations and cash flow as of the end of and for each quarter of such fiscal year and as of the end of and for such fiscal year and describing the assumptions used for purposes of preparing such budget); (d) no Facility Extension Request may be made which would have the effect of extending the Credit Expiration Date to a date later than July 28, 2011; and (e) the Administrative Agent, Issuing Lender, and each of the other Lenders must approve and agree, in writing, to any such extension. The Company acknowledges that (i) neither the Administrative Agent, Issuing Lender, nor any other Lender has made any representations to the Company regarding its intent to agree to any extensions set forth in this Section, (ii) the Administrative Agent, Issuing Lender, and the other Lenders collectively or individually, shall not have any obligation to extend the Credit Expiration

Date, and (iii) the Administrative Agent's, Issuing Lender's, or any other Lender's agreement to one or more extensions shall not commit the Administrative Agent, Issuing Lender, or such other Lender to any additional extensions. On or prior to the date that is thirty (30) days after receipt by the Administrative Agent of a Facility Extension Request (the "NOTICE DATE"), each Lender shall notify the Administrative Agent of its decision regarding such requested extension and the Administrative Agent shall promptly notify the Company of such decision by the Lenders. Any Lender not responding within such time period shall be deemed to have declined such Facility Extension Request. Failure of the Administrative Agent to provide the Company with written notice of renewal or non-renewal on or prior to the applicable Notice Date in respect of the original Credit Expiration Date or any extended Credit Expiration Date, as the case may be, shall be deemed a nonrenewal of the Letter of Credit beyond the original or extended Credit Expiration Date, as the case may be. If the Administrative Agent, Issuing Lender, and all of the other Lenders agree to extend the current Credit Expiration Date, the Credit Expiration Date of the Letter of Credit shall, effective from the Notice Date, be such extended date.

2.5 MANDATORY LETTER OF CREDIT COMMITMENT REDUCTION. The Company shall immediately notify the Administrative Agent in the event of any prepayment, reduction, or cancellation of the EIB Loan. Upon the occurrence of such event, the Company shall cause the Beneficiary to immediately notify the Administrative Agent and Issuing Lender that it desires to reduce the Letter of Credit by a pro-rated amount. Upon receipt of such notice from the Beneficiary, the Letter of Credit and the Letter of Credit Commitment shall be permanently reduced by such amount, and the Committed Sum of each Lender reduced by its Pro Rata share of such reduction.

SECTION 3 REIMBURSEMENT AND PAYMENT OBLIGATIONS.

In consideration of the issuance of the Letter of Credit, the Company hereby agrees to pay to the Administrative Agent and the Lenders without condition or deduction for any counterclaim, defense, recoupment or setoff the amounts set forth in this SECTION 3 on the dates and in the manner provided herein:

- 3.1 REIMBURSEMENT OF DRAWINGS UNDER A LETTER OF CREDIT. An amount equal to the face amount of each Drawing made under the Letter of Credit, which amount is due and payable to the Administrative Agent in accordance with SECTION 2.1(c)
- 3.2 FEES AND COMMISSIONS. The following fees and commissions for the issuance, arrangement, maintenance and amendment of the Letter of Credit:

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

21

- (a) LETTER OF CREDIT FEE. A rate per annum on the Available Amount equal to the Applicable Letter of Credit Fee for the Letter of Credit, payable to the Administrative Agent (for the Pro Rata benefit of the Lenders) quarterly in arrears from the date of issuance of the Letter of Credit, commencing on the Closing Date, and continuing on the last Business Day of each June, September, December and March thereafter through and including the Credit Expiration Date; and any accrued but unpaid fee as of the Credit Expiration Date shall be due and payable on the Credit Expiration Date.
- (b) ISSUANCE FEE. 0.125%, per annum, on the Available Amount, payable to the Administrative Agent (for the benefit of the Issuing Lender) quarterly in arrears from the date of issuance of the Letter of Credit, commencing on the Closing Date, and continuing on the last business day of each June, September, December and March thereafter through and including the Credit Expiration Date; and any accrued but unpaid fee as of such Credit Expiration Date shall be due and payable on the Credit Expiration Date (the "ISSUANCE FEE").
- (c) OTHER FEES. Such other fees for the administration and fronting of the Letter of Credit as shall be agreed to between the Company, the Administrative Agent and the Issuing Lender in any letter agreements entered into in connection herewith, such fees to be solely for the account of the Administrative Agent or the Issuing Lender, as

applicable.

- 3.3 COMPUTATION OF RATES AND FEES. All computations of interest and fees payable hereunder shall be calculated on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year).
- 3.4 ADDITIONAL COSTS. Amounts sufficient to compensate the Administrative Agent, Issuing Lender and each Lender for any additional actual costs resulting from the introduction of or change in any Law, guideline or request or in the interpretation thereof by any Governmental Authority, central bank or comparable agency charged with the administration of such Law or guideline which directly or indirectly: (a) imposes or modifies or deems applicable any reserve, special deposit or similar requirement against letters of credit issued by Issuing Lender or the participation obligation of the Lenders; (b) increases the amount of capital required or funded against letters of credit issued by Issuing Lender or the participation obligation of the Lenders (except to the extent such increase arises as a result of the individual creditworthiness of the Issuing Lender or a Lender); or (c) imposes on the Administrative Agent, Issuing Lender or a Lender some other banking or monetary requirements affecting the cost of issuing the Letter of Credit under this Agreement, and the result of any event referred to in CLAUSES (a), (b), or (c) above shall be to increase the cost to Issuing Lender of issuing or maintaining the Letter of Credit, or to increase the cost to any Lender of maintaining its participation arrangement in this transaction, payable on demand. Promptly after the Administrative Agent, Issuing Lender or an affected Lender becomes aware of any such introduction, change or requirement, the Administrative Agent, Issuing Lender or such Lender, as the case may be, shall notify the Company thereof; provided that the delay or failure to promptly provide such notice shall not affect the Administrative Agent's, Issuing Lender's or affected Lender's rights under this SECTION 3.4; provided further that the Company shall not be required to compensate the Administrative Agent, Issuing Lender, or other Lender, as the case may be, pursuant to this SECTION 3.4 for any increased cost incurred more than 180 days prior to the date that Administrative Agent, Issuing Lender or such affected Lender notifies the Company of the event giving rise to such increased cost and of its intention to claim compensation therefor; and provided further that, if the event giving rise to such increased cost is retroactive, then the 180 day period referred to above shall be extended to include the period of retroactive effect thereof. A certificate in reasonable detail as to such increased cost, which states the basis of calculation thereof, submitted by the Administrative Agent, Issuing Lender or an affected Lender to the Company shall be sufficient for the purpose of making any

22

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

such claim. Such certificate shall certify that such costs are generally being charged by the Administrative Agent, Issuing Lender or an affected Lender to other similarly situated companies in connection with the issuance and maintenance of, or agreements to issue, similar letters of credit. Such increase in cost shall, with respect to Issuing Lender, be based upon a reasonable allocation of Issuing Lender's aggregate cost related to the Letter of Credit and, with respect to the affected Lender, be based upon actual cost incurred.

In the event that any Lender shall claim payment of any increased costs pursuant to this SECTION 3.4, the Company shall have the option, to be exercised in writing, to (a) compensate such Lender for the specified additional costs on the basis, if any, negotiated between such Lender and the Company, or (b) replace such Lender with another commercial bank or other financial institution; provided that such replacement commercial bank or other financial institution shall: (i) be an Eligible Assignee; (ii) unconditionally offer in writing (with a copy to the Administrative Agent) to purchase all of such Lender's rights and assume all of such Lender's obligations hereunder and under this Agreement without recourse at the principal amount of such Lender's participation plus interest and fees accrued thereon to the date of such purchase on a date therein specified; (iii) pay in full the obligations of the Company owing to the Lender being replaced (other than the requested increased costs) to such Lender concurrently with such replacement; (iv) execute an Assignment and Acceptance Agreement, pursuant to which it becomes a party to this Agreement with a participation commitment equal to that of the Lender being replaced; and (v) upon such execution of such documents referred to in CLAUSE (IV) above and the

payment by the Company of all of the requested costs accruing to the date of purchase, the replacement commercial bank or other financial institution shall constitute a "LENDER" hereunder with a participation commitment as so specified and the Lender being so replaced shall no longer constitute a Lender hereunder; provided that: (A) if a Lender accepts such an offer and such commercial bank or financial institution fails to purchase such rights, interests, and obligations on such specified date in accordance with the terms of such offer, the Company shall continue to be obligated to pay the increased costs to such Lender pursuant to this SECTION 3.4, and (B) if such Lender fails to accept such purchase offer, the Company shall not be obligated to pay to such Lender such increased costs from and after the date of such purchase offer.

3.5 EXPENSES; INDEMNIFICATION.

(a) The Company shall promptly pay after request therefor: (i) all reasonable out-of-pocket costs and expenses paid (or incurred, to the extent such costs are presented by the Administrative Agent to the Company for direct payment) by the Administrative Agent in connection with the syndication of this Agreement, (ii) all reasonable out-of-pocket costs and expenses paid (or incurred, to the extent such costs are presented by the Administrative Agent to the Company for direct payment) by the Administrative Agent and Issuing Lender in connection with the preparation, negotiation, execution, delivery, and administration of this Agreement and the other Loan Papers and any amendment, waiver, consent or other modification of the provisions hereof and thereof (whether or not the transactions contemplated hereby or thereby are consummated), and the consummation and administration of the transactions contemplated hereby and thereby, including all reasonable attorneys' fees, disbursements, and other charges of external counsel; (iii) all out-of-pocket costs and expenses of the Administrative Agent, Issuing Lender, and each of the Lenders incurred in connection with the enforcement, attempted enforcement, or preservation of any Rights under this Agreement or the other Loan Papers (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all reasonable attorneys' fees, disbursements, and other charges of external counsel; and (iv) all out-of-pocket costs and expenses of the Administrative Agent incurred by the Administrative Agent as a result of and during the continuance of an Event of

23

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

Default. The agreements in this Section shall survive the termination of the Letter of Credit Commitment and repayment of all the other Obligations.

(B) TO THE FULLEST EXTENT PERMITTED BY LAW, THE COMPANY HEREBY AGREES TO INDEMNIFY THE ADMINISTRATIVE AGENT, ISSUING LENDER, AND THE LENDERS, AND EACH OF THEIR RESPECTIVE AFFILIATES AND REPRESENTATIVES (EACH AN "INDEMNIFIED PARTY") AND HOLD EACH OF THEM HARMLESS FROM AND AGAINST ANY AND ALL CLAIMS, DAMAGES, LOSSES, LIABILITIES, COSTS OR EXPENSES WHICH MAY ARISE OR BE ASSERTED AGAINST ANY OF THEM IN CONNECTION WITH ANY OF THE LOAN PAPERS, THE LETTER OF CREDIT COMMITMENT, THE USE OR CONTEMPLATED USE OF THE PROCEEDS OF THE LETTER OF CREDIT, OR ANY OTHER MATTER RELATED TO THE TRANSACTIONS CONTEMPLATED HEREBY; PROVIDED, HOWEVER, THAT SUCH INDEMNITY SHALL NOT APPLY TO ANY SUCH CLAIMS, DAMAGES, LOSSES, LIABILITIES, COSTS OR EXPENSES CAUSED BY, ARISING FROM OR EXISTING SOLELY BY REASON OF: (I) SUCH INDEMNIFIED PARTY'S WILLFUL BREACH OF ANY OF ITS OBLIGATIONS UNDER THIS AGREEMENT OR THE LETTER OF CREDIT, OR (II) SUCH INDEMNIFIED PARTY'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT. IN THE EVENT ISSUING LENDER IS RESTRAINED OR ENJOINED BY ACTION OF THE COMPANY FROM PAYMENT TO THE BENEFICIARY UNDER THE LETTER OF CREDIT PURSUANT TO ANY JUDICIAL, GOVERNMENTAL OR QUASI-GOVERNMENTAL PROCEEDING, THE COMPANY AGREES TO: (A) REIMBURSE EACH INDEMNIFIED PARTY FOR ALL COSTS AND ATTORNEYS' FEES REASONABLY INCURRED BY SUCH INDEMNIFIED PARTY IN CONNECTION WITH (1) SUCH PROCEEDING, AND (2) ANY ACTIONS OR CLAIMS OF THE BENEFICIARY AGAINST SUCH INDEMNIFIED PARTY RESULTING FROM OR NOTWITHSTANDING SUCH RESTRAINT OR INJUNCTION, AND (B) EXTEND THE PERIOD DURING WHICH THE

COMPANY'S OBLIGATIONS SHALL REMAIN IN FULL FORCE AND EFFECT FOR A PERIOD OF TIME THAT IS THIRTY (30) DAYS BEYOND THAT PERIOD OF TIME THAT THE EXPIRATION DATE OF THE LETTER OF CREDIT IS EXTENDED BY VIRTUE OF SUCH RESTRAINT OR INJUNCTION. NO INDEMNIFIED PARTY SHALL HAVE ANY LIABILITY FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, OR PUNITIVE DAMAGES RELATING TO THIS AGREEMENT OR ANY OTHER LOAN PAPER OR ARISING OUT OF ITS ACTIVITIES IN CONNECTION HEREWITH OR THEREWITH. THE AGREEMENTS IN THIS SECTION SHALL SURVIVE TERMINATION OF THE LETTER OF CREDIT COMMITMENT AND REPAYMENT OF ALL THE OTHER OBLIGATIONS.

3.6 TAXES.

(a) Amounts sufficient such that any and all payments by the Company hereunder shall be made, in accordance with SECTION 3.7 hereof, free and clear of and without deduction for any and all taxes, levies, imposts, deductions, charges or withholdings, excluding, in the case of the Administrative Agent, Issuing Lender and each Lender, taxes imposed on its income, and franchise taxes imposed on it, by the jurisdiction under the Laws of which the Administrative Agent, Issuing Lender or such Lender (as the case may be) is organized or any political subdivision thereof and, in the case of the Administrative Agent, Issuing Lender and each Lender, taxes imposed on its income, and franchise taxes imposed on it, by the jurisdiction of the Administrative Agent's, Issuing Lender's or such Lender's applicable lending office or any political subdivision thereof (all such non-excluded taxes, levies, imposts, deductions, charges, withholdings and liabilities being hereinafter referred to as "TAXES"). If the Company shall be required by Law to deduct any Taxes from or in respect of any sum payable hereunder to the Administrative Agent, Issuing Lender or any Lender, (A) the sum payable shall be increased as may be necessary so that after making all required deductions (including deductions applicable to additional sums payable under this SECTION 3.6) the Administrative Agent, Issuing Lender or such Lender (as the case may be) receives an amount equal to the sum it would have received had no

2.4

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

such deductions been made, (B) the Company shall make such deductions, and (C) the Company shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable Γ_{AW}

- (b) Notwithstanding anything to the contrary contained in this Agreement, the Company shall be entitled, to the extent it is required to do so by Law, to deduct or withhold income or other similar taxes imposed by the United States of America from interest, fees or other amounts payable hereunder for the account of the Administrative Agent, Issuing Lender or any Lender other than the Administrative Agent, Issuing Lender or a Lender (i) which is a domestic corporation (as such term is defined in Section 7701 of the Code) for Federal income tax purposes or (ii) which has the Prescribed Forms on file with the Company for the applicable year to the extent the deduction or withholding of such taxes is not required as a result of the filing of such Prescribed Forms; provided that, if the Company shall so deduct or withhold any such taxes, it shall provide a statement to the Administrative Agent, Issuing Lender or such Lender setting forth the amount of such taxes so deducted or withheld, the applicable rate and any other information or documentation which the Administrative Agent, Issuing Lender or such Lender may reasonably request for assisting the Administrative Agent, Issuing Lender or such Lender to obtain any allowable credits or deductions for the taxes so deducted or withheld in the jurisdiction or jurisdictions in which the Administrative Agent, Issuing Lender or such Lender is subject to tax.
- (c) In addition, the Company agrees to pay any stamp or documentary taxes or any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or from the execution, delivery or registration of, or otherwise with respect to, this Agreement (hereinafter referred to as "OTHER TAXES").

- (d) The Company agrees to indemnify the Administrative Agent, Issuing Lender and each Lender for the full amount of Taxes or Other Taxes (including, without limitation, any Taxes or Other Taxes imposed by any jurisdiction on amounts payable under this SECTION 3.6) paid by the Administrative Agent, Issuing Lender or such Lender (as the case may be) and any liability (including penalties, interest and expenses) arising therefrom or with respect thereto except as a result of the gross negligence or willful misconduct of the Administrative Agent, Issuing Lender or such Lender. This indemnification shall be made within thirty (30) days from the date the Administrative Agent, Issuing Lender or such Lender (as the case may be) makes written demand therefor.
- (e) Within thirty (30) days after the date of any payment of Taxes by or at the direction of the Company, the Company will furnish to the Administrative Agent, at its address referred to in SECTION 11.3 hereof, the original or a certified copy of a receipt evidencing payment thereof. Should the Administrative Agent, Issuing Lender or any Lender, within five (5) years of the termination of this Agreement, receive any refund, credit or deduction from any taxing authority to which the Administrative Agent, Issuing Lender or such Lender would not be entitled but for the payment by the Company of Taxes as required by this SECTION 3.6 (it being understood that the Administrative Agent, Issuing Lender or such Lender shall use reasonable efforts, as determined in its sole judgment, to claim and to pursue such refund, credits or deduction), the Administrative Agent, Issuing Lender or such Lender thereupon shall repay to the Company an amount with respect to such refund, credit or deduction equal to any net reduction in taxes actually obtained by the Administrative Agent, Issuing Lender or such Lender (as the case may be) and determined by the Administrative Agent, Issuing Lender or such Lender (as the case may be) to be attributable to such refund credit or deduction, provided that the Company upon the request of the Administrative Agent, Issuing Lender or such Lender, agrees to repay the amount

25

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

paid over to the Company (plus any penalties, interest or other charges imposed by the relevant taxing authority) to the Administrative Agent, the Issuing Lender or such Lender in the event the Administrative Agent, the Issuing Lender or such Lender is required to repay such amount attributable to such refund, credit or deduction to such taxing authority. The Company shall reimburse the Administrative Agent, Issuing Lender and each Lender for any reasonable costs incurred by the Administrative Agent, Issuing Lender or such Lender with the written consent of or at the written direction of the Company in determining the specific amount of any such refund, credit or deduction due to the Company pursuant to this SECTION 3.6(e). In no event shall the Company be obligated pursuant to the preceding sentence to make a payment to the Administrative Agent, Issuing Lender or a Lender for associated costs exceeding the amount payable to the Company by the Administrative Agent, Issuing Lender or such Lender in connection with any refund, credit or deduction, without the prior written consent of the Company. This subsection shall not be construed to require the Administrative Agent, Issuing Lender or any Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Company or any other Person.

(f) The Administrative Agent, Issuing Lender and each Lender shall endeavor in good faith efforts (consistent with its internal policies and legal and regulatory restrictions) to select a jurisdiction for its lending office or change the jurisdiction for its lending office, as the case may be, so as to avoid the imposition of any Taxes or Other Taxes or to eliminate the amount of any such additional amounts which may thereafter accrue; provided that no such selection or change of the jurisdiction for its lending office shall be made if, in the reasonable judgment of the Administrative Agent, Issuing Lender or such Lender (as the case may be), such selection or change would be materially disadvantageous to it. In the event that any

Lender shall claim payment of any additional amounts pursuant to this SECTION 3.6, such Lender shall, if the Company shall find a commercial bank or other financial institution which, if it is not a Lender, shall be reasonably acceptable to the Administrative Agent and the Issuing Lender, unconditionally offers in writing (with a copy to the Administrative Agent) to purchase all of such Lender's rights and assume all of such Lender's obligations hereunder without recourse at the principal amount of such Lender's participation hereunder plus interest, fees and other payments accrued thereon to the date of such purchase on a date therein specified, accept such purchase offer and the Company shall be obligated to pay the additional amounts to such Lender pursuant to this SECTION 3.6 to the date of such purchase; provided that (i) if the Lender accepts such an offer and such bank or financial institution fails to purchase such rights and assume such obligations on such specified date in accordance with the terms of such offer, the Company shall continue to be obligated to pay the additional amount to such Lender pursuant to this SECTION 3.6, or (ii) if such Lender fails to accept such purchase offer, the Company shall not be obligated to pay to such Lender such additional amounts from and after the date of such purchase offer.

(g) Without prejudice to the survival of any other agreement of the Company hereunder, the agreements and obligations of the Company contained in this SECTION 3.6 shall survive the payment in full of the Obligation.

3.7 FORM AND CURRENCY OF PAYMENTS.

(a) All payments (i) under SECTION 3.1 hereof for reimbursement of any Drawings, (ii) under SECTION 3.2(a) of the Applicable Letter of Credit Fee, and (iii) under SECTION 3.2(b) of the Issuance Fee, shall be payable in Dollars and with respect to CLAUSE (i) above, shall be payable for the full amount of such Drawing (plus any accrued interest on any unpaid reimbursement obligation). All payments of the administrative fee to the Administrative Agent shall be payable in Dollars. The payment of any and all other fees, commissions, interest, and expenses (except as

26

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

set forth in this SECTION 3.7(a)) under this Agreement shall be made in the currency advanced or incurred, as the case may be. All payments under this Agreement shall be made in immediately available funds at the Administrative Agent's office at 1301 Avenue of the Americas, New York, New York 10019, or at such other office as the Administrative Agent shall specify in writing.

- (b) All payments made or required to be made by or on behalf of the Company pursuant to this Agreement shall be made in the currency advanced or incurred, as the case may be, without reduction, setoff, or counterclaim whatsoever; provided, however, that in the event that the Company is prevented by a Governmental Authority from making such payment in the currency borrowed, the Company shall forthwith give notice of such circumstance to the Administrative Agent. The Company, the Administrative Agent, Issuing Lender, and the Lenders shall attempt to make an alternative arrangement for such payments; provided that, in the event that the Lenders, Issuing Lender, the Administrative Agent, and the Company cannot agree on an alternative arrangement, all such payments shall be made in the Dollar Equivalent Amount of such obligation.
- 3.8 ACCELERATION OF UNDRAWN AMOUNTS. Should a Cash Collateral Call Event occur with respect to the Letter of Credit, the Administrative Agent, by written notice to the Company, may, and at the direction of the Determining Lenders, shall demand that the Company pay to the Administrative Agent (for the Pro Rata benefit of the Lenders) for deposit in a segregated Cash Collateral Account, an amount equal to the Available Amount under the Letter of Credit (the "CASH COLLATERAL"), as security for the Obligation related to the Letter of Credit under SECTION 3.1 hereof, whereupon such amount shall forthwith be due and payable, without presentment, demand, protest or any other notice of any

kind, all of which are hereby expressly waived by the Company. The Cash Collateral Account shall be assigned to the Administrative Agent (for the Pro Rata benefit of the Lenders) pursuant to the Assignment of Cash Collateral Account which shall be executed and delivered by the Company concurrently with the deposit of the Cash Collateral in the Cash Collateral Account. Furthermore, the Company shall promptly execute and deliver to the Administrative Agent, such other security agreements, pledge agreements, assignments, financing statements and amendments and modifications thereto in form and substance satisfactory to the Administrative Agent, as the Administrative Agent may reasonably require from time to time for the purpose of granting to, or maintaining or perfecting in favor of the Administrative Agent perfected Liens in the Cash Collateral and the Cash Collateral Account, together with other assurances of the enforceability, perfection and priority of the Administrative Agent's Liens as the Administrative Agent may reasonably require to avoid material impairment of the Liens granted or purported to be granted pursuant to this SECTION 3.8.

The Cash Collateral held in the Cash Collateral Account may be invested and reinvested from time to time in Permitted Investments as the Company shall determine, which Permitted Investments shall be held in the name of the Administrative Agent (for the Pro Rata benefit of the Lenders), for the benefit of the Company. The tax identification number used in connection with each such Permitted Investment shall be that of the Company. Cash Collateral paid into or held in the Cash Collateral Account with respect to the Letter of Credit (including any Permitted Investments acquired or invested in with such Cash Collateral or allocable to the Letter of Credit), which are in excess of the sum of (a) the then Available Amount of the Letter of Credit plus (b) the amount of the Obligation then outstanding under SECTION 3.1 with respect to the Letter of Credit, may at the option of the Company either: (i) be released to the Company upon request, or (ii) remain or be deposited in the Cash Collateral Account. Appropriate measures shall be taken in order to provide the Administrative Agent with a perfected first priority security interest with respect to each Permitted Investment purchased with the Cash Collateral. Upon a Drawing under the Letter of Credit, the Administrative Agent shall have the right to liquidate any Permitted Investment acquired or invested in with Cash Collateral paid into the Cash Collateral Account with respect to the Letter of Credit and deposit the proceeds therefrom in the Cash Collateral Account,

27

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

and apply such proceeds and other sums paid into the Cash Collateral Account with respect to the Letter of Credit to the Obligation under SECTION 3.1 related to the Letter of Credit, retain such amounts as security for such Obligation or exercise such of the other rights and remedies as may be provided in this Acreement.

In addition to releases to the Company of Cash Collateral and Permitted Investments pursuant to the preceding paragraph: (a) upon the full and final payment of the Obligation of the Company to the Administrative Agent, Issuing Lender and the Lenders under this Agreement, termination of the Letter of Credit Commitment, and cancellation of the Letter of Credit, the Administrative Agent will release its Liens in the Cash Collateral Account and any Permitted Investments and return any remaining Cash Collateral and Permitted Investments to the Company, and (b) if a Cash Collateral Call Event ceases to be continuing with respect to the Letter of Credit, all Cash Collateral paid into or held in the Cash Collateral Account with respect to the Letter of Credit (and any Permitted Investments invested in or acquired with such Cash Collateral or allocable to the Letter of Credit) shall be released by the Administrative Agent to the Company upon request of the Company.

3.9 MAXIMUM RATE. Regardless of any provision contained in any Loan Paper, no Lender shall ever be entitled to contract for, charge, take, reserve, receive, or apply, as interest on the Obligation, or any part thereof, any amount in excess of the Maximum Rate, and, if a Lender ever does so, then such excess shall be deemed a partial prepayment of principal and treated hereunder as such and any remaining excess shall be refunded to the Company. In determining if the interest paid or payable exceeds the Maximum Rate, the Company and Lenders shall, to the maximum extent permitted under applicable Law: (a) treat all loans or advances as but a single extension of credit (and Lenders and the Company agree that such is the case and that any provision herein for multiple loans or advances is for convenience only), (b) characterize any

nonprincipal payment otherwise payable under the Loan Papers as an expense, fee or premium, rather than as interest, (c) exclude voluntary prepayments and the effects thereof, and (d) amortize, prorate, allocate and spread the total amount of interest throughout the entire contemplated term of the Obligation; provided that, if the Obligation is paid and performed in full prior to the end of the full contemplated term thereof, and if the interest received for the actual period of existence thereof exceeds the Maximum Amount, Lenders shall refund such excess, and, in such event, Lenders shall not, to the extent permitted by Law, be subject to any penalties provided by any Laws for contracting for, charging, taking, reserving or receiving interest in excess of the Maximum Amount.

SECTION 4 GUARANTY

4.1 GUARANTY OF PAYMENT. Subject to the limitation set forth below, each Guarantor hereby irrevocably and unconditionally, jointly and severally quarantees to each Lender, Issuing Lender and the Administrative Agent the prompt payment of the Obligation in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration or otherwise) and the timely performance of all other obligations under this Agreement and the other Loan Papers. This Guaranty is a guaranty of payment and not of collection and is a continuing guaranty and shall apply to the entire Obligation whenever arising. Notwithstanding any provision to the contrary contained herein or in any of the other Loan Papers, to the extent the obligations of any Guarantor shall be adjudicated to be invalid or unenforceable for any reason (including, without limitation, because of any applicable state or federal Law relating to fraudulent conveyances or transfers) then the obligations of such Guarantor hereunder shall be limited to the maximum amount that is permissible under applicable Law (whether federal or state or otherwise and including, without limitation, Debtor Relief Laws). In particular, the aggregate liability of Flowserve S.A. resulting from the Guaranty or other obligations of Flowserve S.A. in connection with this Agreement or any other documents (if any) supplemental to this Agreement shall be limited to the aggregate free capital of Flowserve S.A. at any given point n time, i.e. limited to the aggregate amount of retained earnings and reserves of Flowserve S.A. to the extent they can be paid out as dividends according to

28

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

Article 675 of the Swiss Code of Obligations (see also Articles 660 et seq. and 671 et seq. of the Swiss Code of Obligations).

4.2 OBLIGATIONS UNCONDITIONAL. The obligations of each Guarantor hereunder are absolute and unconditional, irrespective of the value, genuineness, validity, regularity or enforceability of any of the Loan Papers or any other agreement or instrument referred to therein, to the fullest extent permitted by applicable Law, irrespective of any other circumstance whatsoever which might otherwise constitute a legal or equitable discharge or defense of a surety or quarantor. Each Guarantor agrees that this Guaranty may be enforced by the Administrative Agent, Issuing Lender or the Lenders without the necessity at any time of resorting to or exhausting any other security or collateral and without the necessity at any time of having recourse to the Loan Papers or any collateral, if any, hereafter securing the Obligation or otherwise and each Guarantor hereby waives the right to require the Administrative Agent, Issuing Lender or the Lenders to make demand on or proceed against the Company or any other Person (including any other Guarantor) or to require the Administrative Agent, Issuing Lender or Lenders to pursue any other remedy or enforce any other right. Each Guarantor further agrees that it shall have no right of subrogation, indemnity, reimbursement or contribution against the Company or any other Guarantor of the Obligation for amounts paid under this Guaranty until such time as the Obligation has been indefeasibly paid in full in cash, the Letter of Credit Commitment under this Agreement has been terminated and no Person or Governmental Authority shall have any right to request any return or reimbursement of funds from the Administrative Agent, Issuing Lender or the Lenders in connection with monies received under this Agreement or the other Loan Papers. Each Guarantor further agrees that nothing contained herein shall prevent the Administrative Agent, Issuing Lender or the Lenders from suing on the Loan Papers or foreclosing its or their, as applicable, security interest in or Lien on any collateral, if any, securing the Obligation or from exercising any other rights available to it or them, as applicable, under this Agreement or

the other Loan Papers, or any other instrument of security, if any, and the exercise of any of the aforesaid rights and the completion of any foreclosure proceedings shall not constitute a discharge of any Guarantor's obligations hereunder; it being the purpose and intent of each Guarantor that, subject to such Guarantor's rights to raise defenses to payment that would be available to it if such Guarantor were named as the "Company" hereunder rather than as a Guarantor, its obligations hereunder shall be absolute, independent and unconditional under any and all circumstances. Neither any Guarantor's obligations under this Guaranty nor any remedy for the enforcement thereof shall be impaired, modified, changed or released in any manner whatsoever by an impairment, modification, change, release, increase or limitation of the liability of the Company or any Guarantor or by reason of the bankruptcy or insolvency of the Company or any Guarantor. Each Guarantor waives any and all notice of the creation, renewal, extension or accrual of any of the Obligation and notice of or proof of reliance by the Administrative Agent or any Lender on this Guaranty or acceptance of this Guaranty. The Obligation, and any part of it, shall conclusively be deemed to have been created, contracted or incurred, or renewed, extended, amended or waived, in reliance upon this Guaranty. All dealings between the Company and the Guarantors, on the one hand, and the Administrative Agent, Issuing Lender and the Lenders, on the other hand, likewise shall be conclusively presumed to have been had or consummated in reliance upon this Guaranty. Each Guarantor hereby subordinates to the Obligation all debts, liabilities and other obligations, now existing or hereafter created, whether direct, indirect, primary, secondary, several, joint and several or otherwise, and irrespective of whether such debts, liabilities and obligations be evidenced by note, contract, open account, book entry or otherwise, owing by the Company to such Guarantor.

4.3 MODIFICATIONS. Each Guarantor agrees that: (a) all or any part of the collateral now or hereafter held for the Obligation, if any, may be exchanged, compromised or surrendered from time to time; (b) none of the Lenders, Issuing Lender, and the Administrative Agent shall have any obligation to protect, perfect, secure or insure any such security interests, liens or encumbrances now or hereafter held, if any, for the Obligation; (c) the time or place of payment of the Obligation may be changed or extended, in whole or in part, to a time certain or otherwise, and may be renewed or accelerated, in whole or in part;

29

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

- (d) the Company and any other party liable for payment under the Loan Papers may be granted indulgences generally; (e) any of the provisions of the Loan Papers may be modified, amended or waived; (f) any party (including any other Guarantor) liable for the payment thereof may be granted indulgences or be released; and (g) any deposit balance for the credit of the Company or any other party liable for the payment of the Obligation or liable upon any security therefor may be released, in whole or in part, at, before or after the stated, extended or accelerated maturity of the Obligation, all without notice to or further assent by any Guarantor, which shall remain bound thereon, notwithstanding any such exchange, compromise, surrender, extension, renewal, acceleration, modification, indulgence or release.
- 4.4 Waiver OF RIGHTS. Each Guarantor expressly waives to the fullest extent permitted by applicable Law: (a) notice of acceptance of this Guaranty by Issuing Lender and the Lenders and of all extensions of credit to the Company by Issuing Lender and the Lenders; (b) presentment and demand for payment or performance of any of the Obligation; (c) protest and notice of dishonor or of default (except as specifically required in this Agreement) with respect to the Obligation or with respect to any security therefor; (d) notice of the Lenders obtaining, amending, substituting for, releasing, waiving or modifying any security interest, Lien or encumbrance, if any, hereafter securing the Obligation, or the Lenders subordinating, compromising, discharging or releasing such security interests, Liens or encumbrances, if any; and (e) all other notices to which a Guarantor might otherwise be entitled.
- 4.5 REINSTATEMENT. Notwithstanding anything contained in this Agreement or the other Loan Papers, the obligations of each Guarantor under this SECTION 4 shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of any Person in respect of the Obligation is rescinded or must be otherwise restored by any holder of any of the Obligation, whether as

a result of any proceedings in bankruptcy or reorganization or otherwise, and each Guarantor agrees that it will indemnify the Administrative Agent, Issuing Lender and each Lender on demand for all reasonable costs and out of pocket expenses incurred by such Person in connection with such rescission or restoration, including any such costs and expenses incurred in defending against any claim alleging that such payment constituted a preference, fraudulent transfer or similar payment under any bankruptcy, insolvency or similar Law.

- 4.6 REMEDIES. Each Guarantor agrees that, as between the Guarantors, on the one hand, and the Administrative Agent, Issuing Lender and the Lenders, on the other hand, the Obligation may be declared to be forthwith due and payable as provided in SECTION 9.2 (and shall be deemed to have become automatically due and payable in the circumstances provided in SECTION 9.2) notwithstanding any stay, injunction or other prohibition preventing such declaration (or preventing such Obligation from becoming automatically due and payable) as against any other Person and that, in the event of such declaration (or such Obligation being deemed to have become automatically due and payable), such Obligation (whether or not due and payable by any other Person) shall forthwith become due and payable by the Guarantors.
- 4.7 SUBROGATION. Each Guarantor agrees that, until the indefeasible payment of the Obligation in full in cash and the termination of the Letter of Credit Commitment, it will not exercise, and hereby waives, any right of reimbursement, subrogation, contribution, offset or other claims against the Company arising by contract or operation of law in connection with any payment made or required to be made by the Guarantors under this Agreement or the other Loan Papers. After the indefeasible payment in full in cash of the Obligation (other than any part of the Obligation that represents contingent contractual indemnities) and the termination of the Letter of Credit Commitment, each Guarantor shall be entitled to exercise against the Company all such rights of reimbursement, subrogation, contribution, and offset, and all such other claims, to the fullest extent permitted by Law.

SECTION 5 CONDITIONS PRECEDENT TO ISSUANCE OF THE LETTER OF CREDIT

30

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

The effectiveness of this Agreement and the obligation of the Issuing Lender to issue the Letter of Credit and the Lenders to participate therein shall be subject to the conditions precedent that the Administrative Agent shall have received on the Closing Date, unless otherwise expressly stated, all of the following (in such quantities as may be reasonably requested by the Administrative Agent), each dated or effective as of the date hereof (unless otherwise stated below) in form and substance satisfactory to the Administrative Agent:

- 5.1 AGREEMENT. This Agreement duly-executed by the Company and each of the Guarantors;
- 5.2 OFFICER'S CERTIFICATE. A certificate executed by a Responsible Officer of the Company stating that no event has occurred and is continuing which constitutes a Default or Event of Default;
- 5.3 EVIDENCE OF AUTHORITY. Such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of the Company and each of the Guarantors as the Administrative Agent may require to establish the identities of and verify the authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Papers to which the Company or any Guarantor is a party;
- 5.4 ORGANIZATIONAL DOCUMENTS. Such evidence as the Administrative Agent may reasonably require to verify that the Company and each of the Guarantors are duly organized or formed, validly existing, and in good standing;
- 5.5 OPINION OF COUNSEL. A favorable opinion of John M. Nanos, Senior Associate General Counsel of the Parent, covering such matters relating to the transactions contemplated hereby as reasonably requested by the Administrative Agent, and substantially in a form acceptable to the Administrative Agent;

5.6 FINANCIAL INFORMATION.

- (a) For the Parent, (i) a business plan for the Parent and its consolidated Subsidiaries for the calendar years ending December 31, 2004, December 31, 2005, and December 31, 2006 (the "PROJECTIONS"), in each case in form satisfactory to the Administrative Agent; and (ii) consolidated audited financial statements for the Parent and its Subsidiaries for the fiscal years ended December 31, 2001, December 31, 2002, and December 31, 2003;
- (b) For the Company, (i) statutory audited consolidated financial statements for the Company and its Subsidiaries for the fiscal years ended December 31, 2001 and December 31, 2002; (ii) unaudited consolidated financial statements for the Company and its Subsidiaries for the fiscal years ended December 31, 2001, December 31, 2002, and December 31, 2003, prepared in accordance with GAAP, and certified by a Financial Officer of the Company; and (iii) a business plan for the Company and its consolidated Subsidiaries for the calendar years ending December 31, 2004, December 31, 2005, and December 31, 2006, in each case in form satisfactory to the Administrative Agent;
- 5.7 LITIGATION. A certificate from each of the Company and the Parent, executed by a Responsible Officer of the Company and the Parent, as applicable, stating that except as otherwise disclosed in writing to the Administrative Agent, there is no pending or, to the knowledge of the Company or the Parent, as applicable, threatened action or proceeding against the Company and the Parent or any of its Subsidiaries (including, without limitation, concerning the ownership of the Company and the Parent or any of its Subsidiaries, or any of their respective properties or assets) before any court,

31

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

governmental agency, or arbitrator, which could, individually or in the aggregate, reasonably be expected to result in a Material Adverse Change;

- 5.8 COMPLIANCE WITH FINANCIAL OBLIGATIONS. A certificate from each of the Credit Parties executed by a Responsible Officer of such Credit Party certifying that such Credit Party is in compliance with all of its existing financial obligations;
- 5.9 INTERCOMPANY INDEBTEDNESS. All intercompany Indebtedness of the Company shall have been subordinated to its Obligations under this Agreement on terms acceptable to the Administrative Agent;
 - 5.10 EIB LOAN. A fully-executed copy of the EIB Loan.
- 5.11 PARENT CREDIT AGREEMENT. A fully-executed amendment under the Parent Credit Agreement which permits the incurrence of the Indebtedness evidenced by this Agreement.
- 5.12 ADDITIONAL INFORMATION. Such other information and documents as may reasonably be required by the Administrative Agent and its counsel.

SECTION 6 REPRESENTATIONS AND WARRANTIES

To induce the Lenders to enter into this Agreement and the Issuing Lender to issue the Letter of Credit, each Credit Party represents and warrants to the Administrative Agent, Issuing Lender, and each Lender, whereby all such representations and warranties are limited to the applicable Credit Party or the business of the applicable Credit Party rendering the representation or warranty, that:

6.1 ORGANIZATION, POWERS. Each of the Credit Parties (a)(i) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (ii) has all requisite power and authority to own its property and assets and to carry on its business as now conducted and as proposed to be conducted, and (iii) is qualified to do business in, and is in good standing in, every jurisdiction where such qualification is required, except where any such failure, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change, and (b) has the power and authority to execute, deliver and perform its obligations under each of the Loan Papers and each other agreement or instrument contemplated hereby to

which it is or will be a party.

- 6.2 AUTHORIZATION. This Agreement and the transactions contemplated hereby: (a) have been duly authorized by all requisite corporate and, if required, stockholder action; and (b) will not (i) violate (A) any provision of applicable Law, or of the certificate or articles of incorporation or other constitutive documents or by-laws of each of the Credit Parties, (B) any order of any Governmental Authority, or (C) any provision of any indenture, agreement, or other instrument to which any Credit Party is a party or by which any of them or any of their property is or may be bound, (ii) be in conflict with, result in a breach of or constitute (alone or with notice or lapse of time or both) a default under, or give rise to any right to accelerate or to require the prepayment, repurchase, or redemption of any obligation under any such indenture, agreement, or other instrument, or (iii) result in the creation or imposition of any Lien prohibited by this Agreement.
- 6.3 ENFORCEABILITY. This Agreement has been duly executed and delivered by each Credit Party and constitutes a legal, valid, and binding obligation of such Credit Party enforceable against such Credit Party in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium, or similar laws or equitable principles relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.

32

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

6.4 GOVERNMENTAL APPROVALS. No action, consent or approval of, registration or filing with or any other action by any Governmental Authority is or will be required in connection with this Agreement and the transactions contemplated hereby, except for such as have been made or obtained and are in full force and effect.

6.5 FINANCIAL STATEMENTS.

- (a) The Parent has heretofore furnished to the Administrative Agent its consolidated balance sheets and statements of income, stockholders' equity, and cash flows as of and for the fiscal year ended December 31, 2003, audited by and accompanied by the opinion of PricewaterhouseCoopers, independent public accountants. Such financial statements present fairly the financial condition and results of operations and cash flows of the Parent and its consolidated Subsidiaries as of such dates and for such periods. Such balance sheets and the notes thereto disclose all material liabilities, direct or contingent, of the Parent and its consolidated Subsidiaries as of the dates thereof. Such financial statements were prepared in accordance with GAAP.
- (b) The Company has heretofore furnished to the Administrative Agent its consolidated balance sheets and statements of income, stockholders' equity, and cash flows as of and for the fiscal years ended December 31, 2001, December 31, 2002, and December 31, 2003. Such financial statements present fairly the financial condition and results of operations and cash flows of the Company and its consolidated Subsidiaries as of such dates and for such periods. Such financial statements were prepared in accordance with GAAP.
- 6.6 NO MATERIAL ADVERSE CHANGE. Since December 31, 2003, no event, change or condition has occurred that has had, or could reasonably be expected to result in a Material Adverse Change.

6.7 TITLE TO PROPERTIES; POSSESSION UNDER LEASES.

(a) Each of the Credit Parties has valid title to, or valid leasehold interests in, all its material properties and assets, except for defects in title that do not interfere with its ability to conduct its business as currently conducted or to utilize such properties and assets for their intended purposes. All such material properties and assets are free and clear of Liens, other than Liens expressly permitted by SECTION 8.2.

- (b) Each Credit Party has complied with all material obligations under all material leases to which it is a party and, to such Credit Party's knowledge, all such leases are in full force and effect
- 6.8 SUBSIDIARIES. SCHEDULE 6.8 sets forth as of the Closing Date a list of all of the Subsidiaries of the Parent and the percentage ownership interest of the Parent therein. The shares of capital stock or other ownership interests so indicated on SCHEDULE 6.8 are fully paid and non-assessable and as of date hereof are owned by the Parent, directly or indirectly, free and clear of all Liens (other than Liens permitted by this Agreement).

6.9 LITIGATION; COMPLIANCE WITH LAWS.

(a) Except as set forth on SCHEDULE 6.9, there are not any actions, suits or proceedings at law or in equity or by or before any Governmental Authority now pending or, to the knowledge of the applicable Credit Party, threatened against such Credit Party or any

33

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

business, property or rights of any such Person (i) that involve any Loan Paper or the transactions contemplated hereby, or (ii) which could reasonably be expected, individually or in the aggregate, to result in a Material Adverse Change.

(b) None of the Credit Parties or any of their respective material properties or assets is in violation of any Law (including United Nations and U.S. sanction laws, and any zoning, building, Environmental Law, ordinance, code or approval, or any building permits), or is in default with respect to any judgment, writ, injunction, decree, or order of any Governmental Authority, where such violation or default could reasonably be expected to result in a Material Adverse Change.

6.10 AGREEMENTS.

- (a) No Credit Party is a party to any agreement or instrument or subject to any corporate restriction that has resulted or could reasonably be expected to result in a Material Adverse Change.
- (b) No Credit Party is in default in any manner under any provision of any indenture or other agreement or instrument evidencing Indebtedness, or any other material agreement or instrument to which it is a party or by which it or any of its properties or assets are or may be bound, where such default could reasonably be expected to result in a Material Adverse Change.

6.11 FEDERAL RESERVE REGULATIONS

- (a) No Credit Party is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of buying or carrying Margin Stock.
- (b) No part of the proceeds of the Letter of Credit will be used, whether directly or indirectly, and whether immediately, incidentally or ultimately, for any purpose that entails a violation of, or that is inconsistent with, the provisions of the regulations of the Board, including Regulation T, U, or X.
- 6.12 INVESTMENT COMPANY ACT; PUBLIC UTILITY HOLDING COMPANY ACT. No Credit Party is (a) an "investment company" as defined in, or subject to regulation under, the Investment Company Act of 1940, or (b) a "holding company" as defined in, or subject to regulation under, the Public Utility Holding Company Act of 1935.
- 6.13 TAX RETURNS. Each Credit Party has filed or caused to be filed all tax returns, reports, or materials required to have been filed by it and has paid or caused to be paid all taxes due and payable by it and all material written assessments received by it, except taxes that are being contested in good faith by appropriate proceedings and for which such Credit Party shall have set aside on its books adequate reserves.

6.14 NO MATERIAL MISSTATEMENTS. None of any information, report, financial statement, exhibit, or schedule furnished by or on behalf of any Credit Party to the Administrative Agent or any Lender in connection with the negotiation of this Agreement or any other Loan Paper or included therein or delivered pursuant thereto contained, contains or will contain any material misstatement of fact or omitted, omits or will omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were, are or will be made, not misleading; provided that to the extent any such information, report, financial statement, exhibit, or schedule was based upon or constitutes a forecast or projection, such Credit Party represents only that it acted in good faith and

34

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

utilized reasonable assumptions and due care in the preparation of such information, report, financial statement, exhibit, or schedule.

6.15 EMPLOYEE BENEFIT PLANS. The Parent and its ERISA Affiliates is in compliance in all respects with the applicable provisions of ERISA and the Code and the regulations and published interpretations thereunder, except where such non-compliance, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change. No ERISA Event has occurred or is reasonably expected to occur that, when taken together with all other such ERISA Events, could reasonably be expected to result in a Material Adverse Change.

6.16 ENVIRONMENTAL MATTERS

- (a) Except as set forth in SCHEDULE 6.16 and except with respect to any other matters that, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change, no Credit Party (i) has failed to comply with any Environmental Law or to obtain, maintain or comply with any permit, license or other approval required under any Environmental Law, (ii) has received notice of any claim with respect to any Environmental Liability, or (iii) knows of any basis for any Environmental Liability to which it is or reasonably could become subject.
- (b) Since the date hereof, there has been no change in the status of the matters disclosed on SCHEDULE 6.16 that, individually or in the aggregate, has resulted in, or materially increased the likelihood of, a Material Adverse Change.
- 6.17 LABOR MATTERS. As of the date hereof, there are no strikes, lockouts, or slowdowns against any Credit Party pending or, to the knowledge of the applicable Credit Party, threatened. Except with respect to any violations that, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change, the hours worked by and payments made to employees of each Credit Party have not been in violation of the Fair Labor Standards Act or any other applicable Laws dealing with such matters. All payments due from any Credit Party, or for which any claim may be made against any Credit Party, on account of wages and employee health and welfare insurance and other benefits, have been paid or accrued as a liability on the books of such Credit Party except where the failure to make or accrue any such payments, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Change. The consummation of the transactions contemplated hereby will not give rise to any right of termination or right of renegotiation on the part of any union under any collective bargaining agreement to which any Credit Party is bound.
- 6.18 SOLVENCY. After giving effect to this Agreement and the transactions contemplated hereby, (a) the fair value of the assets of each Credit Party, at a fair valuation, will exceed its debts and liabilities, subordinated, contingent or otherwise; (b) the present fair saleable value of the property of each Credit Party will be greater than the amount that will be required to pay the probable liability of its debts and other liabilities, subordinated, contingent or otherwise, as such debts and other liabilities become absolute and matured; (c) each Credit Party will be able to pay its debts and liabilities, subordinated, contingent or otherwise, as such debts and

liabilities become absolute and matured; and (d) each Credit Party will not have unreasonably small capital with which to conduct the business in which it is engaged as such business is now conducted and is proposed to be conducted following the Closing Date.

6.19 PROJECTIONS. As of the date hereof, there has been no change in the Projections delivered pursuant to SECTIONS $5.6\,(A)\,(I)$ and $5.6\,(B)\,(III)$ that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Change.

35

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

- 6.20 SENIOR INDEBTEDNESS. The Obligations of the Credit Parties hereunder constitute "Senior Indebtedness" under and as defined in the Dollar Subordinated Note Indenture and the Euro Subordinated Note Indenture.
- 6.21 CONSOLIDATED COVERAGE RATIO. After giving effect to the incurrence of Indebtedness under this Agreement and the EIB Loan, the Consolidated Coverage Ratio (as defined in the Dollar Subordinated Note Indenture) of the Parent is $2.77 \pm 0.1 \pm 0.00$
- 6.22 EIB LOAN. The EIB has approved the use of proceeds of the EIB Loan for the purpose of repaying outstanding senior Indebtedness of the Parent.

SECTION 7 AFFIRMATIVE COVENANTS OF THE CREDIT PARTIES

Each Credit Party covenants and agrees that so long as the Letter of Credit shall be outstanding, and until payment in full of the Obligation and termination of the Letter of Credit Commitment, each Credit Party will (unless such Credit Party receives prior written consent to the contrary from the Administrative Agent as authorized by the Determining Lenders), and always limited to the applicable Credit Party or the business of the applicable Credit Party rendering the covenant:

7.1 EXISTENCE; BUSINESSES AND PROPERTIES.

- (a) Do or cause to be done all things necessary to preserve, renew, and keep in full force and effect its legal existence, except as otherwise expressly permitted under SECTION 8.5 and except, with respect to any Subsidiary, where the failure to do so could not reasonably be expected, individually or in the aggregate, to result in a Material Adverse Change.
- (b) Do or cause to be done all things necessary to obtain, preserve, renew, extend, and keep in full force and effect the rights, licenses, permits, franchises, authorizations, patents, copyrights, trademarks, and trade names material to the conduct of the business of each Credit Party; maintain and operate such business in substantially the manner in which it is presently conducted and operated; comply in all material respects with all applicable Laws (including all Environmental Laws) and decrees and orders of any Governmental Authority, whether now in effect or hereafter enacted, except where the failure to do so could not reasonably be expected, individually or in the aggregate, to result in a Material Adverse Change; and at all times maintain and preserve all property material to the conduct of such business and keep such property in good repair, working order and condition and from time to time make, or cause to be made, all needful and proper repairs, renewals, additions, improvements, and replacements thereto necessary in order that the business carried on in connection therewith may be properly conducted at all times, except where the failure to do so could not reasonably be expected, individually or in the aggregate, to result in a Material Adverse Change.
- 7.2 INSURANCE. Keep its insurable properties adequately insured at all times by financially sound and reputable insurers; maintain such other insurance, to such extent and against such risks, including fire and other risks insured against by extended coverage, as is customary with companies in the same or similar businesses operating in the same or similar locations, including public liability insurance against claims for personal injury or death or property damage occurring upon, in, about or in connection with the use of any

properties owned, occupied or controlled by it; and maintain such other insurance as may be required by law.

7.3 OBLIGATIONS AND TAXES. Pay its Indebtedness promptly and in accordance with their terms and pay and discharge promptly when due all taxes, assessments, and governmental charges or

36

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

levies imposed upon it or upon its income or profits or in respect of its property, before the same shall become delinquent or in default, as well as all lawful claims for labor, materials, and supplies or otherwise that, if unpaid, might give rise to a Lien upon such properties or any part thereof; provided, however, that such payment and discharge shall not be required with respect to any such tax, assessment, charge, levy, or claim so long as (a) the validity or amount thereof shall be contested in good faith by appropriate proceedings, (b) such Credit Party shall have set aside on its books adequate reserves with respect thereto in accordance with GAAP, and (c) such contest operates to suspend collection of the contested obligation, tax, assessment, or charge and enforcement of a Lien.

7.4 FINANCIAL STATEMENTS, REPORTS, ETC.

- (a) In the case of the Parent, furnish to the Administrative Agent:
 - (i) concurrently with same being filed with the SEC, but in any event no later than 100 days after the end of each fiscal year, its consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Parent and its consolidated Subsidiaries as of the close of such fiscal year and the results of its operations and the operations of such Subsidiaries during such year, all audited by PricewaterhouseCoopers or other independent public accountants of recognized national standing and accompanied by an opinion of such accountants (which shall not be qualified in any material respect) to the effect that such consolidated financial statements fairly present the financial condition and results of operations of the Parent and its consolidated Subsidiaries on a consolidated basis in accordance with GAAP;
 - (ii) concurrently with same being filed with the SEC, but in any event no later than 50 days after the end of each of the first three fiscal quarters of each fiscal year, its consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Parent and its consolidated Subsidiaries as of the close of such fiscal quarter and the results of its operations and the operations of such Subsidiaries during such fiscal quarter and the then elapsed portion of the fiscal year, compared with the consolidated budget for such fiscal quarter as well as the results of its operations and the operations of its Subsidiaries in the corresponding quarter from the prior fiscal year, all certified by one of its Financial Officers as fairly presenting the financial condition and results of operations of the Parent and its consolidated Subsidiaries on a consolidated basis in accordance with GAAP, subject to normal year-end audit adjustments;
 - (iii) concurrently with any delivery of financial statements under CLAUSE (I) or (II) above, a certificate of a Financial Officer of the Parent certifying such statements and certifying that no Event of Default or Default has occurred or, if such an Event of Default or Default has occurred, specifying the nature and extent thereof and any corrective action taken or proposed to be taken with respect thereto, and, in the case of a certificate delivered with the financial statements required by CLAUSE (I) above, setting forth the

Parent's calculation of Excess Cash Flow;

(iv) at least 10 days prior to the commencement of each fiscal year of the Parent, a detailed consolidated budget for such fiscal year (including a projected consolidated balance sheet and related statements of projected operations and cash flow as of the end of and for each quarter of such fiscal year and as of the end of and for such fiscal year and describing the assumptions used for purposes of preparing such budget) and, promptly when available, any significant revisions of such budget;

37

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

- (v) promptly after the same become publicly available, copies of all periodic and other reports, proxy statements and other materials filed by the Parent with the SEC, or any Governmental Authority succeeding to any or all of the functions of said Commission, or with any national securities exchange, or distributed to its shareholders, as the case may be;
- (vi) promptly after the receipt thereof by the Parent, a copy of any final "management letter" received by any such person from its certified public accountants and the management's response thereto; and
- (vii) promptly, from time to time, such other information regarding the operations, business affairs, and financial condition of the Parent and its Subsidiaries, or compliance with the terms of this Agreement or any other Loan Paper, as the Administrative Agent or any Lender may reasonably request.
- (b) In the case of the Company, furnish to the Administrative Agent:
 - (i) concurrently with the financial statements delivered pursuant to SECTION 7.4(A)(I), its internally generated consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Company and its consolidated Subsidiaries as of the close of such fiscal year and the results of its operations and the operations of such Subsidiaries during such year, certified by a Responsible Officer of the Company and a Financial Officer of the Parent as fairly presenting the financial condition and results of operations of the Company and its consolidated Subsidiaries on a consolidated basis in accordance with GAAP;
 - (ii) within 210 days after the end of each fiscal year, its statutory consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Company and its consolidated Subsidiaries as of the close of such fiscal year and the results of its operations and the operations of such Subsidiaries during such year, all audited by PricewaterhouseCoopers or other independent public accountants of recognized national standing or other independent public accountants of recognized national standing and accompanied by an opinion of such accountants to the effect that such statutory consolidated financial statements fairly present the financial condition and results of operations of the Company and its consolidated Subsidiaries on a consolidated basis;
 - (iii) within 90 days following the delivery of the statutory financial statements set forth in SECTION 7.4(B)(II), a review prepared by PricewaterhouseCoopers or other independent public accountants of recognized national standing of the financial statements set forth in SECTION

(iv) within 90 days after the end of each of the first three fiscal quarters of each fiscal year, its consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of the Company and its consolidated Subsidiaries as of the close of such fiscal quarter and the results of its operations and the operations of such Subsidiaries during such fiscal quarter and the then elapsed portion of the fiscal year, all certified by a Responsible Officer of the Company and a Financial Officer of the Parent as fairly presenting the financial condition and

38

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

results of operations of the Company and its consolidated Subsidiaries on a consolidated basis in accordance with GAAP, subject to normal year-end audit adjustments;

- (v) concurrently with any delivery of financial statements under CLAUSE (I), (II), or (IV) above, a certificate of a Responsible Officer of the Company and a Financial Officer of the Parent certifying such statements and certifying that no Event of Default or Default has occurred or, if such an Event of Default or Default has occurred, specifying the nature and extent thereof and any corrective action taken or proposed to be taken with respect thereto;
- (vi) promptly, from time to time, such other information regarding the operations, business affairs, and financial condition of the Company, or compliance with the terms of any Loan Paper, as the Administrative Agent or any Lender may reasonably request; and
- (vii) no later than September 30, 2004, the Company shall provide the Administrative Agent its statutory audited consolidated financial statements for the fiscal year ended December 31, 2003, and as soon as available thereafter, its audited consolidated financial statements for the fiscal year ended December 31, 2003, prepared in accordance with GAAP, and reviewed by PricewaterhouseCoopers or other independent public accountants of recognized national standing.
- 7.5 Litigation and Other Notices. Furnish to the Administrative Agent, Issuing Lender, and each Lender prompt written notice of the following:
 - (a) any Event of Default or Default, specifying the nature and extent thereof and the corrective action (if any) taken or proposed to be taken with respect thereto;
 - (b) the filing or commencement of, or any threat or notice of intention of any person to file or commence, any action, suit or proceeding, whether at law or in equity or by or before any Governmental Authority, against such Credit Party or any Affiliate thereof that could reasonably be expected to result in a Material Adverse Change;
 - (c) the occurrence of any ERISA Event that, alone or together with any other ERISA Events that have occurred, could reasonably be expected to result in liability of such Credit Party in an aggregate amount exceeding the Dollar Equivalent Amount of \$5,000,000;
 - (d) any notice from S&P or Moody's indicating the possibility of an adverse change in the credit ratings applicable to the Parent or any of its Indebtedness assigned by S&P or Moody's and promptly after the Parent obtains knowledge of any change in the rating established by S&P or Moody's, as applicable, a notice of such change, which notice shall specify the new rating, the date on which such change was publicly announced, and such other information with respect to such

change as the Administrative Agent may reasonably request;

- (e) any development that has resulted in, or could reasonably be expected to result in, a Material Adverse Change.
- 7.6 Maintaining Records; Access to Properties and Inspections. Keep books of record and account in conformity with the generally accepted accounting principles in the jurisdiction of its principal place of business. Each Credit Party will permit any Representatives designated by the Administrative Agent or any Lender to visit and inspect the financial records and the properties of such Credit Party at

39

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

reasonable times and as often as reasonably requested and to make extracts from and copies of such financial records, and permit any Representatives designated by the Administrative Agent or any Lender to discuss the affairs, finances, and condition of such Credit Party with the officers thereof and independent accountants therefor; provided that any such visit or inspection does not interfere with the normal operation of such business conducted at the properties.

- 7.7 Use of Proceeds. Request the issuance of the Letter of Credit only for the purpose set forth in the preamble to this Agreement.
- 7.8 Further Assurances. Execute any and all further documents, financing statements, agreements, and instruments, and take all further action that may be required under applicable Law, or that the Administrative Agent may reasonably request, in order to effectuate the transactions contemplated by this Agreement and any other the Loan Paper.
- 7.9 Interest Rate Protection. The Parent shall maintain the lesser of (a) \$375,000,000 in principal amount, and (b) 50% of its long-term Indebtedness as Indebtedness bearing a fixed rate of interest, whether pursuant to Hedging Agreements (which shall be acceptable to the Administrative Agent) or otherwise.
- 7.10 EIB Loan. No later than thirty (30) days after the Closing Date, the proceeds of the EIB Loan shall be used to reduce outstanding senior Indebtedness of the Parent. The Company and the Parent shall, at all times, comply with the terms and provisions of the EIB Loan.

SECTION 8 NEGATIVE COVENANTS OF THE CREDIT PARTIES

Each Credit Party, as applicable, covenants and agrees that so long as the Letter of Credit shall be outstanding, and until payment in full of the Obligation and termination of the Letter of Credit Commitment, each Credit Party, as applicable, will not (unless such Credit Party receives a prior written consent to the contrary from the Administrative Agent as authorized by the Determining Lenders), and always limited to the applicable Credit Party or the business of the applicable Credit Party rendering the covenant:

- $8.1\ \mbox{INDEBTEDNESS}.$ Incur, create, assume, or permit to exist any Indebtedness, except:
 - (a) Indebtedness of any Credit Party for borrowed money existing on the date hereof and set forth in SCHEDULES 8.1(A), (D), (E), (F), and (G), and any extensions, renewals, or replacements of such Indebtedness to the extent the principal amount of such Indebtedness is not increased, the weighted average life to maturity of such Indebtedness is not decreased, such Indebtedness, if subordinated to the Obligations, remains so subordinated on terms not less favorable to the Lenders, and no Credit Party (unless the original obligor in respect of such Indebtedness) becomes an obligor with respect thereto;
 - (b) Indebtedness of any Credit Party created hereunder and under the other Loan Papers;
 - (c) intercompany Indebtedness of any Credit Party to the extent permitted by SECTION 8.4(c); provided, however, that the Company shall not incur any intercompany Indebtedness in excess of E150,000,000;

(d) Indebtedness of the Parent incurred to finance the acquisition, construction, or improvement of any fixed or capital assets, including any Indebtedness assumed in connection

40

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

with the acquisition of any such assets or secured by a Lien on any such assets prior to the acquisition thereof, and extensions, renewals, and replacements of any such Indebtedness that do not increase the outstanding principal amount thereof; provided that (i) such Indebtedness is incurred prior to or within 90 days after such acquisition or within 90 days after the completion of such construction or improvement, and (ii) the aggregate principal amount of Indebtedness permitted by this SECTION 8.1(d), when combined with the aggregate principal amount of all Capital Lease Obligations incurred pursuant to SECTION 8.1(e) and outstanding Indebtedness listed on SCHEDULES 8.1(d) and 8.1(e), shall not exceed \$25,000,000 at any time outstanding;

- (e) Capital Lease Obligations of the Parent in an aggregate principal amount, when combined with the aggregate principal amount of all Indebtedness incurred pursuant to SECTION 8.1(d) and outstanding Indebtedness listed on SCHEDULES 8.1(d) and 8.1(e), not in excess of \$25,000,000 at any time outstanding;
- (f) Indebtedness of the Parent under industrial revenue bonds in an aggregate principal amount, when combined with the outstanding principal amounts of industrial revenue bonds listed on SCHEDULE 8.1(f), not to exceed \$20,000,000 at any time outstanding;
- (g) Indebtedness incurred by Foreign Subsidiaries (other than the Credit Parties) in an aggregate principal amount, when combined with the outstanding principal amount of all Indebtedness listed on SCHEDULE 8.1(g), not to exceed the Dollar Equivalent Amount of \$25,000,000 at any time outstanding and guarantees thereof by other Foreign Subsidiaries (other than the Credit Parties);
- (h) with respect to the Parent, unsecured guarantees of Indebtedness of its Subsidiaries permitted to be incurred pursuant to this Agreement, provided that if such Indebtedness is subordinated to the Obligations, the guarantee thereof by the Parent shall be subordinated on terms no less favorable to the Lenders;
- (i) Indebtedness of (x) the Parent under the Dollar Subordinated Notes in an aggregate principal amount not to exceed \$188,500,000, and the senior subordinated guarantees thereof by any of the Credit Parties, and (y) FFBV under the Euro Subordinated Notes in an aggregate principal amount not to exceed E65,000,000, and the senior subordinated guarantees thereof by any of the Credit Parties, in each case less the amount of any prepayments or repurchases thereof after May 2, 2002;
- (j) Indebtedness of Finsub incurred pursuant to the Receivables Program Documentation in an amount not exceeding \$200,000,000 in the aggregate at any time outstanding;
- (k) with respect to the Parent, guarantees of loans, in an aggregate amount outstanding at any time not to exceed \$30,000,000, made by third parties to employees who are participants in the Parent's stock purchase program, if implemented, to enable such employees to purchase common stock of the Parent;
- (1) Indebtedness or other contingent obligations (including obligations as an account party under any letter of credit of any Credit Party), solely in respect of surety and performance bonds, bank guarantees, commercial and standby letters of credit, and similar obligations in respect of contractual obligations of any Credit Party, provided that such obligations are (i) incurred in the ordinary course of business of such Credit Party and (ii) except as expressly permitted under Section 8.2(k), unsecured;

- (m) Indebtedness of the Parent in respect of the ABN Standby Credit in an aggregate amount outstanding at any time not to exceed \$10,000,000;
- (n) Indebtedness of the Parent in respect of the Fifth Third Letters of Credit, but not any renewals, extensions, or replacements thereof:
- (o) Indebtedness of the Parent constituting New Subordinated Debt in an aggregate principal amount not to exceed \$250,000,000;
- (p) other unsecured Indebtedness of the Parent in an aggregate principal amount not exceeding \$25,000,000 at any time outstanding;
- (q) unsecured Indebtedness of the Company under Hedging Agreement; and $% \left(1\right) =\left(1\right) +\left(1\right) +\left($
- (r) other unsecured Indebtedness of the Company in an aggregate principal amount not exceeding E1,000,000 at any time outstanding.
- 8.2 LIENS. Create, incur, assume, or permit to exist any Lien on any property or assets (including stock or other securities of any Person) now owned or hereafter acquired by it or on any income or revenues or rights in respect of any thereof, except:
 - (a) Liens on property or assets of any Credit Party existing on the date hereof and set forth in SCHEDULE 8.2, inclusive of Liens otherwise permitted under the Parent Credit Agreement; provided that to the extent such Liens secure obligations, they shall secure only those obligations which they secure on the date hereof and any extensions, renewals, or replacements thereof to the extent the same are permitted under SECTION 8.1;
 - (b) any Lien existing on any property or asset prior to the acquisition thereof by any Credit Party; provided that (i) such Lien is not created in contemplation of or in connection with such acquisition, and (ii) such Lien does not apply to any other property or assets of any Credit Party;
 - (c) with respect to any Credit Party, Liens for taxes not yet due or which are being contested in compliance with SECTION 7.3;
 - (d) carriers', warehousemen's, mechanics', materialmen's, repairmen's, or other like Liens of any Credit Party arising in the ordinary course of business and securing obligations that are not due and payable or which are being contested in compliance with SECTION 7.3;
 - (e) in respect of any Credit Party, pledges and deposits made in the ordinary course of business in compliance with workmen's compensation, unemployment insurance, and other social security laws or regulations;
 - (f) deposits to secure the performance of bids, trade contracts (other than for Indebtedness), leases (other than Capital Lease Obligations), statutory obligations, surety and appeal bonds, performance bonds, and other obligations of a like nature incurred in the ordinary course of business of any Credit Party;
 - (g) zoning restrictions, easements, rights-of-way, restrictions on use of real property, and other similar encumbrances incurred in the ordinary course of business of any Credit Party which, in the aggregate, are not substantial in amount and do not materially detract from the value

of the property subject thereto or interfere with the ordinary conduct of the business of any Credit Party;

- (h) Liens on fixed or capital assets hereafter acquired (or, in the case of improvements, constructed) by any Credit Party; provided that (i) such Liens secure Indebtedness permitted by SECTION 8.1, (ii) such Liens are incurred, and the Indebtedness secured thereby is created, within 90 days after such acquisition (or construction), (iii) except in the case of Capital Lease Obligations, the Indebtedness secured thereby does not exceed 90% of the lesser of the cost or the fair market value of such fixed or capital asset at the time of such acquisition (or construction), and (iv) such Liens do not apply to any other property or assets of any Credit Party;
- (i) Liens on the property of Finsub incurred pursuant to the Receivables Program Documentation;
- (j) Liens arising out of judgments or awards that do not constitute an Event of Default under SECTION 9.1(i) or in respect of which any Credit Party shall in good faith be prosecuting an appeal or proceedings for review in respect of which there shall be secured a subsisting stay of execution pending such appeal or proceedings; provided that the aggregate amount of all such judgments or awards (and any cash and the fair market value of any property subject to such Liens) does not exceed the Dollar Equivalent Amount of \$10,000,000 at any time outstanding; and
- (k) Liens on assets of Foreign Subsidiaries (other than the Credit Parties); provided that (i) such Liens do not extend to, or encumber, the capital stock of any of the Parent's Subsidiaries (except as permitted by SECTION 8.2(a)), and (ii) such Liens extending to the assets of such Foreign Subsidiaries secure only Indebtedness (x) incurred by such Foreign Subsidiary pursuant to SECTION 8.1(g) or (y) of up to the Dollar Equivalent Amount of \$5,000,000 in the aggregate incurred by such Foreign Subsidiaries pursuant to SECTION 8.1(l).
- 8.3 SALE and Lease-Back Transactions. With respect to any Credit Party, enter into any arrangement, directly or indirectly, with any Person whereby it shall sell or transfer any property, real or personal, used or useful in its business, whether now owned or hereafter acquired, and thereafter rent or lease such property or other property which it intends to use for substantially the same purpose or purposes as the property being sold or transferred unless (a) the sale of such property is permitted by SECTION 8.5 and (b) any Capital Lease Obligations or Liens arising in connection therewith are permitted by SECTIONS 8.1 and 8.2, respectively.
- 8.4 INVESTMENTS, Loans and Advances. Purchase, hold, or acquire any capital stock, evidences of indebtedness or other securities of, make or permit to exist any loans or advances to, or make or permit to exist any investment or any other interest in, any other Person, except:
 - (a) (i) investments by the Parent existing on the date hereof in the capital stock or other equity interests of the Parent's Subsidiaries, (ii) investments by the Parent in an aggregate amount not to exceed \$30,000,000, the proceeds of which are used, directly or indirectly, to acquire from any third-party joint venture participant all such participant's interest in a joint venture, with the result that the joint venture becomes a wholly owned Subsidiary of the Parent, (iii) additional investments by the Parent in the capital stock or other equity interests of the Parent's Subsidiaries; provided that the aggregate amount of equity investments made after August 8, 2000 pursuant to this CLAUSE (iii) by the Parent in any of its Subsidiaries that is not a Credit Party shall not exceed \$25,000,000 at any time outstanding; and (iv) additional investments by the Parent in the Captive Insurance Company;

43

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

- (b) Permitted Investments of any Credit Party;
- (c) (i) intercompany loans and advances existing on the date

hereof and set forth on SCHEDULE 8.4 and renewals, refinancings, substitutions, and replacements of such intercompany loans and advances to the extent the aggregate principal amount of all such loans and advances is not increased and, no such renewal, refinancing, substitution, or replacement shall involve a cash transfer from any Credit Party to any Person that is not a Credit Party unless the loan or advance being renewed, refinanced, substituted, or replaced has been repaid in cash in an amount at least equal to such cash transfer; (ii) other loans or advances made by the Parent to any of its Subsidiaries and made by any of the Parent's Subsidiaries to the Parent or any other Subsidiary of the Parent; provided that the amount of such loans and advances made pursuant to this CLAUSE (ii) by the Parent to any of its Subsidiaries that are not Credit Parties shall not exceed the Dollar Equivalent Amount of \$25,000,000 at any time outstanding; (iii) other loans or advances made by the Parent to a Foreign Subsidiary ("OUTBOUND DEBTS"); provided that (A) within 10 Business Days of the making of any such loan or advance by the Parent to any of its Subsidiaries that is not a Credit Party, one or more of the Foreign Subsidiaries shall return an equivalent amount of funds to the Parent ("INBOUND DISTRIBUTIONS") by way of (x) a dividend or return of capital or (y) a loan or advance to the Parent, which loan or advance shall be subordinated in right of payment to the Obligations in a manner satisfactory to the Administrative Agent and (B) the aggregate amount of Outbound Debts outstanding at any time to any of the Parent's Subsidiaries that are not Credit Parties shall not exceed the aggregate amount of Inbound Distributions from such Subsidiaries that are not Credit Parties by more than the Dollar Equivalent Amount of \$100,000,000 at any time. In addition to the foregoing, the Parent or a wholly owned Subsidiary of the Parent may own the Defeased IRBs; and (iv) intercompany loans and advances made by and among the Company and its Subsidiaries for the purpose of achieving tax efficiencies;

- (d) investments received in connection with the bankruptcy or reorganization of, or settlement of delinquent accounts and disputes with, customers and suppliers, in each case in the ordinary course of business of any Credit Party;
- (e) the Parent and its Subsidiaries may make loans and advances in the ordinary course of business to their respective employees so long as the aggregate principal amount thereof at any time outstanding shall not exceed \$5,000,000;
- (f) the Parent may enter into Hedging Agreements that (i) are required by SECTION 7.9 or (ii) are not speculative in nature, are entered into in the ordinary course of business and are related to interest rate hedging for floating interest rate exposure or hedging of bookings, sales, income and dividends derived from the foreign operations of the Parent or any of its Subsidiaries or otherwise related to purchases from foreign suppliers;
- (g) the Parent or any of its Subsidiaries may acquire all or substantially all the assets of a Person or line of business of such Person, or not less than 100% of the capital stock or other equity interests of a Person (referred to herein as the "Acquired Entity"); provided that (i) such acquisition was not preceded by an unsolicited tender offer for such capital stock or other equity interest by, or proxy contest initiated by, the Parent or any of its Subsidiaries; (ii) the Acquired Entity shall be a going concern, shall be in a similar line of business as that of the Parent and its Subsidiaries as conducted during the current and most recent calendar year; and (iii) at the time of such transaction (A) both before and after giving effect thereto, no Event of Default or Default shall have occurred and be continuing; (B) the Parent would be in compliance with the covenants set forth in SECTIONS 8.11 and 8.12 and the Leverage Ratio shall be at least 25 basis points below the maximum Leverage Ratio permitted at the time pursuant to SECTION 8.13, in each case as of

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

the most recently completed period of four consecutive fiscal quarters ending prior to such transaction for which the financial statements and

certificates required by SECTION 7.4(a)(i) or 7.4(a)(ii) have been delivered or for which comparable financial statements have been filed with the SEC, after giving pro forma effect to such transaction and to any other event occurring after such period as to which pro forma recalculation is appropriate (including any other transaction described in this SECTION 8.4(g) occurring after such period) as if such transaction had occurred as of the first day of such period; (C) the Company would be in compliance with the covenant set forth in SECTION 8.13(b) as of the most recently completed period of four consecutive fiscal quarters ending prior to such transaction for which the financial statements and certificates required by SECTION 7.4(b)(i), 7.4(b)(ii), 7.4(b)(iv) or 7.4(b)(v) have been delivered, after giving pro forma effect to such transaction and to any other event occurring after such period as to which pro forma recalculation is appropriate (including any other transaction described in this SECTION 8.4(g) occurring after such period) as if such transaction had occurred as of the first day of such period, and (D) the aggregate of the consideration paid in connection with all such acquisitions (including any Indebtedness of the Acquired Entity that is assumed by the Parent or any of its Subsidiaries following such acquisition) shall not exceed (x) in the event that the Parent's Leverage Ratio calculated as provided under CLAUSE (iii) (B) above is greater than 2.5 to 1.0, \$50,000,000 for all such acquisitions, and (y) in the event that the Parent's Leverage Ratio calculated as provided under CLAUSE (iii) (B) above is equal to or less than 2.5 to 1.0, \$100,000,000 in any fiscal year of the Parent, which amount may be increased in any fiscal year (to an amount not to exceed \$150,000,000) by the amount of consideration for any such acquisition in such fiscal year consisting solely of common stock of the Parent (any acquisition of an Acquired Entity meeting all the criteria of this SECTION 8.4(g) being referred to herein as a "PERMITTED ACQUISITION"). All pro forma calculations required to be made pursuant to this SECTION 8.4(q) shall (i) include only those adjustments that would be permitted or required by Regulation S-X under the Securities Act of 1933, as amended, are reviewed by the Parent's independent certified public accountants and are based on reasonably detailed written assumptions reasonably acceptable to the Administrative Agent and (ii) be certified to by a Financial Officer as having been prepared in good faith based upon reasonable assumptions;

(h) in addition to investments permitted by CLAUSES (a) through (g) above, additional investments, loans, and advances (other than equity investments in Foreign Subsidiaries) by the Parent and its Subsidiaries (other than the Company and the Material Subsidiaries) so long as the aggregate amount invested, loaned, or advanced pursuant to Section 6.04(h) of the Original Credit Agreement and this CLAUSE (h) (determined without regard to any write-downs or write-offs of such investments, loans, and advances) does not exceed \$10,000,000 in the aggregate.

8.5 MERGERS, Consolidations, Sales of Assets and Acquisitions.

(a) Merge into or consolidate with any other Person, or permit any other Person to merge into or consolidate with it, or sell, transfer, lease, or otherwise dispose of (in one transaction or in a series of transactions) all or substantially all the assets of the Parent or any other Credit Party (whether now owned or hereafter acquired) or any capital stock of any Subsidiary of the Parent (except as permitted by SECTION 8.5(b), below), or purchase, lease, or otherwise acquire (in one transaction or a series of transactions) all or any substantial part of the assets of any other Person, except that (i) the Parent and any of its Subsidiaries may purchase and sell inventory in the ordinary course of business, (ii) if at the time thereof and immediately after giving effect thereto no Event of Default or Default shall have occurred and be continuing (x) any wholly owned Subsidiary of the Parent may merge into the Parent in a transaction in which the Parent is the surviving corporation and (y) any wholly owned Subsidiary of the Parent (other than the Company) may merge into or consolidate with any other wholly owned Subsidiary of the

Parent in a transaction in which the surviving entity is a wholly owned Subsidiary of the Parent (and a Credit Party, if the merged Subsidiary was a Credit Party) and no Person other than the Parent or a wholly owned Subsidiary of the Parent receives any consideration, (iii) the Parent and its Subsidiaries may make Permitted Acquisitions (including by way of merger of a Person or Persons into the Parent or any Subsidiary of the Parent), (iv) any Subsidiary of the Parent (other than the Company) may be liquidated if the assets and liabilities of such Subsidiary have been (or as a result of such liquidation are) assigned to and assumed by the Parent or another Subsidiary of the Parent (which must be a Credit Party if the liquidated Subsidiary is a Credit Party) in a manner permitted hereunder, (v) any Credit Party (other than the Company and the Parent) may sell, transfer, lease, or otherwise dispose of (in one transaction or a series of transactions) all or substantially all of the assets of such Credit Party to another Credit Party, and (vi) the Parent or any of its Subsidiaries may sell Program Receivables to Finsub, and Finsub may sell Program Receivables pursuant to the Receivables Program Documentation.

- (b) With respect to the Parent, engage in any Asset Sale otherwise permitted under SECTION 8.5(a) unless (i) such Asset Sale is for consideration at least 75% of which is cash, (ii) such consideration is at least equal to the fair market value (as certified by a Responsible Officer of the Parent or, in the case of an asset with a fair market value in excess of \$20,000,000, determined in good faith by the board of directors of the Parent) of the assets being sold, transferred, leased, or disposed of and (iii) the fair market value of all assets sold, transferred, leased, or disposed of pursuant to this SECTION 8.5(b) shall not exceed (x) \$50,000,000 in any fiscal year or (y) \$150,000,000 in the aggregate from and after the August 8, 2000, provided that the limitations set forth in this CLAUSE (iii) shall not apply to the non-recourse factoring of accounts receivable by Foreign Subsidiaries, provided that the aggregate outstanding amount of accounts receivable (assuming each such account receivable remains outstanding for the number of days provided in the applicable invoice for non-delinquent payment) at any time which have been so factored since August 8, 2000 shall not exceed \$50,000,000. Any Asset Sale by a Subsidiary of the Parent of all or substantially all of its assets and permitted by this SECTION $8.5\ \mathrm{may}$ be effected by a sale of all of the capital stock of such Subsidiary.
- (c) With respect to the Parent, engage in any Asset Swap otherwise permitted by SECTION 8.5(a) unless all of the following conditions are met: (i) such exchange complies with the definition of Asset Swap, (ii) if the fair market value of the assets transferred exceeds \$25,000,000, the board of directors of the Parent approves such exchange and the Parent secures an appraisal given by an unaffiliated third party in form and substance reasonably satisfactory to the Administrative Agent, (iii) the fair market value of all assets of the Parent and its Subsidiaries transferred pursuant to Asset Swaps since August 8, 2000 shall not exceed \$100,000,000 in the aggregate, and (iv) the fair market value of any property or assets received is at least equal to the fair market value of the property or assets so transferred.
- 8.6 DIVIDENDS and Distributions; Restrictive Agreements.
- (a) Declare or pay, directly or indirectly, any dividend or make any other distribution (by reduction of capital or otherwise), whether in cash, property, securities, or a combination thereof, with respect to any shares of its capital stock or directly or indirectly redeem, purchase, retire or otherwise acquire for value (or permit any Subsidiary of the Parent to purchase or acquire) any shares of any class of its capital stock or set aside any amount for any such purpose (a "RESTRICTED PAYMENT"); provided, however, that (i) the Company may declare and pay dividends or make other distributions ratably to its shareholders, so long as the FBV Fixed Charge Coverage Ratio for the four fiscal quarters then ended is not less than 1.10:1.00 after giving effect to such dividend or distribution payment, (ii) any Subsidiary of the Parent may

declare and pay dividends or make other distributions ratably to its shareholders, except as otherwise set forth in CLAUSE (i) above, (iii) so long as no Event of Default or Default shall have occurred and be continuing or would result therefrom, the Parent may repurchase shares of its capital stock owned by employees of the Parent or its Subsidiaries or make payments to employees of the Parent or its Subsidiaries in connection with the exercise of stock options, stock appreciation rights, or similar equity incentives or equity based incentives upon termination of employment or in connection with the death or disability of such employees, in an aggregate amount not to exceed \$5,000,000 in any fiscal year, (iv) so long as no Event of Default or Default shall have occurred and be continuing or would result therefrom, the Parent may repurchase shares of its capital stock for contribution to employee benefit plans maintained by the Parent and its Subsidiaries, in an aggregate amount not to exceed \$5,000,000 in any fiscal year and (v) so long as the Parent has obtained and maintains Investment Grade Ratings, and so long as no Event of Default or Default shall have occurred and be continuing or would result therefrom, the Parent may make Restricted Payments in an amount not to exceed in the aggregate 50% of the Consolidated Net Income accrued during the period (treated as one accounting period) from the beginning of the fiscal quarter immediately following the fiscal quarter during which the Investment Grade Ratings are obtained to the end of the most recent fiscal quarter for which financial statements have been delivered to the Lenders pursuant to SECTION 7.4(a)(i) or 7.4(a)(II) prior to the date of such Restricted Payment. Notwithstanding the foregoing, the Company may make a repatriation of funds from the Company to its shareholders of up to E70,000,000 from proceeds of the EIB Loan during the fiscal year ending December 31, 2004.

- (b) Enter into, incur or permit to exist any agreement or other arrangement that prohibits, restricts, or imposes any condition upon (i) the ability of the Parent or any of its Subsidiaries to create, incur, or permit to exist any Lien upon any of its property or assets to secure the Obligations, or (ii) the ability of any Subsidiary of the Parent to pay dividends or other distributions with respect to any shares of its capital stock (except as set forth in SECTION 8.6(a)(i)) or to make or repay loans or advances to the Parent or any of its Subsidiaries or to guarantee Indebtedness of the Company or any of its Subsidiaries; provided that (A) the foregoing shall not apply to restrictions and conditions imposed by Law or by any Loan Paper or Subordinated Note Document, (B) the foregoing shall not apply to customary restrictions and conditions contained in agreements relating to the sale of a Subsidiary of the Parent pending such sale, provided such restrictions and conditions apply only to the Subsidiary that is to be sold and such sale is permitted hereunder, (C) the foregoing shall not apply to restrictions and conditions imposed on Finsub under the Receivables Program Documentation, (D) the foregoing shall not apply to restrictions and conditions imposed on any Foreign Subsidiary by the terms of any Indebtedness of such Foreign Subsidiary permitted to be incurred hereunder, (E) CLAUSE (i) of the foregoing shall not apply to restrictions or conditions imposed by any agreement relating to secured Indebtedness permitted by this Agreement if such restrictions or conditions apply only to the property or assets securing such Indebtedness, and (F) CLAUSE (i) of the foregoing shall not apply to customary provisions in leases and other contracts restricting the assignment thereof.
- 8.7 TRANSACTIONS with Affiliates. Except for transactions by or among any Credit Party, sell or transfer any property or assets to, or purchase or acquire any property or assets from, or otherwise engage in any other transactions with, any of its Affiliates, except that: (a) any Credit Party may engage in any of the foregoing transactions in the ordinary course of business at prices and on terms and conditions not less favorable to such Credit Party than could be obtained on an arm's-length basis from unrelated third parties, (b) reasonable and customary fees may be paid to members of the board of directors, officers, employees, and consultants of any Credit Party for services rendered to such Credit Party in the ordinary course of business, together with customary indemnities in connection therewith and in accordance with applicable Law, (c) dividends and other payments permitted to be made under

SECTION 8.6 will be permitted, (d) the Parent, Finsub, and any other Subsidiaries of the Parent may consummate the transactions contemplated by the Receivables Program Documentation, (e) any Credit Party may enter into intercompany transactions permitted by SECTION 8.4, (f) any Credit Party may incur intercompany Indebtedness permitted by SECTION 8.1.

- 8.8 BUSINESS of the Credit Parties. Engage at any time in any business or business activity other than the business currently conducted by the Credit Parties and business activities reasonably incidental thereto; provided, however, that the Parent shall be entitled to create a wholly-owned Subsidiary engaged solely in the business of providing the insurance coverage required under SECTION 7.2 hereof solely to the Credit Parties, so long as such Subsidiary is adequately capitalized to satisfy the requirements of SECTION 7.2 and investments therein do not exceed \$1,000,000 in the aggregate (a "CAPTIVE INSURANCE COMPANY").
 - 8.9 Other Indebtedness and Agreements.
 - (a) Permit any waiver, supplement, modification, amendment, termination, or release of any indenture, instrument, or agreement pursuant to which any Material Indebtedness of any Credit Party is outstanding if the effect of such waiver, supplement, modification, amendment, termination, or release would materially increase the obligations of the obligor or confer additional material rights on the holder of such Indebtedness in a manner adverse to any Credit Party or the Lenders.
 - (b) (i) With respect to any Credit Party, make any distribution, whether in cash, property, securities, or a combination thereof, other than regular scheduled payments of interest as and when due (to the extent not prohibited by applicable subordination provisions), in respect of, or pay, or offer or commit to pay, or directly or indirectly redeem, repurchase, retire, or otherwise acquire for consideration, or set apart any sum for the aforesaid purposes, any subordinated Indebtedness (including the Subordinated Notes), or (ii) pay in cash any amount in respect of any Indebtedness or preferred equity interests that may at the obligor's option be paid in kind or in other securities; provided that the Parent was permitted to repurchase and/or prepay the Subordinated Notes to the extent provided in the final proviso of Section 2.13(c) of the Original Credit Agreement.
- 8.10 CAPITAL Expenditures. Until such time as the Parent obtains, and during any period in which the Parent does not maintain, Investment Grade Ratings, permit the aggregate amount of Capital Expenditures made by the Parent and its Subsidiaries in any fiscal year, to exceed \$65,000,000 in the aggregate.

The amount of permitted Capital Expenditures set forth above in respect of any fiscal year commencing with the fiscal year ending on December 31, 2004, shall be increased (but not reduced) by (i) the amount of unused permitted Capital Expenditures for the immediately preceding fiscal year less (ii) an amount equal to unused Capital Expenditures carried forward to such preceding fiscal year.

8.11 INTEREST Coverage Ratio. With respect to the Parent, permit its Interest Coverage Ratio for any period of four consecutive fiscal quarters, in each case taken as one accounting period, ending during any period set forth below to be less than the ratio set forth opposite such period below:

PERIOD	RATIO

From and including December 31, 2005 through and 3.75x including March 31, 2006

Thereafter 4.00x

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

48

8.12 Fixed Charge Coverage Ratio. With respect to the Parent, permit its Fixed Charge Coverage Ratio for any period of four consecutive fiscal quarters, in each case taken as one accounting period, ending during any period set forth below to be the less than the ratio set forth opposite such period below:

PERIOD	RATIO		
From and including September 30, 2000 through and including December 30, 2005	1.10 to 1.00		
From and including December 31, 2005 through and including September 29, 2006	1.15 to 1.00		
Thereafter	1.20 to 1.00		

8.13 MAXIMUM Leverage Ratio.

(a) With respect to the Parent, permit its Leverage Ratio at any time during a period set forth below to be greater than the ratio set forth opposite such period below:

PERIOD 		
From and including June 30, 2002 through and including September 29, 2004	4.00x	
From and including September 30, 2004 through and including March 30, 2005 $$	3.75x	
From and including March 31, 2005 through and including September 29, 2005	3.50x	
From and including September 30, 2005 through and including December 30,	3.25x	
2005 Thereafter	3.00x	

⁽b) With respect to the Company, permit the FBV Leverage Ratio at any time for the four consecutive fiscal quarters most recently ended to be greater than 2.25 x.

^{8.14} DESIGNATED Senior Indebtedness. With respect to the Parent, designate any Indebtedness as "Designated Senior Indebtedness" for purposes of either the Dollar Subordinated Note Indenture or the Euro Subordinated Note Indenture.

8.15 FISCAL Year. With respect to the Parent, change its fiscal year-end to a date other than December 31; provided that the Parent may use a 52/53 week fiscal year ending around December 31.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

49

SECTION 9 EVENTS OF DEFAULT

- 9.1 EVENTS OF DEFAULT. An "EVENT OF DEFAULT" shall exist if any one or more of the following events (herein collectively called "EVENTS OF DEFAULT") shall occur and be continuing:
 - (a) any representation or warranty made or deemed made in or in connection with this Agreement or any other Loan Paper, or any representation, warranty, statement or information contained in any report, certificate, financial statement, or other instrument furnished in connection with or pursuant to this Agreement or any other Loan Paper, shall prove to have been false or misleading in any material respect when so made or furnished;
 - (b) default shall be made in the payment of the amount of any Drawing when the same becomes due and payable pursuant to SECTION 3.1 hereof:
 - (c) default shall be made in the payment of any interest on the Obligation or any fees set forth in SECTION 3.2, when and as the same shall become due and payable, and such default shall continue unremedied for a period of three Business Days;
 - (d) default shall be made in the due observance or performance by any Credit Party of any covenant, condition, or agreement contained in SECTION 7.1(a), 7.5, or 7.7 or in SECTION 8;
 - (e) default shall be made in the due observance or performance by any Credit Party of any covenant, condition, or agreement contained in any Loan Paper (other than those specified in CLAUSES (b), (c), or (d) above) and such default shall continue unremedied for a period of 15 days after notice thereof from the Administrative Agent or any Lender to the Company;
 - (f) any Credit Party shall: (i) fail to pay any principal or interest, regardless of amount, due in respect of any Material Indebtedness when and as the same shall become due and payable (after giving effect to any applicable grace period), or (ii) fail to observe or perform any other term, covenant, condition or agreement contained in any agreement or instrument evidencing or governing any such Indebtedness if the effect of any failure referred to in this CLAUSE (II) is to cause, or to permit the holder or holders of such Indebtedness or a trustee on its or their behalf to cause, such Indebtedness to become due prior to its stated maturity;
 - (g) an involuntary proceeding shall be commenced or an involuntary petition shall be filed in a court of competent jurisdiction seeking: (i) relief in respect of any Credit Party, or of a substantial part of the property or assets of any Credit Party, under Title 11 of the United States Code, as now constituted or hereafter amended, or any other federal, state or foreign bankruptcy, insolvency, receivership, or similar law, (ii) the appointment of a receiver, trustee, custodian, sequestrator, conservator, or similar official for any Credit Party or for a substantial part of the property or assets of any Credit Party, or (iii) the winding-up or liquidation of any Credit Party; and such proceeding or petition shall continue undismissed for 60 days or an order or decree approving or ordering any of the foregoing shall be entered;
 - (h) any Credit Party shall: (i) voluntarily commence any proceeding or file any petition seeking relief under Title 11 of the United States Code, as now constituted or hereafter amended, or any other federal, state or foreign bankruptcy, insolvency, receivership,

or similar law, (ii) consent to the institution of, or fail to contest in a timely and appropriate manner, any proceeding or the filing of any petition described in CLAUSE (g) above, (iii) apply for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for any Credit Party or for a substantial part of the property or assets of any Credit Party, (iv) file an answer admitting the material allegations of a petition filed against it in any such proceeding, (v) make a general assignment for the benefit of creditors, (vi) become unable, admit in writing its

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

50

inability or fail generally to pay its debts as they become due, or (vii) take any action for the purpose of effecting any of the foregoing;

- (i) one or more judgments for the payment of money in an aggregate amount in excess of the Dollar Equivalent Amount of \$10,000,000 shall be rendered against any Credit Party or any combination thereof and the same shall remain undischarged for a period of 30 consecutive days during which execution shall not be effectively stayed, or any action shall be legally taken by a judgment creditor to levy upon assets or properties of any Credit Party to enforce any such judgment;
- (j) an ERISA Event shall have occurred that, in the reasonable opinion of the Determining Lenders, when taken together with all other such ERISA Events, could reasonably be expected to result in a Material Adverse Change;
- (k) the Guaranty by any Guarantor for any reason shall cease to be in full force and effect (other than in accordance with its terms), any Guarantor fails to pay on the Guaranty in respect of the Obligation, or any Guarantor shall deny in writing that it has any further liability under the Guaranty (other than as a result of the discharge of such Guarantor in accordance with the terms of this Agreement);
- (1) the Parent shall cease to be the beneficial owner of 100% of the voting stock of the Company;
- (m) any of the Obligations shall cease to constitute "Senior Indebtedness" under and as defined in the Dollar Subordinated Note Indenture or the Euro Subordinated Note Indenture; or
- (n) the occurrence and continuance of an "Event of Default" under the Parent Credit Agreement, or the occurrence and continuance of an event which gives rise to the right of the EIB to demand repayment under Section 10.01 of the EIB Loan.
- 9.2 REMEDIES UPON EVENT OF DEFAULT. If an Event of Default shall occur and be continuing, then: (a) the Administrative Agent may (and, subject to the terms of SECTION 10, shall upon the request of Determining Lenders) or Determining Lenders may declare the Letter of Credit Commitment and the obligation of Issuing Lender to issue the Letter of Credit on or after the date thereof terminated, whereupon the same shall forthwith terminate; (b) the Company shall pay to the Administrative Agent (for the Pro Rata benefit of the Lenders) immediately after the notice referred to in SECTION 3.8 hereof the full Available Amount of the Letter of Credit, regardless of whether any Drawing has been made under the Letter of Credit, which sum shall be placed in the Cash Collateral Account with the Administrative Agent for the payment of any future Drawings; (c) the Company shall pay to the Administrative Agent immediately and upon written notice and demand from the Administrative Agent to the Company the unpaid amount of any of the remainder of the Obligation then due and owing to the Administrative Agent, Issuing Lender, and the Lenders hereunder; and (d) the Administrative Agent may (and, subject to the terms of SECTION 10, shall upon the request of Determining Lenders) or Determining Lenders may (i) reduce any claim to judgment, and/or (ii) pursue and enforce any of the Administrative Agent's, Issuing Lender's, or any Lender's Rights under this Agreement, or

otherwise provided under or pursuant to any applicable Law or agreement; provided, however, that if any Event of Default specified in SECTION 9.1(g) or 9.1(h) hereof shall occur, the Obligation (including the obligation to pay the amount set forth in CLAUSES (b) and (c) above) shall thereupon become due and payable concurrently therewith, without any further action by Administrative Agent, Issuing Lender, or any Lender and without presentment, demand, protest, notice of default, notice of acceleration or of intention to accelerate, or other notice of any kind other than as expressly provided herein, all of which the Company hereby

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

51

expressly waives to the fullest extent permitted by applicable Law, other than such notice as is expressly required herein.

SECTION 10 AGREEMENT AMONG LENDERS

10.1 ADMINISTRATIVE AGENT.

- (a) Each Lender hereby appoints Calyon New York Branch (and Calyon New York Branch, hereby accepts such appointment) as its nominee and agent, in its name and on its behalf: (i) to act as nominee for and on behalf of such Lender in and under all Loan Papers; (ii) to arrange the means whereby the funds of the Lenders are to be made available to or for the benefit of the Company under the Loan Papers; (iii) to take such action as may be requested by any Lender under the Loan Papers (when such Lender is entitled to make such request under the Loan Papers and after such requesting Lender has obtained the concurrence of such other Lenders as may be required under the Loan Papers); (iv) to receive all documents and items to be furnished to Lenders under the Loan Papers; (v) to be the secured party, mortgagee, beneficiary, and similar party in respect of, and to receive, as the case may be, any collateral for the benefit of Lenders; (vi) to timely distribute, and the Administrative Agent agrees to so distribute, to each Lender all material information, requests, documents, and items received from the Company under the Loan Papers; (vii) to promptly distribute to each Lender its ratable part of each payment or prepayment (whether voluntary, as proceeds of collateral upon or after foreclosure, as proceeds of insurance thereon, or otherwise) in accordance with the terms of the Loan Papers; (viii) to deliver to the appropriate Persons requests, demands, approvals, and consents received from the Lenders; and (ix) to execute, on behalf of the Lenders, such releases or other documents or instruments as are permitted by the Loan Papers or as directed by the Lenders or Determining Lenders (when entitled to so authorize) from time to time; provided, however, the Administrative Agent shall not be required to take any action which exposes the Administrative Agent to personal liability or which is contrary to the Loan Papers or applicable Law.
- (b) The Administrative Agent may resign at any time as the Administrative Agent under the Loan Papers by giving written notice thereof to the Lenders. Should the initial or any successor Administrative Agent ever cease to be a party hereto or should the initial or any successor Administrative Agent ever resign as the Administrative Agent, then Determining Lenders shall elect the successor Administrative Agent from among the Lenders (other than the resigning Administrative Agent). If no successor Administrative Agent shall have been so appointed by Determining Lenders, within 30 days after the retiring Administrative Agent's giving of notice of resignation, then the retiring Administrative Agent may, on behalf of Lenders, appoint a successor Administrative Agent, which shall be a commercial bank having a combined capital and surplus of at least \$1,000,000,000. Upon the acceptance of any appointment as the Administrative Agent under the Loan Papers by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the Rights of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations of Administrative Agent under the Loan Papers, and each Lender shall execute such

documents as any Lender may reasonably request to reflect such change in and under the Loan Papers. After any retiring Administrative Agent's resignation as the Administrative Agent under the Loan Papers, the provisions of this SECTION 10 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was the Administrative Agent under the Loan Papers.

(c) The Administrative Agent, in its capacity as a Lender, shall have the same Rights under the Loan Papers as any other Lender and may exercise the same as though it were not

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

52

acting as the Administrative Agent; the term "Lender" shall, unless the context otherwise indicates, include the Administrative Agent; and any resignation of the Administrative Agent hereunder shall not impair or otherwise affect any Rights which it has or may have in its capacity as an individual Lender. Each Lender and each Credit Party agrees that the Administrative Agent is not a fiduciary for the Lenders or for the Credit Parties but simply is acting in the capacity described herein to alleviate administrative burdens for both the Credit Parties and the Lenders, that the Administrative Agent has no duties or responsibilities to the Lenders or the Credit Parties except those expressly set forth herein, and that the Administrative Agent in its capacity as a Lender has all Rights of any other Lender.

- (d) The Administrative Agent and its Affiliates may now or hereafter be engaged in one or more loan, letter of credit, leasing, or other financing transactions with the Credit Parties, act as trustee or depositary for the Credit Parties, or otherwise be engaged in other transactions with the Credit Parties (collectively, the "other activities") not the subject of the Loan Papers. Without limiting the Rights of the Lenders specifically set forth in the Loan Papers, the Administrative Agent and its Affiliates shall not be responsible to account to the Lenders for such other activities, and no Lender shall have any interest in any other activities, any present or future guaranties by or for the account of the Credit Parties which are not contemplated or included in the Loan Papers, any present or future offset exercised by the Administrative Agent and its Affiliates in respect of such other activities, any present or future property taken as security for any such other activities, or any property now or hereafter in the possession or control of the Administrative Agent or its Affiliates which may be or become security for the obligations of the Credit Parties arising under the Loan Papers by reason of the general description of indebtedness secured or of property contained in any other agreements, documents or instruments related to any such other activities; provided that, if any payments in respect of such guaranties or such property or the proceeds thereof shall be applied to reduction of the obligations of the Credit Parties arising under the Loan Papers, then each Lender shall be entitled to share in such application ratably.
- 10.2 EXPENSES. Upon demand by the Administrative Agent, each Lender shall pay its Pro Rata Part of any reasonable expenses (including, without limitation, court costs, reasonable attorneys' fees and other costs of collection) incurred by the Administrative Agent in connection with any of the Loan Papers if and to the extent the Administrative Agent does not receive reimbursement therefor from other sources within 60 days after incurred; provided that, each Lender shall be entitled to receive its Pro Rata Part of any reimbursement for such expenses, or part thereof, which the Administrative Agent subsequently receives from such other sources.
- 10.3 PROPORTIONATE ABSORPTION OF LOSSES. Except as otherwise provided in the Loan Papers, nothing in the Loan Papers shall be deemed to give any Lender any advantage over any other Lender insofar as the Obligation arising under the Loan Papers is concerned, or to relieve any Lender from absorbing its Pro Rata Part of any losses sustained with respect to the Obligation (except to the extent such losses result from unilateral actions or inactions of any Lender that are not made in accordance with the terms and provisions of the Loan

Papers).

10.4 DELEGATION OF DUTIES; RELIANCE. The Administrative Agent may perform any of its duties or exercise any of its Rights under the Loan Papers by or through its Representatives. The Administrative Agent and its Representatives shall: (a) be entitled to rely upon (and shall be protected in relying upon) any writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telecopy, telegram, telex or teletype message, statement, order, or other documents or conversation believed by it or them to be genuine and correct and to have been signed or made by the proper Person and, with respect to legal matters, upon opinion of counsel selected by the Administrative Agent, (b) be entitled to deem and

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

53

treat each Lender as the owner and holder of the Obligation owed to such Lender for all purposes until, subject to SECTION 11.12, written notice of the assignment or transfer thereof shall have been given to and received by the Administrative Agent (and any request, authorization, consent, or approval of any Lender shall be conclusive and binding on each subsequent holder, assignee, or transferee of the Obligation owed to such Lender or portion thereof until such notice is given and received), (c) not be deemed to have notice of the occurrence of an Event of Default unless a responsible officer of the Administrative Agent, who handles matters associated with the Loan Papers and transactions thereunder, has actual knowledge thereof or the Administrative Agent has been notified thereof by a Lender or the Company, and (d) be entitled to consult with legal counsel (including counsel for the Company), independent accountants and other experts selected by the Administrative Agent and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such counsel, accountants or experts.

10.5 LIMITATION OF LIABILITY.

- (a) Neither the Administrative Agent nor any of its Representatives shall be liable for any action taken or omitted to be taken by it or them under the Loan Papers in good faith and reasonably believed by it or them to be within the discretion or power conferred upon it or them by the Loan Papers or be responsible for the consequences of any error of judgment, except for gross negligence or willful misconduct; and neither the Administrative Agent nor any of its Representatives has any fiduciary relationship with any Lender by virtue of the Loan Papers (provided that nothing herein shall negate the obligation of the Administrative Agent to account for funds received by it for the account of any Lender).
- (b) Unless indemnified to its satisfaction against loss, cost, liability, and expense, the Administrative Agent shall not be compelled to do any act under the Loan Papers or to take any action toward the execution or enforcement of the powers thereby created or to prosecute or defend any suit in respect of the Loan Papers. If the Administrative Agent requests instructions from the Lenders or Determining Lenders, as the case may be, with respect to any act or action (including, but not limited to, any failure to act) in connection with any Loan Paper, the Administrative Agent shall be entitled (but shall not be required) to refrain (without incurring any liability to any Person by so refraining) from such act or action unless and until it has received such instructions. In no event, however, shall the Administrative Agent or any of its Representatives be required to take any action which it or they determine could incur for it or them criminal or onerous civil liability. Without limiting the generality of the foregoing, no Lender shall have any right of action against the Administrative Agent as a result of the Administrative Agent's acting or refraining from acting hereunder in accordance with the instructions of Determining Lenders.
- (c) The Administrative Agent shall not be responsible in any manner to any Lender or any Participant for, and each Lender represents and warrants that it has not relied upon the Administrative Agent in respect of, (i) the creditworthiness of the Credit Parties and the risks involved to such Lender, (ii) the effectiveness, enforceability,

genuineness, validity, or the due execution of any Loan Paper, (iii) any representation, warranty, document, certificate, report, or statement made therein or furnished thereunder or in connection therewith, (iv) the existence, priority, or perfection of any Lien hereafter granted or purported to be granted under any Loan Paper, or (v) observation of or compliance with any of the terms, covenants, or conditions of any Loan Paper on the part of any of the Credit Parties. Each Lender agrees to indemnify the Administrative Agent and its Representatives and hold them harmless from and against (but limited to such Lender's Pro Rata Part of) any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, reasonable expenses, and reasonable disbursements of any kind or nature whatsoever which may be imposed on, asserted against, or incurred by them in

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

54

any way relating to or arising out of the Loan Papers or any action taken or omitted by them under the Loan Papers, to the extent the Administrative Agent and its Representatives are not reimbursed for such amounts by the Credit Parties (provided that, neither the Administrative Agent nor its Representatives shall have the right to be indemnified hereunder for its or their own gross negligence or willful misconduct).

10.6 DEFAULT; COLLATERAL. Upon the occurrence and continuance of an Event of Default, the Lenders agree to promptly confer in order that Determining Lenders or the Lenders, as the case may be, may agree upon a course of action for the enforcement of the Rights of Lenders; and the Administrative Agent shall be entitled to refrain from taking any action (without incurring any liability to any Person for so refraining) unless and until the Administrative Agent shall have received instructions from Determining Lenders. In actions with respect to any property of the Credit Parties, the Administrative Agent is acting for the ratable benefit of each Lender. Any and all agreements to subordinate (whether made heretofore or hereafter) other Indebtedness or obligations of the Company to the Obligation shall be construed as being for the ratable benefit of each Lender. If the Administrative Agent acquires any security for the Obligation upon or in lieu of foreclosure, the same shall be held for the Pro Rata benefit of all Lenders.

- 10.7 LIMITATION OF LIABILITY. To the extent permitted by Law: (a) the Administrative Agent (acting in its agent capacity) shall not incur any liability to any other Lender or Participant except for acts or omissions resulting from its own gross negligence or willful misconduct, and (b) neither the Administrative Agent, any Lender, nor any Participant shall incur any liability to any other Person for any act or omission of any other Lender or Participant.
- $10.8\ \text{RELATIONSHIP}$ OF LENDERS. Nothing herein shall be construed as creating a partnership or joint venture among the Administrative Agent and the Lenders.
- 10.9 BENEFITS OF AGREEMENT. Except for the representations and covenants in SECTION 10.1(c) in favor of the Credit Parties, none of the provisions of this SECTION 10 shall inure to the benefit of the Credit Parties or any other Person other than Lenders; consequently, neither the Credit Parties nor any other Person shall be entitled to rely upon, or to raise as a defense, in any manner whatsoever, the failure of the Administrative Agent or Lender to comply with such provisions.
- 10.10 OBLIGATION SEVERAL. The obligations of the Lenders hereunder are several, and each Lender hereunder shall not be responsible for the obligations of the other Lenders hereunder, nor will the failure of one Lender to perform any of its obligations hereunder relieve the other Lenders from the performance of their respective obligations hereunder.

SECTION 11 MISCELLANEOUS

11.1 HEADINGS. The headings, captions, and arrangements used in any of

the Loan Papers are, unless specified otherwise, for convenience only and shall not be deemed to limit, amplify, or modify the terms of the Loan Papers, nor affect the meaning thereof.

11.2 NONBUSINESS DAYS. In any case where any payment or action is due under any Loan Paper on a day which is not a Business Day, such payment or action may be delayed until the next-succeeding Business Day, but interest and fees shall continue to accrue in respect of any payment to which it is applicable until such payment is in fact made.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

55

- 11.3 COMMUNICATIONS. Unless specifically otherwise provided, whenever any Loan Paper requires or permits any consent, approval, notice, request, or demand from one party to another, such communication must be in writing (which may be by telex or telecopy) to be effective and shall be deemed to have been given: (a) if by telex, when transmitted to the telex number, if any, for such party, and the appropriate answer back is received, (b) if by telecopy, when transmitted to the telecopy number for such party (and all such communications sent by telecopy shall be confirmed promptly thereafter by personal delivery or mailing in accordance with the provisions of this section; provided that any requirement in this parenthetical shall not affect the date on which such telecopy shall be deemed to have been delivered), (c) if by certified (return receipt) or registered mail, shall be deemed to have been given when received, and (d) if by hand or overnight courier service, shall be deemed to have been given when received. Until changed by notice pursuant hereto, the address (and telex and telecopy numbers, if any) for the Administrative Agent and each Lender is set forth on SCHEDULE 2.1, and for each Credit Party is the address set forth by such Credit Party's signature on the signature page of this Agreement.
- 11.4 FORM AND NUMBER OF DOCUMENTS. Each agreement, document, instrument, or other writing to be furnished under any provision of this Agreement must be in form and substance and in such number of counterparts as may be reasonably satisfactory to the Administrative Agent and its counsel.
- 11.5 SURVIVAL. All covenants, agreements, undertakings, representations, and warranties made in any of the Loan Papers shall survive all closings under the Loan Papers and, except as otherwise indicated, shall not be affected by any investigation made by any party. All rights of, and provisions relating to, reimbursement and indemnification of the Administrative Agent, Issuing Lender or any Lender shall survive termination of this Agreement and payment in full of the Obligation.
- 11.6 Governing Law. THE LAWS OF THE STATE OF NEW YORK AND OF THE UNITED STATES OF AMERICA SHALL GOVERN THE RIGHTS AND DUTIES OF THE PARTIES TO THE LOAN PAPERS AND THE VALIDITY, CONSTRUCTION, ENFORCEMENT, AND INTERPRETATION OF THE LOAN PAPERS.
- 11.7 INVALID PROVISIONS. If any provision in any Loan Paper is held to be illegal, invalid, or unenforceable, such provision shall be fully severable; the appropriate Loan Paper shall be construed and enforced as if such provision had never comprised a part thereof; and the remaining provisions thereof shall remain in full force and effect and shall not be affected by such provision or by its severance therefrom. The Administrative Agent, Issuing Lender, the Lenders, and the Credit Parties agree to negotiate, in good faith, the terms of a replacement provision as similar to the severed provision as may be possible and be legal, valid, and enforceable.
- 11.8 Entirety. THE RIGHTS AND OBLIGATIONS OF THE CREDIT PARTIES, LENDERS, ISSUING LENDER, AND THE ADMINISTRATIVE AGENT SHALL BE DETERMINED SOLELY FROM WRITTEN AGREEMENTS, DOCUMENTS, AND INSTRUMENTS, AND ANY PRIOR ORAL AGREEMENTS BETWEEN SUCH PARTIES ARE SUPERSEDED BY AND MERGED INTO SUCH WRITINGS. THIS AGREEMENT (AS AMENDED IN WRITING FROM TIME TO TIME) AND THE OTHER WRITTEN LOAN PAPERS EXECUTED BY THE CREDIT PARTIES, ANY LENDER, ISSUING LENDER, OR THE ADMINISTRATIVE AGENT (TOGETHER WITH ALL COMMITMENT LETTERS AND FEE LETTERS AS THEY RELATE TO THE PAYMENT OF FEES AFTER THE CLOSING DATE) REPRESENT THE FINAL AGREEMENT BETWEEN THE CREDIT PARTIES, LENDERS, ISSUING LENDER, AND THE ADMINISTRATIVE AGENT AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS BY SUCH PARTIES. THERE ARE NO

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

56

11.9 JURISDICTION; VENUE; SERVICE OF PROCESS; JURY TRIAL. EACH PARTY HERETO, IN EACH CASE FOR ITSELF, ITS SUCCESSORS AND ASSIGNS, HEREBY: (A) IRREVOCABLY SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE BOROUGH OF MANHATTAN OR SOUTHERN DISTRICT OF NEW YORK, AND AGREES AND CONSENTS THAT SERVICE OF PROCESS MAY BE MADE UPON IT IN ANY LEGAL PROCEEDING ARISING OUT OF OR IN CONNECTION WITH THE LOAN PAPERS AND THE OBLIGATION BY SERVICE OF PROCESS AS PROVIDED BY NEW YORK LAW, (B) IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY LITIGATION ARISING OUT OF OR IN CONNECTION WITH THE LOAN PAPERS AND THE OBLIGATION BROUGHT IN ANY SUCH COURT, (C) IRREVOCABLY WAIVES ANY CLAIMS THAT ANY LITIGATION BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM, (D) IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH LITIGATION BY THE MAILING OF COPIES THEREOF BY CERTIFIED MAIL, RETURN RECEIPT REQUESTED, POSTAGE PREPAID, TO ITS PROCESS AGENT SET FORTH BELOW, AND (E) IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF ANY LOAN PAPER OR THE TRANSACTIONS CONTEMPLATED THEREBY. In order to effectuate such submission to jurisdiction, each Credit Party hereby agrees that service of all writs, process, and summonses in any such suit, action, or proceeding brought in the State of New York may be brought upon its process agent appointed below. Each Credit Party hereby irrevocably appoints CT Corporation, 111 Eighth Avenue, 13th Floor, New York, NY 10011, its process agent, as its true and lawful attorney-in-fact in the name, place and stead of such Credit Party to accept such service of any and all such writs, process, and summonses, and each Credit Party agrees that the failure of such process agent to notify any Credit Party of the service of such writ, process, or summons shall not impair the validity of such service or any judgment based thereon. The scope of each of the foregoing waivers is intended to be all-encompassing of any and all disputes that may be filed in any court and that relate to the subject matter of this transaction, including, without limitation, contract claims, tort claims, breach of duty claims, and all other common law and statutory claims. The Credit Parties and each other party to this Agreement acknowledge that this waiver is a material inducement to the agreement of each party hereto to enter into a business relationship, that each has already relied on this waiver in entering into this agreement, and each will continue to rely on each of such waivers in related future dealings. The Credit Parties and each other party to this Agreement warrant and represent that they have reviewed these waivers with their legal counsel, and that they knowingly and voluntarily agree to each such waiver following consultation with legal counsel. THE WAIVERS IN THIS SECTION 11.9 ARE IRREVOCABLE, MEANING THAT THEY MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THESE WAIVERS SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, SUPPLEMENTS, AND REPLACEMENTS TO OR OF THIS AGREEMENT OR ANY OTHER LOAN PAPER. In the event of litigation, this Agreement may be filed as a written consent to a trial by the court.

11.10 AMENDMENTS, CONSENTS, CONFLICTS, AND WAIVERS.

- (a) Except as otherwise specifically provided, (i) this Agreement may only be amended, modified or waived by an instrument in writing executed jointly by the Company and Determining Lenders, and, in the case of any matter affecting the Administrative Agent, by the Administrative Agent, and may only be supplemented by documents delivered or to be delivered in accordance with the express terms hereof, and (ii) the other Loan Papers may only be the subject of an amendment, modification, or waiver if the Company and Determining Lenders, and, in the case of any matter affecting the Administrative Agent, the Administrative Agent, have approved same.
- (b) Any amendment to or consent or waiver under this Agreement or any Loan Paper which purports to accomplish any of the following must be by an instrument in writing executed

by the Company and executed (or approved, as the case may be) by each Lender, and, in the case of any matter affecting the Administrative Agent or Issuing Lender, by the Administrative Agent or Issuing Lender, as applicable: (i) extends the Credit Expiration Date or the due date of any scheduled payment of the Obligation (including interest and fees) arising under the Loan Papers beyond the date specified in the Loan Papers; (ii) reduces the interest rate or decreases the amount of any scheduled payment, interest, fees, or other sums payable to the Administrative Agent or the Lenders hereunder (except such reductions as are contemplated by this Agreement); (iii) changes the percentage specified in the definition of "DETERMINING LENDERS" or any other provision hereof specifying the number or percentage of the Lenders which are required to amend, waive, or modify any rights hereunder or otherwise make any determination or grant any consent hereunder; (iv) changes this CLAUSE (b) or any other matter specifically requiring the consent of all Lenders hereunder; (v) consents to the assignment or transfer of the Company of any of its rights and obligations under this Agreement; (vi) releases any Guarantor from the Guaranty; (vii) change SECTION 2.1(c) in any manner that would alter the Pro Rata sharing of payments required thereby; or (viii) change SECTION 3.8 or any other provision of this Agreement that would alter the rights of Lenders with respect to Cash Collateral. Without the consent of such Lender, no Lender's "COMMITTED SUM" under the Letter of Credit Commitment may be increased. Without the consent of the Administrative Agent and Determining Lenders, no provision of SECTION 10 may be amended, modified, or waived.

- (c) Any conflict or ambiguity between the terms and provisions herein and terms and provisions in any other Loan Paper shall be controlled by the terms and provisions herein.
- (d) No course of dealing nor any failure or delay by the Administrative Agent, any Lender, or any of their respective Representatives with respect to exercising any Right of the Administrative Agent or any Lender hereunder shall operate as a waiver thereof. A waiver must be in writing and signed by the Administrative Agent and Determining Lenders (or by all Lenders, if required hereunder) to be effective, and such waiver will be effective only in the specific instance and for the specific purpose for which it is given.
- 11.11 MULTIPLE COUNTERPARTS. This Agreement may be executed in a number of identical counterparts, each of which shall be deemed an original for all purposes and all of which constitute, collectively, one agreement; but, in making proof of this Agreement, it shall not be necessary to produce or account for more than one such counterpart. It is not necessary that each Lender execute the same counterpart so long as identical counterparts are executed by the Credit Parties, each Lender, and the Administrative Agent. This Agreement shall become effective when counterparts hereof shall have been executed and delivered to the Administrative Agent by each Lender, the Administrative Agent, and each Credit Party, or, when the Administrative Agent shall have received telecopied, telexed, or other evidence satisfactory to it that such party has executed and is delivering to the Administrative Agent a counterpart hereof.

11.12 SUCCESSORS AND ASSIGNS; ASSIGNMENTS AND PARTICIPATIONS.

(a) This Agreement shall be binding upon, and inure to the benefit of the parties hereto and their respective successors and assigns, except that (i) the Company may not, directly or indirectly, assign or transfer, or attempt to assign or transfer, any of its Rights, duties or obligations under any Loan Papers without the express written consent of all Lenders, and (ii) except as permitted under this Section, no Lender may transfer, pledge, assign, sell any participation in, or otherwise encumber its portion of the Obligation.

- (b) Each Lender may assign to one or more Eligible Assignees all or a portion of its Rights and obligations under this Agreement and the other Loan Papers; provided, however, that:
 - (i) each such assignment shall be to an Eligible Assignee;
 - (ii) except in the case of an assignment to another Lender or to an Affiliate of a Lender, or an assignment of all of a Lender's Rights and obligations under this Agreement and the other Loan Papers, any such partial assignment shall be in an amount at least equal to \$5,000,000, and, after such assignment, no Lender shall have a Committed Sum of less than \$5,000,000 (in each case unless otherwise agreed by the Company and the Administrative Agent);
 - (iii) each such assignment by a Lender shall be of a constant, and not varying, percentage of all of its Rights and obligations under this Agreement; and
 - (iv) the parties to such assignment shall execute and deliver to the Administrative Agent for its acceptance an Assignment and Acceptance Agreement, together with a processing fee of \$3,500.

Upon execution, delivery, and acceptance of such Assignment and Acceptance Agreement, the assignee thereunder shall be a party hereto and, to the extent of such assignment, have the obligations, Rights, and benefits of a Lender under the Loan Papers and the assigning Lender shall, to the extent of such assignment, relinquish its rights and be released from its obligations under the Loan Papers. If the assignee is not incorporated under the Laws of the United States of America or a state thereof, it shall deliver to the Company and the Administrative Agent certification as to exemption from deduction or withholding of Taxes in accordance with SECTION 3.6.

- (c) The Administrative Agent shall maintain at its address referred to in SECTION 11.3 a copy of each Assignment and Acceptance Agreement delivered to and accepted by it and a register for the recordation of the names and addresses of the Lenders and their respective Committed Sums, and principal amount of the Drawings, if any, owing to each Lender from time to time (the "REGISTER"). The entries in the Register shall be conclusive and binding for all purposes, absent manifest error, and the Company, the Administrative Agent, Issuing Lender, and the Lenders may treat each Person whose name is recorded in the Register as a Lender hereunder for all purposes of the Loan Papers. The Register shall be available for inspection by the Company or any Lender at any reasonable time and from time to time upon reasonable prior notice. Upon the consummation of any assignment in accordance with this Section, SCHEDULE 2.1 shall automatically be deemed amended (to the extent required) by the Administrative Agent to reflect the name, address, and respective Committed Sums of the assignor and assignee.
- (d) Upon its receipt of an Assignment and Acceptance Agreement executed by the parties thereto, and payment of the processing fee, the Administrative Agent shall, if such Assignment and Acceptance has been completed and is in substantially the form of EXHIBIT B hereto: (i) accept such Assignment and Acceptance Agreement, (ii) record the information contained therein in the Register, and (iii) give prompt notice thereof to the parties thereto.
- (e) Subject to the provisions of this Section and in accordance with applicable Law, any Lender may, in the ordinary course of its commercial lending business and in accordance with applicable Law, at any time sell to one or more Persons (each a "PARTICIPANT") participating interests in its portion of the Obligation. In the event of any such sale to a Participant, (i) such

5 a

Lender shall remain a "Lender" under this Agreement and the Participant shall not constitute a "Lender" hereunder, (ii) such Lender's obligations under this Agreement shall remain unchanged, (iii) such Lender shall remain solely responsible for the performance thereof, (iv) such Lender shall remain the holder of its share of the Obligation for all purposes under this Agreement, (v) the Company and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's Rights and obligations under the Loan Papers, and (vi) such Lender shall be solely responsible for any withholding taxes or any filing or reporting requirements relating to such participation and shall hold the Company and the Administrative Agent and their respective successors, permitted assigns, officers, directors, employees, agents, and representatives harmless against the same. Participants shall have no Rights under the Loan Papers, other than certain voting Rights as provided below. Subject to the following, each Lender shall be entitled to obtain (on behalf of its Participants) the benefits of SECTION 3.4 with respect to all participations in its part of the Obligation outstanding from time to time so long as the Company shall not be obligated to pay any amount in excess of the amount that would be due to such Lender under SECTION 3.4 calculated as though no participations have been made. No Lender shall sell any participating interest under which the Participant shall have any Rights to approve any amendment, modification, or waiver of any Loan Paper, except to the extent such Lender has voting rights pursuant to SECTION 11.10; provided that, in those cases where a Participant is entitled to the benefits of SECTION 3.4 or a Lender grants Rights to its Participants to approve amendments to or waivers of the Loan Papers respecting the matters previously described in this sentence, such Lender must include a voting mechanism in the relevant participation agreement or agreements, as the case may be, whereby a majority of such Lender's portion of the Obligation (whether held by such Lender or Participant) shall control the vote for all of such Lender's portion of the Obligation. Except in the case of the sale of a participating interest to another Lender, the relevant participation agreement shall not permit the Participant to transfer, pledge, assign, sell participations in, or otherwise encumber its portion of the Obligation, unless the consent of the transferring Lender (which consent will not be unreasonably withheld) has been obtained.

- (f) Notwithstanding any other provision set forth in this Agreement, any Lender may at any time assign and pledge all or any portion of its loans and advances hereunder to any Federal Reserve Bank as collateral security pursuant to Regulation A and any Operating Circular issued by such Federal Reserve Bank. No such assignment shall release the assigning Lender from its obligations hereunder.
- (g) Any Lender may furnish any information concerning the Credit Parties in the possession of such Lender from time to time to Eligible Assignees and Participants (including prospective Eligible Assignees and Participants), subject, however, to SECTION 11.16 hereof.
- 11.13 DISCHARGE ONLY UPON PAYMENT IN FULL; REINSTATEMENT IN CERTAIN CIRCUMSTANCES. The Credit Parties' obligations under the Loan Papers shall remain in full force and effect until termination of the Letter of Credit Commitment and payment in full of all interest, fees, and other amounts of the Obligation then due and owing, and termination of the Letter of Credit except that SECTION 2.1(h), SECTION 3.4, SECTION 3.5(a), SECTION 3.5(b), and SECTION 11, and any other provisions under the Loan Papers expressly intended to survive by the terms hereof or by the terms of the applicable Loan Papers, shall survive such termination. If at any time any payment of the principal, interest, fees, or any other amount payable by any Credit Party under any Loan Paper is rescinded or must be otherwise restored or returned upon the insolvency, bankruptcy, or reorganization of such Credit Party or otherwise, the obligations of such Credit Party under the Loan Papers with respect to such payment shall be reinstated as though such payment had been due but not made at such time.

11.14 Judgment Currency.

- (a) If, for the purposes of obtaining judgment in any court, it is necessary to convert a sum due hereunder from a currency (the "ORIGINAL CURRENCY") into another currency (the "OTHER CURRENCY"), the parties hereto agree, to the fullest extent that they may effectively do so, such conversion shall be done at the Exchange Rate used on the second Business Day preceding that day on which final judgment is given.
- (b) The obligation of the Credit Parties in respect of any sum due in the Original Currency from any of them to any Lender, Issuing Lender, or the Administrative Agent under this Agreement shall, notwithstanding any judgment in any Other Currency, be discharged only if and to the extent that on the Business Day following receipt by such Lender, Issuing Lender, or the Administrative Agent of any sum adjudged to be so due in such Other Currency such Lender, Issuing Lender, or the Administrative Agent may in accordance with normal banking procedures purchase such amount of the Original Currency with such Other Currency at its principal office in London, England which the Administrative Agent could have purchased on the second Business Day preceding that on which the final judgment referred to in SECTION 11.9 is given; if the amount of the Original Currency so purchased is less than the amount of the Original Currency which the Administrative Agent could have purchased on the second Business Day preceding that on which such final judgment is given, each Credit Party agrees, as a separate obligation and notwithstanding any such judgment, to indemnify such Lender, Issuing Lender, or the Administrative Agent against such difference, and if the amount of the Original Currency so purchased exceeds the amount of the Original Currency which the Administrative Agent could have purchased on the second Business Day preceding that on which such final judgment is given, such Lender, Issuing Lender, or the Administrative Agent agrees to remit such excess to the applicable Credit Party.
- 11.15 Setoff. In addition to any rights and remedies of the Lenders provided by Law, upon the occurrence and during the continuance of any Event of Default, each Lender is authorized at any time and from time to time, without prior notice to any Credit Party or any other obligor, any such notice being waived by such Credit Party to the fullest extent permitted by Law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held by, and other indebtedness at any time owing by, such Lender to or for the credit or the account of such Credit Party, or any of them, against any and all Obligations owing to such Lender, now or hereafter existing, irrespective of whether or not the Administrative Agent or such Lender shall have made demand under this Agreement or any other Loan Paper and although such Obligations may be contingent or unmatured. Each Lender agrees promptly to notify such Credit Party and Administrative Agent after any such setoff and application made by such Lender; provided, however, that the failure to give such notice shall not affect the validity of such set-off and application.
- 11.16 CONFIDENTIALITY. The Administrative Agent and each Lender (each, a "LENDING PARTY") agrees to use its best efforts to keep confidential any information furnished or made available to it by the Credit Parties pursuant to this Agreement that is marked confidential; provided that nothing herein shall prevent any Lending Party from disclosing such information: (a) to any other Lending Party, or any officer, director, employee, agent, or advisor of any Lending Party, (b) to its auditors, accountants, legal counsel or other professional advisors, (c) as required by any Law, (d) upon the order of any court or administrative agency, (e) upon the request or demand of any regulatory agency or authority, (f) that is or becomes available to the public or that is or becomes available to any Lending Party other than as a result of a disclosure by any Lending Party prohibited by this Agreement, (g) in connection with any litigation to which such Lending Party or any of its affiliates may be a party, (h) to the extent necessary in connection with the exercise of any remedy under this Agreement or any other Loan Paper, and (i) subject

to provisions substantially similar to those contained in this Section, to any Affiliate of any Lending Party or to any actual or proposed participant or assignee.

11.17 USA PATRIOT ACT NOTICE. Each Lender that is subject to the Act (as hereinafter defined) and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Company that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "ACT"), it is required to obtain, verify and record information that identifies the Company and the Guarantors, which information includes the name and address of the Company and the Guarantors and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Company and the Guarantors in accordance with the Act.

 $\,$ EXECUTED by the parties hereto as of the day and year first above written.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK. SIGNATURE PAGE(S) FOLLOW.

LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

62

COMPANY:

FLOWSERVE B.V.

By: /s/ John M. Nanos

John M. Nanos Managing Director

ADDRESS FOR NOTICES:

5215 N. O'Connor Blvd., Suite 2300

Irving, Texas 75039

Attention: David F. Chavenson Telephone No.: 972-443-6563 Facsimile No.: 972-445-6863

Signature Page to Letter of Credit and Reimbursement Agreement

 ${\tt GUARANTORS:}$

ADDRESS FOR NOTICES:

FLOWSERVE CORPORATION

5215 N. O'Connor Blvd., Suite 2300

Irving, Texas 75039

Attention: David F. Chavenson Telephone No.: 972-443-6563 Facsimile No.: 972-445-6863

By: /s/ David F. Chavenson

David F. Chavenson Vice President

FLOWSERVE INTERNATIONAL B.V. FLOWSERVE REPAIR & SERVICES B.V.

FLOWSERVE DORTMUND GMBH & CO. KG

By: FLOWSERVE DORTMUND

VERWALTUNGS-GMBH, its General

Partner

By: /s/ John M. Nanos

John M. Nanos Managing Director By: /s/ John M. Nanos.

John M. Nanos Managing Director

FLOWSERVE S.A.

Chairman

FLOWSERVE BELGIUM N.V.

By: /s/ John M. Nanos

John M. Nanos

By: /s/ John M. Nanos

John M. Nanos Director

By: /s/ Jan van Rens

Jan van Rens Managing Director

FLOWSERVE FLOW CONTROL BENELUX B.V.

By: FLOWSERVE B.V., its Managing Director

By: /s/ John M. Nanos.

John M. Nanos Managing Director

Signature Page to Letter of Credit and Reimbursement Agreement

ADMINISTRATIVE AGENT, ISSUING LENDER, AND A LENDER:

CALYON NEW YORK BRANCH

By: /s/ Philippe Soustra

Name: Philippe Soustra

Title: Executive Vice President

By: /s/ Attila Coach

Name: Attila Coach Title: Sr. Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

ABN AMRO BANK N.V.

By: /s/ Terrence J. Ward -----

Name: Terrence J. Ward

Title: Senior Vice President

By: /s/ Angela Noique

Name: Angela Noique

Title: Group Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

BANK OF AMERICA, N.A.

By: /s/ Brian D. Corum

Name: Brian D. Corum Title: Managing Director

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

BANK OF TOKYO-MITSUBISHI, LTD.

By: /s/ D. Barnell

Name: D. Barnell Title: Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

CITICORP NORTH AMERICA, INC.

By: /s/ Svetoslav Nikov

Name: Svetoslav Nikov Title: Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

CREDIT INDUSTRIEL ET COMMERCIAL

By: /s/ Anthony Rock /s/ Sean Mounier

Name: Anthony Rock Sean Mounier
Title: Vice President First Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

CREDIT SUISSE FIRST BOSON, ACTING THROUGH ITS CAYMAN ISLANDS BRANCH

By: /s/ James Moran

._____

Name: James Moran Title: Director

By: /s/ Denise Alvarez

Name: Denise Alvarez Title: Associate

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

MIZUHO CORPORATE BANK, LTD.

By: /s/ Raymond Ventura

Name: Raymond Ventura Title: Senior Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

LENDER:

WACHOVIA BANK, NATIONAL ASSOCIATION

By: /s/ Nathan R. Rantala

Name: Nathan R. Rantala Title: Vice President

Signature Page to Letter of Credit and Reimbursement Agreement

FIRST AMENDMENT AND LIMITED WAIVER TO LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT

THIS FIRST AMENDMENT AND LIMITED WAIVER TO LETTER OF CREDIT AND REIMBURSEMENT AGREEMENT (this "Amendment") is entered into as of March 15, 2005, among FLOWSERVE B.V., a company organized under the laws of the Netherlands ("Company"), FLOWSERVE CORPORATION, a New York corporation ("Parent"), the Material Subsidiaries (as defined in the Agreement) (each of Parent and each Material Subsidiary a "Guarantor" and collectively, the "Guarantors"), the LENDERS (hereinafter defined), and CALYON NEW YORK BRANCH, a duly licensed branch under the New York Banking Law of a foreign banking corporation organized under the laws of the Republic of France, as Administrative Agent (hereinafter defined) for itself and the other Lenders.

Reference is made to the Letter of Credit and Reimbursement Agreement, dated as of the 28th day of July, 2004 (as amended to date, the "Agreement"), among Company, Guarantors, Administrative Agent, and the Lenders party thereto. Unless otherwise defined in this Amendment, capitalized terms used herein shall have the meaning set forth in the Agreement; all Section references herein are to Sections in the Agreement; and all Paragraph references herein are to Paragraphs in this Amendment.

RECITALS

- A. Pursuant to **Sections 7.4(a)(i)** and **(iii)** of the Agreement, Parent is required to deliver the audited financial statements, opinion, and the accountants' certificate referred to therein within 100 days after the end of each fiscal year;
- B. Pursuant to *Sections 7.4(b)(ii)* and (v) of the Agreement, Company is required to deliver the audited financial statements, opinion, and accountants certificate referred to therein within 210 days after the end of each fiscal year;
- C. Company has requested that Determining Lenders waive compliance with such requirements under **Sections 7.4(a)(i)** and **(iii)** and **(v)** of the Agreement with respect to the fiscal year ended December 31, 2004, until September 30, 2005, and any consequences resulting from such non-compliance during such period; and
- D. Company and Lenders desire to amend the Agreement to permit senior unsecured debt and to modify certain of its financial definitions as set forth below.

Accordingly, for adequate and sufficient consideration, the parties hereto agree as follows:

1. Amendments to Agreement.

a. The definition of "Consolidated EBITDA" in Section 1.1 of the Agreement is hereby amended and restated in its entirety as follows:

"Consolidated EBITDA" means, for any period, Consolidated Net Income for such period, plus (a) without duplication and to the extent deducted in determining such Consolidated Net Income, the sum of (i) Consolidated Interest Expense for such period, (ii) consolidated income tax expense for such period, (iii) all amounts attributable to depreciation and amortization for such period, (iv) any

First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement extraordinary losses or extraordinary non-cash charges for such period, (v) the amount of premium payments made by Parent or its Subsidiaries associated with the repurchase or prepayment of the Subordinated Notes from the proceeds of the Fall 2001 Equity Issuance and the amount of such premium payments and unamortized fees associated with any further repurchase or prepayment of the Subordinated Notes to the extent such repurchase or prepayment is permitted hereunder, (vi) integration and restructuring charges in connection with the IDP Transactions and taken with respect to periods ended on or prior to December 31, 2001, (vii) integration and restructuring charges in connection with the Acquisition and taken with respect to periods ended on or prior to June 30, 2004, in an aggregate amount not to exceed \$40,000,000, and (viii) restructuring and integration charges taken with respect to periods beginning on July 1, 2003 and ended on or prior to December 31, 2004, in an aggregate amount not to exceed \$15,000,000, and minus (b) without duplication and to the extent included in determining such Consolidated Net Income, any extraordinary gains for such period, all determined on a consolidated basis in accordance with GAAP.

b. The definition of "Consolidated Interest Expense" in Section 1.1 of the Agreement is hereby amended and restated in its entirety as follows:

"Consolidated Interest Expense" means, for any period, for the Parent and its Subsidiaries on a consolidated basis, the sum, without duplication, of: (a) all interest, premium payments, fees, charges, and related expenses payable by the Parent and its Subsidiaries in connection with borrowed money(including capitalized interest) (other than premium payments associated with the repurchase or prepayment of the Subordinated Notes from proceeds of the Fall 2001 Equity Issuance and premium payments and unamortized fees associated with any further repurchase or prepayment of the Subordinated Notes to the extent such repurchase or prepayment is permitted hereunder) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP and payable in cash, (b) the portion of rent payable by the Parent and its Subsidiaries with respect to such period under capital leases that is treated as interest in accordance with GAAP and payable in cash, and (c) all fees, discounts, premiums, expenses, or similar amounts incurred by the Parent or any of its Subsidiaries in connection with the Receivables Program for such period, including purchase discounts (net of any loss reserves), purchase premiums, operating expense fees, structuring fees, collection agent fees, unutilized purchase limit fees, and other similar fees and expenses.

c. The definition of "New Subordinated Debt" in Section 1.1 of the Agreement amended and restated in its entirety as follows:

"New Unsecured Debt" means Indebtedness having the following characteristics: (i) the obligors shall be the Parent and/or Flowserve Finance B.V., (ii) such Indebtedness shall be unsecured, (iii) if such Indebtedness is subordinated, such Indebtedness shall be subordinated in right of payment to the Obligations in a manner reasonably acceptable to Administrative Agent, (iv) such Indebtedness shall not have any scheduled payment of principal, scheduled prepayment, scheduled mandatory redemption or sinking fund payment prior to December 31, 2009, (v) such Indebtedness shall not contain any provision prohibiting the creation or assumption of any Lien on any of the properties or assets of Parent or its Subsidiaries, whether then owned or thereafter acquired, or prohibiting guaranties by Parent or any of its Subsidiaries to secure payment of the Obligations or any agreement renewing, refinancing or extending the Obligations or this Agreement, (vi) the Parent shall be in compliance with Sections 8.11, 8.12 and 8.13 on a pro forma basis after giving effect to the incurrence of such Indebtedness, (vii) other terms and conditions shall be no less favorable to the Parent or its Subsidiaries or the Lenders in any material respect than the terms and conditions

applicable to the Subordinated Notes, and (viii) such Indebtedness shall be issued pursuant to documentation reasonably satisfactory to the Administrative Agent.

d. The definition of "Total Debt" in Section 1.1 of the Agreement is hereby amended and restated in its entirety as follows:

"Total Debt" means, at any time, the total consolidated Indebtedness of the Parent and its Subsidiaries at such time (excluding (a) Indebtedness under Section 8.1(k), and (b) Indebtedness of the type described in clause (i) of the definition of such term and under Section 8.1(l), except in each case to the extent of any unreimbursed drawings or payments thereunder).

e. Section 8.1(0) of the Agreement is hereby amended and restated in its entirety to read as follows:

(o) Indebtedness of the Parent constituting New Unsecured Debt in an aggregate principal amount not to exceed \$325,000,000 or its Dollar Equivalent; provided, however, that the proceeds of any such New Unsecured Debt shall be used to redeem, repurchase, prepay, or otherwise acquire for consideration, the Indebtedness permitted under clause (i) above (including any premium payments) until repaid in full and, once repaid, such amounts referenced in clause (i) above shall no longer be permitted under this Agreement.

f. Section 8.8 of the Agreement is hereby amended and restated in its entirety to read as follows:

Engage at any time in any business or business activity other than the business currently conducted by the Credit Parties and business activities reasonably incidental thereto, including any activities permitted hereunder; provided, however, that the Parent shall be entitled to create a whollyowned Subsidiary engaged solely in the business of providing the insurance coverage required under Section 7.2 hereof solely to the Credit Parties, so long as such Subsidiary is adequately capitalized to satisfy the requirements of Section 7.2 and investments therein do not exceed \$1,000,000 in the aggregate (a "Captive Insurance Company").

g. Section 8.9(b) of the Agreement is hereby amended and restated in its entirety to read as follows:

(b) (i) With respect to any Credit Party, make any distribution, whether in cash, property, securities, or a combination thereof, other than regular scheduled payments of interest as and when due (to the extent not prohibited by applicable subordination provisions), in respect of, or pay, or offer or commit to pay, or directly or indirectly redeem, repurchase, retire, or otherwise acquire for consideration, or set apart any sum for the aforesaid purposes, any subordinated Indebtedness (except the redemption or repurchase of the Subordinated Notes with the proceeds of the issuance of the New Unsecured Debt), or (ii) pay in cash any amount in respect of any Indebtedness or preferred equity interests that may at the obligor's option be paid in kind or in other securities; provided that the Parent was permitted to repurchase and/or prepay the Subordinated Notes to the extent provided in the final proviso of Section 2.13(c) of the Original Credit Agreement.

2. Waiver of certain Financial Reporting Covenants.

A. Subject to the terms and conditions set forth herein and in reliance on the representations and warranties of each Credit Party herein contained, the Lenders hereby temporarily waive (i) compliance with the

requirement under **Sections 7.4(a)(i)** and **(iii)** and **(v)** of the Agreement that the financial statements for the fiscal year ending December 31, 2004, delivered pursuant to **Section 7.4(a)(i)** and **7.4(b)(ii)** of the Agreement are audited and accompanied by an opinion of an independent public accountant and a certificate of such accountant certifying matters set forth in **Section 7.4(a)(ii)** and **7.4(b)(v)** of the Agreement, (ii) any consequences or further affirmative obligations of the Parent or the Company under the Agreement resulting from such noncompliance, and (iii) any Default or Event of Default arising from an event of default under any agreement or instrument evidencing or governing Material Indebtedness of the Credit Parties arising from the failure to provide similar financial information (or opinions or certifications thereof) or annual or quarterly public filings during or for the fiscal year ending December 31, 2004, or for the first three fiscal quarters of the fiscal year ending December 31, 2005; in each case, resulting from such noncompliance during the period from the Effective Date (as defined in **Section 3**) through and including the earlier to occur of (a) the date of delivery of audited financial statements for the fiscal year ending December 31, 2004, and the accompanying accountant's opinion and certificate in accordance with **Sections 7.4(a)(i)** and **(iii)** and **7.4(b)(ii)** and **(v)** of the Agreement to the Lenders, and (b) September 30, 2005 (such temporary waiver being the "**Waiver**"); provided, however, all waivers and accommodations made to the Credit Parties in this Waiver shall be rescinded and be null and void and of no force and effect upon the commencement by any holder of Material Indebtedness of the Credit Parties of any action, exercising rights with respect to collateral or rights to accelerate arising from any default or event of default under any agreement or instrument evidencing or governing Material Indebtedness of the Credit Parties.

- B. The Waiver shall be limited precisely as written and relates solely to the temporary noncompliance by the Parent and the Company with **Sections 7.4(a)** (i) and (iii) and 7.4(b)(ii) and (v) of the Agreement in the manner and to the extent described above, and nothing in this Waiver shall be deemed to:
 - 1. constitute a waiver of compliance by any Credit Party with respect to (i) **Section 7.4(a)(i)** or (iii) and 7.4(b)(ii) and (v) of the Agreement in any other instance or (ii) any other term, provision or condition of the Agreement or any other instrument or agreement referred to therein; or
 - 2. prejudice any right or remedy that Administrative Agent or any Lender may now have (except to the extent such right or remedy was based upon existing defaults that will not exist after giving effect to this Waiver) or may have in the future under or in connection with the Agreement or any other instrument or agreement referred to therein.
- C. Except as expressly set forth herein, the terms, provisions and conditions of the Agreement shall remain in full force and effect and in all other respects are hereby ratified and confirmed.
- 3. Effective Date. Notwithstanding any contrary provision, this Amendment is not effective until the date (the "Effective Date") upon which (a) the representations and warranties in this Amendment are true and correct; (b) Administrative Agent receives counterparts of this Amendment executed by Company, each Guarantor, and Determining Lenders; (c) Company pays all reasonable costs, fees, and expenses of Administrative Agent's counsel incurred in connection with the Loan Papers, including without limitation, all reasonable legal fees and expenses outstanding on the Effective Date; and (d) Administrative Agent shall have received payment for the account of each Lender that executes this Amendment on or before 5:00 p.m. New York time on March 15, 2005, of an amendment fee equal to 0.02% of such Lender's Committed Sum on the Effective Date.
- 4. <u>Acknowledgment and Ratification</u>. As a material inducement to Administrative Agent and the Lenders to execute and deliver this Amendment, Company and each Guarantor (a) consent to the agreements in this Amendment and (b) agree and acknowledge that the execution, delivery, and performance of this

First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement Amendment shall in no way release, diminish, impair, reduce, or otherwise affect the respective obligations of Company or Guarantors under their respective Loan Papers, which Loan Papers shall remain in full force and effect, and all Liens, guaranties, and rights thereunder are hereby ratified and confirmed.

- 5. Representations. As a material inducement to Lenders to execute and deliver this Amendment, Company and each Guarantor represent and warrant to Lenders (with the knowledge and intent that Lenders are relying upon the same in entering into this Amendment) that as of the Effective Date and as of the date of execution of this Amendment (after giving effect to the amendments and waiver set forth herein), (a) all representations and warranties in the Loan Papers are true and correct in all material respects as though made on the date hereof, except to the extent that (i) any of them speak to a different specific date, (ii) the facts on which any of them were based have been changed by transactions contemplated or permitted by the Loan Papers, or (iii) the financial statements delivered pursuant to Section 6.5 were subject to subsequent restatements, and (b) no Default or Event of Default exists other than as waived herein.
- 6. <u>Delivery of Financial Statements</u>. Parent and Company hereby agrees to provide within 100 days after the end of the fiscal year ending December 31, 2004, its unaudited consolidated balance sheet and related statements of income, stockholders' equity and cash flows showing the financial condition of Parent and its consolidated Subsidiaries and of Company and its consolidated Subsidiaries as of the close of the fiscal year ending 2004 and the results of its operations and the operations of such Subsidiaries during such year, and a certificate of a Financial Officer opining on or certifying such statements and certifying that no Event of Default or Default has occurred or, if such an Event of Default or Default has occurred, specifying the nature and extent thereof and any corrective action taken or proposed to be taken with respect thereto, and, setting forth the Parent's calculation of Excess Cash Flow.
- 7. Expenses. Company shall pay all reasonable costs, fees, and expenses paid or incurred by Administrative Agent incident to this Amendment, including, without limitation, the reasonable fees and expenses of Administrative Agent's counsel in connection with the negotiation, preparation, delivery, and execution of this Amendment and any related documents.

8. Miscellaneous.

- 8.1 This Amendment is a "Loan Paper" referred to in the Agreement, and the provisions relating to Loan Papers in Section 11 of the Agreement are incorporated in this Amendment by reference. Unless stated otherwise (a) the singular number includes the plural and vice versa and words of any gender include each other gender, in each case, as appropriate, (b) headings and captions may not be construed in interpreting provisions, (c) this Amendment must be construed, and its performance enforced, under New York law, (d) if any part of this Amendment is for any reason found to be unenforceable, all other portions of it nevertheless remain enforceable, and (e) this Amendment may be executed in any number of counterparts with the same effect as if all signatories had signed the same document, and all of those counterparts must be construed together to constitute the same document.
- 8.2 The Loan Papers shall remain unchanged and in full force and effect, except as provided in this Amendment, and are hereby ratified and confirmed. On and after the Effective Date, all references to the "Agreement" shall be to the Agreement as herein amended. The execution, delivery, and effectiveness of this Amendment shall not, except as expressly provided herein, operate as a waiver of any rights of Lenders under any Loan Paper, nor constitute a waiver under any of the Loan Papers.
- 9. <u>Entire Agreement.</u> This Amendment represents the final agreement between the parties about the subject matter of this Amendment and may not be contradicted by

First Amendment and Limited Waiver to Letter of Credit and Reimbursement Agreement evidence of prior, contemporaneous, or subsequent oral agreements of the parties. There are no unwritten oral agreements between the parties.

10. Parties. This Amendment binds and inures to Company, Guarantors, Administrative Agent, Lenders, and their respective successors and assigns.

The parties hereto have executed this Amendment in multiple counterparts to be effective as of the Effective Date.

Remainder of Page Intentionally Blank. Signature Pages to Follow.

COMPANY:

FLOWSERVE B.V.

By: /s/ John M. Nanos

John M. Nanos Managing Director

Address for Notices:

5215 N. O'Connor Blvd., Suite 2300 Irving, Texas 75039

Attention: Paul Fehlman Telephone No.: 972-443-6517 Facsimile No.: 972-445-6817

GUARANTORS:

Address for Notices:

5215 N. O'Connor Blvd., Suite 2300 Irving, Texas 75039 Attention: Paul Fehlman

Acting Vice President and Treasurer Telephone No.: 972-443-6517 Facsimile No.: 972-445-6817

FLOWSERVE INTERNATIONAL B.V. FLOWSERVE REPAIR & SERVICES B.V.

By: /s/ John M. Nanos

John M. Nanos Managing Director

FLOWSERVE S.A.

By: /s/ John M. Nanos

John M. Nanos Chairman

FLOWSERVE FLOW CONTROL BENELUX B.V.

By: FLOWSERVE B.V., its Managing Director

By: /s/ John M. Nanos John M. Nanos Managing Director

FLOWSERVE CORPORATION

By: /s/ Paul Fehlman

Paul Fehlman

Acting Vice President and Treasurer

FLOWSERVE DORTMUND GMBH & CO. KG

By: FLOWSERVE DORTMUND VERWALTUNGS-GMBH, its General

Partner

By: /s/ John M. Nanos

John M. Nanos Managing Director

FLOWSERVE BELGIUM N.V.

By: /s/ John M. Nanos

John M. Nanos Director

By: /s/ Jan van Rens

Jan van Rens Managing Director

ADMINISTRATIVE AGENT, ISSUING LENDER, AND A LENDER:

CALYON NEW YORK BRANCH

By: /s/ Philippe Soustra

Name: Philippe Soustra Title: Senior Vice President

By: /s/ Attila Coach

Name: Attila Coach Title: Managing Director

ABN AMRO BANK N.V.

By: /s/ Terrence J. Ward

Name: Terrence J. Ward Title: Senior Vice President

By: /s/ Agnacio Pineros

Name: Agnacio Pineros Title: Vice President

BANK OF AMERICA, N.A.

By: /s/ John Pocalyko

Name: John Pocalyko Title: Senior Vice President

BANK OF TOKYO-MITSUBISHI, LTD.

By: /s/
Name:
Title:

CITICORP NORTH AMERICA, INC.

By: /s/ Cornelius Mahon

Name: Cornelius Mahon Title: Vice-President

CRÉDIT INDUSTRIEL ET COMMERCIAL

By: /s/ Anthony Rock

Name: Anthony Rock Title: Vice President

By: /s/ Brian O'Leary

Name: Brian O'Leary Title: Vice President

CREDIT SUISSE FIRST BOSTON, ACTING THROUGH ITS CAYMAN ISLANDS BRANCH

By: /s/ James Moran

Name: James Moran Title: Managing Director

By: /s/ Denise L. Alvarez

Name: Denise L. Alvarez Title: Associate

MIZUHO CORPORATE BANK, LTD.

By: /s/ Robert Gallagher

Name: Robert Gallagher Title: Senior Vice President

WACHOVIA BANK, NATIONAL ASSOCIATION

By: /s/ Nathan R. Rantala

Name: Nathan R. Rantala Title: Vice President